UNITED STATES SECURITIES AND EXCHANGE COMMISSION

Washington, D.C. 20549

FORM 10-K (Mark One)

(v v)
\boxtimes	ANNUAL REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934
	For the fiscal year ended December 31, 2021

or

☐ TRANSITION REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934
For the transition period from to

Commission File Number 001-35985



CDW CORPORATION

(Exact name of registrant as specified in its charter)

Delaware

(State or other jurisdiction of incorporation or organization)

75 Tri-State International Lincolnshire , Illinois

(Address of principal executive offices)

26-0273989

(I.R.S. Employer Identification No.)

60069

(Zip Code)

(847) 465-6000

(Registrant's telephone number, including area code)

None

(Former name, former address and former fiscal year, if changed since last report)

Securities registered pursuant to Section 12(b) of the Act:

Title of each class	Trading symbol(s)	Name of each exchange on which registered
Common stock, par value \$0.01 per share	CDW	Nasdaq Global Select Market

Securities registered pursuant to Section 12(g) of the Act: None
ndicate by check mark if the registrant is a well-known seasoned issuer, as defined in Rule 405 of the Securities Act. 🗵 Yes 🗆 No
ndicate by check mark if the registrant is not required to file reports pursuant to Section 13 or Section 15(d) of the Act. Yes No
ndicate by check mark whether the registrant (1) has filed all reports required to be filed by Section 13 or 15(d) of the Securities Exchange Act of 1934 during the preceding 12 nonths (or for shorter period that the registrant was required to file such reports), and (2) has been subject to such filing requirements for the past 90 days. Yes No
ndicate by check mark whether the registrant has submitted electronically every Interactive Data File required to be submitted pursuant to Rule 405 of Regulation S-T §232.405 of this chapter) during the preceding 12 months (or for such shorter period that the registrant was required to submit such files). Yes No

Table of Contents

		elerated filer, a non-accelerated filer, a smaller reporting company, or emerging ground company," and "emerging growth company" in Rule 12b-2 of the Exchange							
Large accelerated filer	×	Accelerated filer							
Non-accelerated filer		Smaller reporting company							
		Emerging growth company							
If an emerging growth company, indicate by check mark if the revised financial accounting standards provided pursuant to Section	egistrant has el on 13(a) of the	lected not to use the extended transition period for complying with any new or Exchange Act.							
		ation to its management's assessment of the effectiveness of its internal control of the registered public accounting firm that prepared or issued its audit report.							
Indicate by check mark whether the registrant is a shell company (as defined in R	Rule 12b-2 of the Exchange Act). ☐ Yes 🗵 No							
The aggregate market value of the voting and non-voting common recently completed second fiscal quarter, was \$24,115 million, bas		y non-affiliates of the registrant as of June 30, 2021, the last business day of the share closing sale price of \$174.65 on that date.	registrant's most						
As of February 24, 2022, there were 134,944,328 shares of commo	on stock, \$0.01	par value, outstanding.							
DOCU	UMENTS INC	CORPORATED BY REFERENCE							
	Certain parts of the registrant's definitive proxy statement for its 2022 annual meeting of stockholders to be held on May 19, 2022, which will be filed with the Securities and Exchange Commission on or before April 30, 2022, are incorporated by reference into Part III of this Annual Report on Form 10-K.								

CDW CORPORATION AND SUBSIDIARIES

ANNUAL REPORT ON FORM 10-K Year Ended December 31, 2021

TABLE OF CONTENTS

<u>Item</u>		<u>Page</u>
PART I		
Item 1.	<u>Business</u>	<u>4</u>
Item 1A.	Risk Factors	<u>10</u>
Item 1B.	<u>Unresolved Staff Comments</u>	<u>22</u>
Item 2.	<u>Properties</u>	22 22 22
Item 3.	<u>Legal Proceedings</u>	<u>22</u>
Item 4.	Mine Safety Disclosures	<u>22</u>
	<u>Information about our Executive Officers</u>	<u>23</u>
PART II		
Item 5.	Market for Registrant's Common Equity, Related Stockholder Matters and Issuer Purchases of Equity Securities	<u>24</u>
Item 6.	[RESERVED]	<u>26</u>
Item 7.	Management's Discussion and Analysis of Financial Condition and Results of Operations	<u>27</u>
Item 7A.	Quantitative and Qualitative Disclosures About Market Risk	<u>41</u>
Item 8.	Financial Statements and Supplementary Data	<u>42</u>
Item 9.	Changes in and Disagreements with Accountants on Accounting and Financial Disclosure	<u>80</u>
Item 9A.	Controls and Procedures	<u>80</u>
Item 9B.	Other Information	<u>82</u>
Item 9C.	Disclosure Regarding Foreign Jurisdictions that Prevent Inspections	<u>82</u>
PART III		
Item 10.	<u>Directors, Executive Officers and Corporate Governance</u>	<u>83</u>
Item 11.	Executive Compensation	<u>83</u>
Item 12.	Security Ownership of Certain Beneficial Owners and Management and Related Stockholder Matters	<u>83</u>
Item 13.	Certain Relationships and Related Transactions, and Director Independence	<u>83</u>
Item 14.	Principal Accountant Fees and Services	<u>83</u>
PART IV		
Item 15.	Exhibits and Financial Statement Schedules	<u>84</u>
Item 16.	Form 10-K Summary	<u>89</u>
SIGNATURES		<u>90</u>

FORWARD-LOOKING STATEMENTS

This report contains "forward-looking statements" within the meaning of the federal securities laws. All statements other than statements of historical fact are forward-looking statements. These statements relate to analyses and other information, which are based on forecasts of future results or events and estimates of amounts not yet determinable. These statements also relate to our future prospects, developments and business strategies. We claim the protection of The Private Securities Litigation Reform Act of 1995 for all forward-looking statements in this report.

These forward-looking statements are identified by the use of terms and phrases such as "anticipate," "assume," "believe," "estimate," "expect," "goal," "intend," "plan," "potential," "predict," "project," "target" and similar terms and phrases or future or conditional verbs such as "could," "may," "should," "will," and "would." However, these words are not the exclusive means of identifying such statements. Although we believe that our plans, intentions and other expectations reflected in or suggested by such forward-looking statements are reasonable, we cannot assure you that we will achieve those plans, intentions or expectations. All forward-looking statements are subject to risks and uncertainties that may cause actual results or events to differ materially from those that we expected.

Important factors that could cause actual results or events to differ materially from our expectations, or cautionary statements, are disclosed under the sections entitled "Risk Factors" and "Management's Discussion and Analysis of Financial Condition and Results of Operations" included elsewhere in this report. All written and oral forward-looking statements attributable to us, or persons acting on our behalf, are expressly qualified in their entirety by those cautionary statements as well as other cautionary statements that are made from time to time in our other Securities and Exchange Commission ("SEC") filings and public communications. You should evaluate all forward-looking statements in the context of these risks and uncertainties.

We caution you that the important factors referenced above may not reflect all of the factors that could cause actual results or events to differ from our expectations. In addition, we cannot assure you that we will realize the results or developments we expect or anticipate or, even if substantially realized, that they will result in the consequences or affect us or our operations in the way we expect. The forward-looking statements included in this report are made only as of the date hereof or, with respect to any documents incorporated by reference, available at the time such document was prepared or filed with the SEC. We undertake no obligation to publicly update or revise any forward-looking statement as a result of new information, future events or otherwise, except as otherwise required by law.

PART I

Item 1. Business

Our Company

CDW Corporation (together with its subsidiaries, the "Company," "CDW" or "we"), a Fortune 500 company and member of the S&P 500 Index, is a leading multi-brand provider of information technology ("IT") solutions to small, medium and large business, government, education and healthcare customers in the United States ("US"), the United Kingdom ("UK") and Canada. Our broad array of offerings ranges from discrete hardware and software products to integrated IT solutions and services that include on-premise, hybrid and cloud capabilities across hybrid infrastructure, digital experience and security.

On December 1, 2021, we completed our previously announced acquisition of Sirius Computer Solutions, Inc. ("Sirius"). This strategic acquisition is expected to enhance our services and solutions capabilities in key areas, including hybrid infrastructure, security, digital and data innovation, and cloud and managed services, as well as add services scale, further balancing and diversifying our portfolio mix. The addition of Sirius strengthens our role as the trusted technology advisor to our customers, with the expertise and portfolio breadth, depth and scale to orchestrate complete customer-centric solutions.

We are vendor, technology and consumption model "agnostic", with a solutions portfolio including more than 100,000 products and services from more than 1,000 leading and emerging brands. Our solutions are delivered in physical, virtual and cloud-based environments through approximately 9,900 customer-facing coworkers, including sellers, highly-skilled technology specialists and advanced service delivery engineers. We are a leading sales channel partner for many original equipment manufacturers ("OEMs"), software publishers and cloud providers (collectively, our "vendor partners"), whose products we sell or include in the solutions we offer. We provide our vendor partners with a cost-effective way to reach customers and deliver a consistent brand experience through our established end-market coverage, technical expertise and extensive customer access.

We simplify the complexities of technology across design, selection, procurement, integration and management for our customers. Our goal is to have our customers, regardless of their size, view us as a trusted adviser and extension of their IT resources. Our multi-brand offering approach enables us to identify the products or combination of products from our vendor partners that best address each customer's specific IT requirements.

We have capabilities to provide integrated IT solutions in more than 150 countries for customers with primary locations in the US, UK and Canada, which are large and growing markets. According to the International Data Corporation ("IDC"), the total US, UK and Canadian IT market generated approximately \$1.2 trillion in sales in 2021. We believe our addressable markets in the US, UK and Canada represent approximately \$400 billion in annual sales. These are highly fragmented markets served by thousands of IT resellers and solutions providers. For the year ended December 31, 2021, we estimate that our total Net sales of \$20.8 billion represented approximately 5% of our addressable markets. We believe that demand for IT will continue to outpace general economic growth in the markets we serve, fueled by new technologies, including hybrid and cloud computing, virtualization and mobility as well as growing end-user demand for security, efficiency and productivity.

Value Proposition

We are positioned in the middle of the IT ecosystem where we procure products from OEMs, software publishers, cloud providers and wholesale distributors and provide added value to our customers by helping them navigate through complex options and implement the best solution for their business. In this role, we believe we provide unique value to both our vendor partners and our customers.

Our value proposition to our customers

- · Broad selection of products and multi-branded IT solutions
- Value-added services with integration capabilities
- Highly-skilled specialists and engineers
- Solutions across IT lifecycle

Our value proposition to our vendor partners

- Access to over 250,000 customers
- Large and established customer channels
- Strong distribution and implementation capabilities
- Customer relationships driving insight into technology roadmaps

Customers

We provide integrated IT solutions to over 250,000 small, medium and large business, government, education and healthcare customers throughout the US, UK and Canada.

We serve our customers through sales teams focused on customer end-markets that are supported by technical specialists and highly-skilled service delivery engineers. Our market segmentation allows us to customize our offerings and to provide enhanced expertise in designing and implementing IT solutions that meet our customer's specific needs.

We have three reportable segments, Corporate, Small Business and Public. Our Corporate segment primarily serves US private sector business customers with more than 250 employees. Our Small Business segment primarily serves US private sector business customers with up to 250 employees. Our Public segment is comprised of government agencies and education and healthcare institutions in the US. We also have two other operating segments: CDW UK and CDW Canada, each of which do not meet the reportable segment quantitative thresholds and, accordingly, are included in an all other category ("Other").

In our US business, which represents approximately 90% of our revenues, we currently have five dedicated customer channels: corporate, small business, government, education and healthcare, each of which generated \$1.8 billion or greater in Net sales in 2021. Net sales to customers in the UK and Canada combined generated \$2.6 billion in 2021. We believe this diversity of customer end-markets provides us with multiple avenues for growth and has been a key factor in our ability to weather economic and technology cycles and continue to gain market share.

Partners

We provide more than 100,000 products and services from more than 1,000 partners, including well-established companies such as Adobe, APC, Apple, Cisco, Dell EMC, Google, Hewlett Packard Enterprise, HP Inc., IBM, Intel, Lenovo, Microsoft, NetApp, Nutanix, Palo Alto Networks, Poly, Samsung, and VMware, as well as from emerging technology companies to expand our portfolio. This broad portfolio of partners and technologies enables us to offer customers significant options and meet customer demand for the products and solutions that best meet their needs. We believe our value proposition to vendor partners enables us to evolve our offering as new technologies emerge and new companies seek us as a channel partner.

In 2021, we generated over \$1.0 billion each of Net sales from five vendor partners and over \$100 million of Net sales from each of fifteen other vendor partners. We have received the highest level of certification from major vendor partners such as Cisco, Dell EMC, Hewlett Packard Enterprise, LG, Microsoft, Palo Alto Networks, Samsung, and VMware which reflects the extensive product and solution knowledge and capabilities that we bring to our customers' IT challenges. These certifications also provide us with access to favorable pricing, tools and resources, including vendor incentive programs, which we use to provide additional value to our customers. Our vendor partners also regularly recognize us with top awards and select us to develop and grow new customer solutions.

Product Procurement

We may purchase all or only some of the products our vendor partners offer for resale to our customers or for inclusion in the solutions we offer. Each vendor partner agreement provides for specific terms and conditions, which may include one or more of the following: product return privileges, price protection policies, purchase discounts and vendor incentive programs, such as purchase or sales rebates and cooperative advertising reimbursements. We also purchase software from major software publishers and cloud providers for resale to our customers or for inclusion in the solutions we offer. Our agreements allow us to resell cloud based solutions, software or other licensed products to the end-user customer.

In addition to purchasing products directly from our vendor partners, we purchase products from wholesale distributors for resale to our customers or for inclusion in the solutions we offer. These wholesale distributors provide logistics management and supply-chain services for us, as well as for our vendor partners.

For our US operations in 2021, we purchased approximately 50% of the products we sold as discrete products or as components of a solution directly from our vendor partners and the remaining 50% from wholesale distributors. Purchases from our two largest wholesale distributors, Ingram Micro and TD SYNNEX, were over 30% of total US purchases in 2021.

Inventory Management

We operate two distribution centers in North America: a 513,000 square foot facility in North Las Vegas, Nevada, and a 442,000 square foot facility in Vernon Hills, Illinois. We also operate a 120,000 square foot distribution center in Rugby, Warwickshire, UK. Leveraging our distribution and logistics capabilities, we handle and ship over 45 million units annually on an aggregate basis from our distribution centers.

We also have drop-shipment arrangements with many of our OEMs and wholesale distributors, which permit us to offer products to our customers without having to take physical delivery at our distribution centers. These arrangements represented approximately 50% of total North America Net sales in 2021. Electronic delivery for software licenses is approximately 20% of total North America Net sales in 2021.

We believe that the location of our distribution centers allows us to efficiently ship products to our customers and provide timely access to our principal distributors. We believe that our logistics and configuration capabilities delivered by our highly skilled and certified team enable us to customize technology for our customers to meet their unique needs.

We believe competitive sources of supply are available in substantially all of the product categories that we offer.

Competition

The market for technology products and services is highly competitive and subject to economic conditions and rapid technological changes. Competition is based on many things, including the ability to tailor specific solutions to customer needs, the quality and breadth of product and service offerings, knowledge and expertise of sales force, customer service, price, product availability, speed of delivery and credit availability. We face competition from resellers, direct manufacturers, large service providers, cloud providers, telecommunication companies, and to a lesser extent e-tailers and retailers. Smaller, local or regional value-added resellers typically focus on a single solution suite or portfolio of solutions from one or two vendor partners.

We believe we are well positioned to compete within this marketplace due to our competitive advantages. We expect the competitive landscape to continue to evolve as new technologies are developed. While innovation can help our business as it creates new offerings for us to sell, it can also disrupt our business model and create new and stronger competitors. For additional information on the risks associated with competition, see "Item 1A. Risk Factors."

We believe we have sustainable competitive advantages that differentiate us in the marketplace. We have built a strong sales organization and deep services and solutions capabilities over time and expect to continue to invest to enhance these capabilities, which we believe when combined with our competitive advantages of scale and a performance driven culture, will help drive sustainable, profitable growth for us today and in the future. Our scale enables us to have a national and international footprint, as well as invest in resources to meet specific customer end-market needs. Our sellers are organized around unique customer end-markets that are both vertically and geographically focused. Our scale enables our ability to invest in technical coworkers who work directly with our sellers to help customers implement increasingly complex IT solutions. Our scale also enables us to operate our three distribution centers (two in the US and one in the UK), which combined are more than 1 million square feet in size. We have cross-border relationships that enable us to serve the needs of our US, UK and Canadian-based customers in more than 150 countries. Our strong, execution-oriented culture is underpinned by our compensation system.

Our Offerings

Our offerings range from discrete hardware and software products and services to complex integrated solutions including one or more of these elements. We believe our customers increasingly view technology purchases as integrated solutions rather than discrete product and services categories. We estimate that more than 40% of our Net sales in 2021 in the US came from sales of product categories and services typically associated with solutions. Our hardware products include notebooks/mobile devices (including tablets), network communications, desktop computers, video monitors, enterprise and data storage, and other hardware. Our software products include application suites, security, virtualization, operating systems and network management. Our services include advisory and design, software development, implementation, managed services and warranties.

IT is critical to both "run the business" and drive greater growth and productivity. To help our customers accomplish this, we have built a robust portfolio of solutions across hybrid infrastructure, digital experience, security and services that we provide in physical, virtual, or cloud-based environments.

We provide customers with cloud solutions and services through public cloud solutions, which reside off customer premises on a public (shared) infrastructure, private cloud solutions, which reside on customer premises, and hybrid cloud solutions that deliver the benefits of both public and private solutions. Our migration, integration and managed services help our customers simplify cloud adoption, as well as the ongoing management of cloud solutions, across the entire IT lifecycle. Service delivery engineers work with our customers to design cloud solutions meeting their organizational, technology and financial objectives.

We offer a broad portfolio of integrated solutions that include the following on-premise, hybrid and cloud capabilities:

• Services: We help organizations design, orchestrate and manage technology for their unique needs. Our offerings are designed to highlight our expertise in the most critical technology areas for our customers. Our service delivery engineers have expertise which include integrated cloud, collaboration, data center, mobility and security business technology, from the physical to the application layer. We leverage best-in-class partner technology platforms to seamlessly architect and manage disparate IT platforms into integrated business technology solutions.

- Hybrid Infrastructure: We assess our customers application infrastructure need, design flexible, resilient and efficient solutions and manage the solution throughout its lifecycle. Our broad portfolio of hardware and software products, encompassing both on and off-premise solutions, enables us to provide well-integrated solutions, including converged and hyper-converged infrastructure, physical and virtualized servers, software defined automation and orchestration solutions, hybrid storage, energy-efficient power and cooling, and data center networking.
- Digital Experience: We build end-to-end solutions that deliver access to applications that improve our customers' productivity regardless of device or location. We connect our customers' physical devices, including laptops, desktops, IP Phones, mobile devices and print systems. We utilize collaboration solutions to unite applications via the integration of products that facilitate the use of multiple enterprise communication methods including email, persistent chat, social media, voice and video. We also host cloud-based collaboration solutions. Our solutions provide the tools that allow our customers' employees to share knowledge, ideas and information among each other and with clients and partners effectively, securely and quickly.
- Security: We assess our customers' security needs and provide them with tools and services to help effectively manage risk. We are a security solutions integrator that combines our expertise in design, solution architecture and implementation services. Our customer solutions can take the form of hardware, software or Software as a Service across a multitude of categories such as: endpoint security, email security, web security, intrusion prevention, authentication, firewall, virtual private network services and network access control. Security consulting engagements include security assessment, policy and procedure gap analysis, security roadmaps and health checks.

Although we believe customers increasingly view technology purchases as solutions rather than discrete product and service categories, our Net sales by major category, based upon our internal category classifications, was as follows

Year Ended December 31,											
	2	2021				2020					
					Dollars in Millions	Percentage of Total Net Sales				Percer of Total N	
\$	6,659.4	3	32.0 %	\$	5,486.2		29.7 %	\$	4,344.9		24.1 %
	1,950.9		9.4		1,955.0		10.6		2,189.1		12.1
	1,203.6		5.8		1,132.4		6.1		1,547.3		8.6
	1,605.0		7.7		1,190.8		6.4		1,272.9		7.1
	992.1		4.8		947.4		5.1		1,147.6		6.4
	4,358.6	2	20.9		4,121.6		22.3		3,980.4		22.1
	16,769.6	8	30.6		14,833.4		80.2		14,482.2		80.4
	2,802.4	1	3.5		2,581.0		14.0		2,585.0		14.3
	1,126.1		5.4		913.9		4.9		840.9		4.7
	122.7		0.5		139.2		0.9		124.3		0.6
\$	20,820.8	10	00.0 %	\$	18,467.5		100.0 %	\$	18,032.4		100.0 %
		Dollars in Millions \$ 6,659.4	Millions of Total Net \$ 6,659.4 3 1,950.9 1,203.6 1,605.0 992.1 4,358.6 2 16,769.6 8 2,802.4 1 1,126.1 1 122.7 1	Dollars in Millions Percentage of Total Net Sales \$ 6,659.4 32.0 % 1,950.9 9.4 1,203.6 5.8 1,605.0 7.7 992.1 4.8 4,358.6 20.9 16,769.6 80.6 2,802.4 13.5 1,126.1 5.4 122.7 0.5	Dollars in Millions Percentage of Total Net Sales \$ 6,659.4 32.0 % 1,950.9 9.4 1,203.6 5.8 1,605.0 7.7 992.1 4.8 4,358.6 20.9 16,769.6 80.6 2,802.4 13.5 1,126.1 5.4 122.7 0.5	2021 Dollars in Millions Percentage of Total Net Sales Dollars in Millions \$ 6,659.4 32.0 % \$ 5,486.2 1,950.9 9.4 1,955.0 1,203.6 5.8 1,132.4 1,605.0 7.7 1,190.8 992.1 4.8 947.4 4,358.6 20.9 4,121.6 16,769.6 80.6 14,833.4 2,802.4 13.5 2,581.0 1,126.1 5.4 913.9 122.7 0.5 139.2	Dollars in Millions Percentage of Total Net Sales Dollars in Millions Percentage of Total Net Sales \$ 6,659.4 32.0 % \$ 5,486.2 1,950.9 9.4 1,955.0 1,203.6 5.8 1,132.4 1,605.0 7.7 1,190.8 992.1 4.8 947.4 4,358.6 20.9 4,121.6 16,769.6 80.6 14,833.4 2,802.4 13.5 2,581.0 1,126.1 5.4 913.9 122.7 0.5 139.2	Dollars in Millions Percentage of Total Net Sales Dollars in Millions Percentage of Total Net Sales \$ 6,659.4 32.0 % \$ 5,486.2 29.7 % 1,950.9 9.4 1,955.0 10.6 1,203.6 5.8 1,132.4 6.1 1,605.0 7.7 1,190.8 6.4 992.1 4.8 947.4 5.1 4,358.6 20.9 4,121.6 22.3 16,769.6 80.6 14,833.4 80.2 2,802.4 13.5 2,581.0 14.0 1,126.1 5.4 913.9 4.9 122.7 0.5 139.2 0.9	Dollars in Millions Percentage of Total Net Sales Dollars in Millions Dollars in Millions Percentage of Total Net Sales Dollars in Millions Dollars in Millions Percentage of Total Net Sales Dollars in Millions Percentage of Total Net Sales Dollars in Millions Dollars in Millions	Dollars in Millions Percentage of Total Net Sales Dollars in Millions Percentage of Total Net Sales Dollars in Millions \$ 6,659.4 32.0 % \$ 5,486.2 29.7 % \$ 4,344.9 1,950.9 9.4 1,955.0 10.6 2,189.1 1,203.6 5.8 1,132.4 6.1 1,547.3 1,605.0 7.7 1,190.8 6.4 1,272.9 992.1 4.8 947.4 5.1 1,147.6 4,358.6 20.9 4,121.6 22.3 3,980.4 16,769.6 80.6 14,833.4 80.2 14,482.2 2,802.4 13.5 2,581.0 14.0 2,585.0 1,126.1 5.4 913.9 4.9 840.9 122.7 0.5 139.2 0.9 124.3	Dollars in Millions Percentage of Total Net Sales \$ 6,659.4 32.0 % \$ 5,486.2 29.7 % \$ 4,344.9 1,950.9 9.4 1,955.0 10.6 2,189.1 1,203.6 5.8 1,132.4 6.1 1,547.3 1,605.0 7.7 1,190.8 6.4 1,272.9 992.1 4.8 947.4 5.1 1,147.6 4,358.6 20.9 4,121.6 22.3 3,980.4 16,769.6 80.6 14,833.4 80.2 14,482.2 2,802.4 13.5 2,581.0 14.0 2,585.0 1,126.1 5.4 913.9 4.9 840.9 122.7 0.5 139.2 0.9 124.3

- (1) Certain software and services revenue is recorded on a net basis for accounting purposes, so the category percentage of Net sales is not representative of the category percentage of gross profits.
- (2) Includes items such as delivery charges to customers.

Our Internal Capabilities

Human Capital Management

Our culture is reflected through our coworkers, who are driven to serve our customers, our partners, our communities and all our stakeholders. We provide our coworkers with diverse experiences, engagement opportunities, strong training and development, competitive compensation and meaningful careers, which creates a high-performance culture that is central to CDW's success. We know that an inclusive environment produces the best ideas and our coworkers are driven to finding the best technology solutions to enable the mission-driven needs of our customers.

We have approximately 13,900 coworkers across the globe, with 11,500 coworkers in the US, 1,500 in the UK and 900 in Canada. More than 50% of our US Net sales are generated by account managers who have more than seven years of tenure with CDW. Our coworker relations are strong and none of our coworkers are covered by collective bargaining agreements.

Diversity, Equity and Inclusion

CDW's commitment to diversity, equity and inclusion is a core value-shaping who we are, and how we work, grow and do business. We remain steadfast in our commitment to a culture of inclusion and equity, where everyone feels they belong.

Our diversity, equity and inclusion efforts prioritize fostering an inclusive environment for coworkers and job candidates that cannot be separated from how we work with customers, partners and the community. It all comes back to our character, values and ethics as an organization. We are intent on making sure our values are not just words on a page, but spur behavior where everyone feels they are seen, heard and valued.

Coworker Engagement

We strive to create a culture of collaboration, belonging and individual growth and reward – one in which every coworker has a voice and where all voices are heard. Our coworker engagement strategy utilizes frequent, short surveys as well as virtual listening groups to gain a real-time understanding of the coworker experience at CDW. As a result of our coworkers' consistent engagement, we have garnered meaningful feedback and recommendations, which have led to measurable and impactful results.

Training & Development

We focus on skills enhancement, leadership development, innovation excellence and professional growth throughout our coworkers' careers at CDW. Our programs include: leadership development trainings, unique developmental opportunities for our high-potential emerging leaders, a 24-month training program for new North American sales coworkers, technical skill development training, an 18-month apprentice-style program for aspiring engineers, and coworker access to over 15,000 on-demand educational modules.

Total Rewards

Our Pay-for-Performance total rewards philosophy provides market competitive compensation aligned with company performance. We further align our sellers' compensation to their individual performance by providing substantially uncapped commission opportunity. We provide a comprehensive benefits package to our coworkers, including healthcare, retirement plans with profit sharing and match, tuition assistance, inclusive parental leave policies, adoption assistance, paid time off, paid volunteer hours and philanthropic match programs based upon eligibility and location.

Health and Safety

At the beginning of the pandemic, we identified three key principles, which have guided us. First, safeguard the health and well-being of our coworkers, second, serve the mission-driven needs of our customers, and third, support our communities. We implemented precautions to help keep our coworkers healthy and safe, including activating a cross-functional response team led by senior leadership, moving to remote work for our office coworkers, and implementing safety protocols at our distribution centers, including social distancing measures, segmented shifts, additional personal protective equipment, enhanced facility cleanings, expanded health and safety training, increased available mental health resources, and increased sick days for impacted coworkers.

Oversight and Management

Our Coworker Services organization is responsible for the strategy and management of coworker-related matters, working in concert with all our leaders. Our Board understands the importance of our inclusive, performance-driven culture to our ongoing

success and is actively engaged with our President and Chief Executive Officer and our Chief Human Resources Officer across a broad range of human capital management topics.

Marketing

We market the CDW brand to US, UK and Canadian audiences using a variety of channels that include online, broadcast, print, social and other media. We market to current and prospective customers through integrated marketing programs including behaviorally targeted email, print, online media, events and sponsorships, as well as broadcast media. This promotion is also supported by integrated communication efforts targeting decision-makers, influencers and the general public using a combination of news releases, case studies, media interviews and speaking opportunities.

As a result of our relationships with our vendor partners, a significant portion of our advertising and marketing expenses is reimbursed through cooperative advertising programs. These programs are at the discretion of our vendor partners and are typically tied to sales or other commitments to be met by us within a specified period of time. We believe that our results and analytical techniques that measure the efficacy of our marketing programs differentiate us from our competitors.

Information Technology Systems

We maintain customized IT and unified communication systems that enhance our ability to provide prompt, efficient and expert service to our customers. In addition, these systems enable centralized management of key functions, including purchasing, inventory management, billing and collection of accounts receivable, sales and distribution. Our systems provide us with thorough and detailed information regarding key aspects of our business. These capabilities help us to continuously enhance productivity, ship customer orders quickly and efficiently, respond appropriately to industry changes and provide high quality customer service. We believe our websites, which provide electronic order processing and advanced tools, such as order tracking, reporting and asset management, make it easy for customers to transact business with us and ultimately strengthen our customer relationships.

History

Founded in 1984, CDW became a public company in 1993. In 2006, we acquired Berbee Information Networks Corporation to expand our capabilities in customized engineering services and managed services. In 2007, we went private and then became public again in 2013.

In 2015, we acquired control of 100% of UK-based IT solutions provider, Kelway TopCo Limited. Rebranded CDW UK in 2016, the acquisition extended our footprint into the UK.

In 2019, we acquired Canada-based technology solutions provider, Scalar Decisions Inc.

CDW's AmplifiedTM Services portfolio has grown into a billion-dollar business over the past few years, aided by acquisitions of various companies. In addition to the acquisition of Sirius in 2021, an IT solutions integrator, as described above, we further strengthened our consulting and services expertise by acquiring Aptris, an IT service management solutions provider and ServiceNow Elite partner, in 2019. In 2020, we acquired IGNW, a cloud-native services, software development and data orchestration capability provider. In 2021, we acquired Amplified IT, which has expert capability in Google Workspace for Education and Focal Point Data Risk, which has expert capabilities in cybersecurity services.

Available Information

We maintain a website at www.cdw.com. You may access our Annual Reports on Form 10-K, Quarterly Reports on Form 10-Q, Current Reports on Form 8-K and amendments to those reports filed or furnished pursuant to Section 13(a) or 15(d) of the Securities Exchange Act of 1934 with the SEC free of charge at our website as soon as reasonably practicable after such material is electronically filed with, or furnished to, the SEC. Our website and the information contained on that site, or connected to that site, are not incorporated into and are not a part of this report.

Item 1A. Risk Factors

There are many factors that could adversely affect our business, results of operations and cash flows, some of which are beyond our control. The following is a description of some important factors that may cause our business prospects, results of operations and cash flows in future periods to differ materially from those currently expected or desired. Factors not currently known to us or that we currently deem to be immaterial may also materially and adversely affect our business, results of operations and cash flows.

Business and Operational Risks

The outbreak of the novel coronavirus ("COVID-19") pandemic has adversely impacted and could continue to adversely impact our business and results of operations and could also adversely impact our cash flows, financial condition and liquidity.

The global spread of COVID-19 continues to create significant macroeconomic uncertainty, volatility and disruption. Many governments and health authorities have implemented recommendations or mandates intended to slow the further spread of the disease, such as shelter-in-place orders, resulting in the temporary closure of schools and non-essential businesses, or social distancing measures, resulting in modified operations of various businesses including ours, and these measures may remain in place for a significant period of time. While some of these restrictions have been lifted or eased in certain jurisdictions, other restrictions such as vaccine mandates and testing requirements have been newly imposed, and the recovery process is uncertain. We have experienced and could continue to experience disruptions, including as a result of resurgences of COVID-19, that prevent us from meeting the demands of our customers, such as product constraints from our vendor partners and wholesale distributors and other disruptions to our supply chain, disruptions in or restrictions on the ability of our coworkers to work effectively, temporary closures of our distribution facilities, modifications in the operation of facilities that remain open and disruptions of commercial delivery services. The impact of COVID-19 and measures implemented to slow the spread have caused and could continue to cause delay in, or limit the ability of, our customers to place orders for our products and services and make timely payments to us and could materially increase our labor, logistics and other costs. As long as the pandemic continues, our coworkers will continue to be exposed to health risks, and we could be negatively impacted in the future if a significant number of our coworkers, or coworkers who perform critical functions, become unable to work as a result of exposure to COVID-19. In addition, the pandemic has resulted in a widespread health crisis that has adversely affected the economies and financial markets of many countries, including the US, the UK and Canada. During the COVID-19 pandemic and even after it has subsided, we may experience adverse impacts to our business as a result of the pandemic's global economic impact, including any recession, economic downturn or volatility, government spending cuts, tightening of credit markets or increased unemployment that has occurred or may occur in the future, which could cause our customers and potential customers to postpone or reduce spending on technology products or services or put downward pressure on prices. In addition, we may experience inflationary pressures, resulting in increased product prices that we may be unable to pass on to our customers.

Individually and collectively, the consequences of the COVID-19 pandemic have adversely impacted and could continue to adversely impact our business and results of operations and could also adversely impact our cash flows, financial condition and liquidity. The extent to which the COVID-19 pandemic continues to impact our business, results of operations, cash flows, financial condition and liquidity will depend on future developments, which are highly uncertain and cannot be predicted, including, but not limited to, the ultimate duration and severity of the pandemic, future resurgences and emergences of new variants of the virus, the availability, efficacy and acceptance of a vaccine and treatments, actions taken to contain the virus including reimplementation of closures, and the effectiveness of these actions, and how quickly and to what extent normal economic and operating conditions can resume and be sustained. The COVID-19 pandemic has and may continue to have the effect of heightening many of the other risks described in this "Risk Factors" section.

Our business depends on our vendor partner relationships and the terms of the agreements governing those relationships.

Our solutions portfolio includes products and services from OEMs, software publishers and cloud providers. We are authorized by these vendor partners to sell all or some of their products and services via direct marketing activities. Our authorization with each vendor partner is subject to specific terms and conditions regarding such things as sales channel restrictions, product return privileges, services performance commitments, price protection policies, purchase discounts and vendor partner programs and funding, including purchase rebates, sales volume rebates, purchasing incentives and cooperative advertising reimbursements. However, we do not have any long-term contracts with our vendor partners and many of these arrangements are terminable upon notice by either party. A reduction in vendor partner programs or funding or our failure to timely react to changes in vendor partner programs or funding could have an adverse effect on our business, results of operations or cash flows. In addition, a reduction in the amount or a change in the terms of credit granted to us by our vendor partners could increase our

need for, and the cost of, working capital and could have an adverse effect on our business, results of operations or cash flows, particularly given our level of indebtedness.

From time to time, vendor partners may terminate or limit our right to sell some or all of their products or change the terms and conditions or reduce or discontinue the incentives that they offer us. For example, there is no assurance that, as our vendor partners continue to sell directly to end users and through resellers, they will not limit or curtail the availability of their products to solutions providers like us. Any such termination or limitation or the implementation of such changes could have a negative impact on our business, results of operations or cash flows.

We purchase the products included in our portfolio both directly from our vendor partners and from wholesale distributors. Although we purchase from a diverse vendor base, in 2021, products we purchased from wholesale distributors Ingram Micro and TD SYNNEX, together, represented over 30% of total US purchases. In addition, sales of products manufactured by Apple, Cisco, Dell EMC, HP Inc., Lenovo and Microsoft, whether purchased directly from these vendor partners or from a wholesale distributor, represented over 50% of our 2021 consolidated Net sales. Sales of products manufactured by Dell EMC and Lenovo represented over 20% of our 2021 consolidated Net sales. The loss of, or change in business relationship with, any of these or any other wholesale distributors or key vendor partners, or the diminished availability of their products, including due to backlogs for their products, could reduce the supply and increase the cost of products we sell and negatively impact our competitive position.

Further, the sale, spin-off or combination of any of our wholesale distributors or key vendor partners and/or certain of their business units, including any such sale to or combination with a vendor with whom we do not currently have a commercial relationship or whose products we do not sell, or our inability to develop relationships with new and emerging vendors and vendors that we have not historically represented in the marketplace, could have an adverse impact on our business, results of operations or cash flows.

Our sales are dependent on continued innovations in hardware, software and services by our vendor partners and the competitiveness of their offerings, and our ability to partner with new and emerging technology providers.

The technology industry is characterized by rapid innovation and the frequent introduction of new and enhanced hardware, software and services, such as cloud-based solutions, including Software as a Service ("SaaS"), Infrastructure as a Service ("IaaS") and Platform as a Service ("PaaS"); Device as a Service ("DaaS"); the Internet of Things ("IoT"); and artificial intelligence ("AI"). We have been and will continue to be dependent on innovations in hardware, software and services, as well as the acceptance of those innovations by customers. Also, customers may delay spending while they evaluate new technologies. A decrease in the rate of innovation, a lack of acceptance of innovations by our customers or delays in technology spending by our customers, could have an adverse effect on our business, results of operations or cash flows.

In addition, if we are unable to anticipate and expand our capabilities to keep pace with changes in technology and new hardware, software and services, for example by providing the appropriate training to our account managers, technology specialists and engineers to enable them to effectively sell and deliver such new offerings to customers, our business, results of operations or cash flows could be adversely affected.

We also are dependent upon our vendor partners for the development and marketing of hardware, software and services to compete effectively with hardware, software and services of vendors whose products and services we do not currently offer or that we are not authorized to offer in one or more customer channels. To the extent that a vendor's offering that is in high demand is not available to us for resale in one or more customer channels, and there is not a competitive offering from another vendor that we are authorized to sell in such customer channels, our business, results of operations or cash flows could be adversely impacted.

Substantial competition could reduce our market share and significantly harm our financial performance.

Our current competition includes:

- resellers and service providers, such as Computacenter, Connection, ePlus, Insight Enterprises, NTT, Optiv, Presidio, SCC, Softchoice, World Wide Technology and many smaller resellers and service providers;
- manufacturers who sell directly to customers, such as Adobe, Apple, Dell EMC, HP Inc. and Hewlett Packard Enterprise;
- large service providers and system integrators, such as Accenture, Dell EMC, Hewlett Packard Enterprise and IBM;
- communications service providers, such as AT&T, CenturyLink and Verizon;

- cloud providers, such as Amazon Web Services, Google and Microsoft;
- e-tailers, such as Amazon and Newegg; and
- retailers (including their e-commerce activities), such as Office Depot and Staples.

We expect the competitive landscape to continue to evolve as new technologies and consumption models emerge, such as cloud-based and other "as a service" solutions, hyper-converged infrastructure and embedded software solutions. Our continued competitiveness depends upon our ability to anticipate and evolve at pace and scale with new technologies, services and solutions through strategic and timely investments in innovation, expansion of offerings and the capabilities necessary to implement them.

While innovation can help our business as it creates new offerings for us to sell, it can also disrupt our business model and create new and stronger competitors. For instance, while cloud-based solutions present an opportunity for us, cloud-based solutions and technology solutions as a service could increase the amount of sales directly to customers rather than through solutions providers like us, or could reduce the amount of hardware we sell. In addition, some of our hardware and software vendor partners sell, and could intensify their efforts to sell, their products directly to our customers. Moreover, traditional OEMs have increased their services capabilities through mergers and acquisitions with service providers, which could potentially increase competition in the market to provide comprehensive technology solutions to customers. If we are unable to effectively respond to the evolving competitive landscape, or respond in a manner that is less effective than that of our competitors, our business, results of operations or cash flows could be adversely impacted.

We focus on providing high quality service to gain new customers and retain existing customers. To the extent we face increased competition to gain and retain customers, we may be required to reduce prices, increase advertising expenditures or take other actions which could adversely affect our business, results of operations or cash flows. Additionally, some of our competitors may reduce their prices in an attempt to stimulate sales, which may require us to reduce prices. This would require us to sell a greater number of products to achieve the same level of Net sales and Gross profit. If such a reduction in prices occurs and we are unable to attract new customers and sell increased quantities of products, our sales growth and profitability could be adversely affected.

The success of our business depends on the continuing development, maintenance and operation of our information technology systems.

Our success is dependent on the accuracy, proper utilization and continuing operation, maintenance and development of our information technology systems, including our business systems, such as our sales, customer management, financial and accounting, marketing, purchasing, warehouse management, e-commerce and mobile systems, as well as our operational platforms, including voice and data networks and power systems. The quality and our utilization of the information generated by our information technology systems, and our success in implementing new systems and upgrades, affects, among other things, our ability to:

- conduct business with our customers, including delivering services and solutions to them;
- provide the means to effectively manage global operations across time zones;
- keep pace with changes and innovation and compete effectively;
- effectuate comprehensive and reliable data collection, maintenance and governance;
- manage our inventory, accounts receivable and accounts payable;
- support planned growth in services and solutions and continued evolution of the business;
- purchase, sell, ship and invoice our hardware and software products and provide and invoice our services efficiently and on a timely basis; and
- maintain our cost-efficient operating model while scaling our business.

The integrity of our information technology systems is vulnerable to disruption due to forces beyond our control. While we have taken steps to protect our information technology systems from a variety of threats, both internal and external, and from human error, there can be no guarantee that those steps will be effective. Furthermore, although we have redundant systems at a separate location to back up our primary systems, there can be no assurance that these redundant systems will operate properly if and when required. Any disruption to or infiltration of our information technology systems could significantly harm our reputation, business and results of operations due to failure to comply with customer, partner, legal or regulatory obligations.

Breaches of data security and the failure to protect our information technology systems from cybersecurity threats could adversely impact our business.

Our business involves the handling, storage and transmission of proprietary information and sensitive or confidential data, including personal information of coworkers, customers, partners and others. In connection with our services business, some of our coworkers have access to our customers' confidential data and other information. Additionally, third parties, such as data center colocation and hosted solution partners, provide services to us and also provide services as a component of our services delivery to customers. These third parties or others that are a part of our supply chain could also be a source of security risk in the event of a failure to protect their own products, security systems and infrastructure and we may not be able to control the manner in which these third parties respond to any security breach. We have privacy and data security policies, practices and controls in place that are designed to prevent security breaches; however, as newer technologies evolve, as more business is conducted on line and remotely, and as the portfolio of the service providers we exchange confidential information, software and/or hardware with expands, we are exposed to increased risks from breaches in security, including those arising from human error, negligence or mismanagement or from illegal or fraudulent acts, such as cyberattacks. Although we have not experienced a material security breach to date, we regularly experience malicious attacks and other attempts to gain authorized access to our systems. The evolving nature of threats to data security, in light of new and sophisticated methods used by criminals and cyberterrorists, state-sponsored organizations and nation-states, including computer viruses, malware, ransomware, phishing, misrepresentation, social engineering and forgery, make it increasingly challenging to anticipate and adequately mitigate these threats should they materialize.

Breaches in security could expose us, our supply chain, our customers or other individuals to significant disruptions and a risk of public disclosure, loss or misuse of this information. Security breaches could result in legal claims or proceedings, liability or regulatory penalties under laws protecting the privacy of personal information (including those under the European Union General Data Protection Regulation and the California Consumer Privacy Act), significant remediation costs as well as the loss of existing or potential customers and, ultimately, damage to our brand and reputation. While we maintain insurance coverages that are intended to address certain aspects of data security, such insurance may be insufficient to cover all losses or all types of claims that may arise. Moreover, media or other reports of perceived vulnerabilities in our network security or perceived lack of security within our environment, even if inaccurate, could materially adversely impact our reputation and business. The cost and operational consequences of implementing further data protection measures could also be significant. Such breaches, costs and consequences could adversely affect our business, results of operations or cash flows.

If we or our third-party service providers fail to provide high-quality services to our customers, our reputation, brand, business, results of operations or cash flows could be adversely affected.

Our services include professional services, managed services, warranties, configuration services, partner services and telecom services. Additionally, we deliver and manage mission critical software, systems and network solutions for our customers. We also offer certain services, such as implementation and installation services and repair services, to our customers through various third-party service providers engaged to perform these services on our behalf. If we or our third-party service providers fail to provide high-quality services to our customers or such services result in an unplanned disruption of our customers' businesses, this could, among other things, result in legal claims and proceedings and liability for us. Moreover, as we expand our services and solutions business and provide increasingly complex services and solutions, we may be exposed to additional operational, regulatory and other risks. We also could incur liability for failure to comply with the rules and regulations applicable to the new services and solutions we provide to our customers. If any of the foregoing were to occur, our reputation with our customers, our brand and our business, results of operations or cash flows could be adversely affected.

If we lose any of our key personnel, are unable to attract and retain the talent required for our business, our labor costs significantly increase or if our approach to workforce management is ineffective, our business could be disrupted and our financial performance could suffer.

Our success is heavily dependent upon our ability to attract, develop, engage and retain key personnel to manage, lead, innovate and grow our business, including our key executive, management, sales, services and technical coworkers. The proposed federal vaccinate mandate, along with any other vaccine requirements applicable to our coworkers, and the uncertainty and unpredictability of the COVID-19 environment, could make it more difficult to attract or retain key personnel.

Our future success will depend to a significant extent on the efforts of our leadership team, as well as the effectiveness of our succession planning and efforts to develop and promote top talent. Our future success also will depend on our ability to retain and motivate our customer-facing coworkers, who have been given critical CDW knowledge regarding, and the opportunity to develop strong relationships with, many of our customers. In addition, as we seek to expand our offerings of value-added services and solutions, our success will even more heavily depend on attracting and retaining highly skilled technology specialists and engineers, for whom the market is extremely competitive.

In order to attract, retain and motivate key personnel in a competitive marketplace, it is important to provide a competitive compensation package. If our compensation package is not viewed as being competitive, our ability to attract, retain and motivate key personnel could be adversely affected. Additionally, as minimum wage rates increase or related laws and regulations change, we have and may need to continue to increase not only the wage rates of our minimum wage coworkers, but also the wages paid to our other hourly or salaried coworkers.

We have observed an overall tightening and increasingly competitive labor market, in particular with highly skilled technology specialists and engineers. A sustained labor shortage or increased turnover rates within our coworker base, whether caused by COVID-19 or as a result of general macroeconomic factors occurring throughout the US economy, could lead to increased costs, such as increased overtime to meet demand and increased wage rates to attract and retain coworkers, and could adversely affect our business, results of operations or cash flows. Additionally, if we fail to effectively manage our workforce, we may need to terminate or reposition coworkers within our Company to eliminate an abundance of or to reconfigure resources, which could damage our coworker relations and our ability to attract and retain key personnel.

If we are unable to attract, develop, engage and retain key personnel, or if our approach to workforce management is ineffective, our relationships with our vendor partners and customers and our ability to expand our offerings of value-added services and solutions could be adversely affected. Moreover, if we are unable to continue to train our sales, services and technical personnel effectively to meet the rapidly changing technology needs of our customers, the overall quality and efficiency of such personnel could decrease. Such consequences could adversely affect our business, results of operations or cash flows.

A natural disaster or other adverse occurrence at one of our primary facilities or a third-party provider location could damage our business.

If the warehouse and distribution equipment or operations at one of our distribution centers were to be seriously damaged or disrupted by a natural disaster, which may increase in number or severity as a result of climate change, or other adverse occurrence, including disruption related to political or social unrest, we could utilize another distribution center or third-party distributors to ship products to our customers. However, this may not be sufficient to avoid interruptions in our service and may not enable us to meet all of the needs of our customers and would cause us to incur incremental operating costs. In addition, we operate numerous facilities which may contain both business-critical data and confidential information of our customers and third parties, such as data center colocation and hosted solution partners, and third-parties provide services as a component of our services delivery to customers. A natural disaster or other adverse occurrence at any of our major data storage locations or third-party provider locations could negatively impact our business, results of operations or cash flows.

Increases in the cost of commercial delivery services or disruptions of those services could materially adversely impact our business.

We generally ship hardware products to our customers by FedEx, United Parcel Service and other commercial delivery services and invoice customers for delivery charges. If we are unable to pass on to our customers future increases in the cost of commercial delivery services (including those that may result from an increase in fuel or personnel costs or a need to use higher cost delivery channels during periods of increased demand), our profitability could be adversely affected. Additionally, strikes, inclement weather, natural disasters or other service interruptions by such shippers or periods of increased demand on delivery services, such as those we have experienced during the COVID-19 pandemic, could materially adversely affect our ability to deliver or receive products on a timely basis.

We are exposed to accounts receivable and inventory risks.

We extend credit to our customers for a significant portion of our sales. We are subject to the risk that our customers may not pay for the products they have purchased, may pay at a slower rate than we have historically experienced, or may seek extended payment terms. This risk is heightened during periods of global or industry-specific economic downturn or uncertainty, during periods of rising interest rates or, in the case of public sector customers, during periods of budget constraints. Significant failures of customers to timely pay all amounts due to us could adversely affect our business, results of operations or cash flows.

We are also exposed to inventory risks as a result of the rapid technological changes that affect the market and pricing for the products we sell. In addition to drop-ship arrangements with many of our OEMs and wholesale distributors, we seek to minimize our inventory exposure through a variety of inventory management procedures and policies, including our rapid-turn inventory model, as well as vendor price protection and product return programs. However, if we were unable to maintain our rapid-turn inventory model, if there were unforeseen product developments that created more rapid obsolescence or if our vendor partners were to change their terms and conditions, our inventory risks could increase. We also from time to time take advantage of cost savings associated with certain opportunistic bulk inventory purchases offered by our vendor partners or we

may decide to carry high inventory levels of certain products that have limited or no return privileges due to customer demand or request or to manage supply chain interruptions. If we purchase inventory in anticipation of customer demand that does not materialize, or if customers reduce or delay orders, and if we were unable to return the inventory to a vendor partner, we would be exposed to an increased risk of inventory obsolescence.

Achieving the anticipated benefits of the Sirius acquisition remains subject to a number of uncertainties.

On December 1, 2021, the Company completed its acquisition of Sirius (the "Sirius Acquisition"). Risks and uncertainties associated with the integration of Sirius include, among other things, our ability to retain key personnel and maintain relationships with customers, suppliers and other third parties. Moreover, achieving the anticipated benefits of the Sirius Acquisition is subject to a number of uncertainties, including that the anticipated benefits may not be fully realized or may take longer to realize than expected, that the Sirius Acquisition may not be accretive to the extent anticipated, and that the Company's acquisition and integration of Sirius may involve unanticipated liabilities and costs. Failure to achieve the anticipated benefits of the Sirius Acquisition in the expected timeframe or at all could materially adversely affect our business, results of operations, cash flows and common stock price.

We could be exposed to additional risks if we continue to make strategic investments or acquisitions or enter into alliances.

We may continue to pursue transactions, including strategic investments, acquisitions or alliances, in an effort to extend or complement our existing business. These types of transactions involve numerous business risks, including finding suitable transaction partners and negotiating terms that are acceptable to us, the diversion of management's attention from other business concerns, extending our product or service offerings into areas in which we have limited experience, entering into new geographic markets, the potential loss of key coworkers or business relationships and successfully integrating acquired businesses. There can be no assurance that the intended benefits of our investments, acquisitions and alliances will be realized, or that those benefits will offset these numerous risks or other unforeseen factors, any of which could adversely affect our business, results of operations or cash flows.

In addition, our financial results could be adversely affected by financial adjustments required by generally accepted accounting principles in the United States of America ("US GAAP") in connection with these types of transactions, including the Sirius Acquisition, where significant goodwill or intangible assets are recorded. To the extent the value of goodwill or identifiable intangible assets becomes impaired, we may be required to incur material charges relating to the impairment of those assets.

Our future operating results may fluctuate significantly, which may result in volatility in the market price of our stock and could impact our ability to operate our business effectively.

We may experience significant variations in our future quarterly results of operations. These fluctuations may cause the market price of our common stock to be volatile and may result from many factors, including the condition of the technology industry in general, shifts in demand and pricing for hardware, software and services, the introduction of new products or upgrades. Further, if our customers' businesses are adversely affected by the impact of COVID-19, they may delay or reduce purchases from us, which could adversely affect our results of operations.

Our operating results are also highly dependent on Gross profit as a percentage of Net sales. Our Gross profit percentage fluctuates due to numerous factors, some of which may be outside of our control, including general macroeconomic conditions, such as inflation; pricing pressures; changes in product costs from our vendor partners; the availability of price protection, purchase discounts and incentive programs from our vendor partners; changes in product, order size and customer mix; the risk of some items in our inventory becoming obsolete; increases in product and delivery costs that we cannot pass on to customers; and general market and competitive conditions.

In addition, our cost structure is based, in part, on anticipated sales and gross margins. Therefore, we may not be able to adjust our cost structure quickly enough to compensate for any unexpected sales or gross margin shortfall, and any such inability could have an adverse effect on our business, results of operations or cash flows.

Fluctuations in foreign currency have an effect on our reported results of operations.

Our exposure to fluctuations in foreign currency rates results primarily from the translation exposure associated with the preparation of our Consolidated Financial Statements. While our Consolidated Financial Statements are reported in US dollars, the financial statements of our subsidiaries outside the US are prepared using the local currency as the functional currency and translated into US dollars. As a result, fluctuations in the exchange rate of the US dollar relative to the local currencies of our international subsidiaries, particularly the British pound and the Canadian dollar, could cause material fluctuations in our reported results of operations. We also have foreign currency exposure to the extent sales and purchases are not denominated in a subsidiary's functional currency, which could have an adverse effect on our business, results of operations or cash flows.

Macroeconomic and Industry Risks

Global and regional economic and political conditions may have an adverse impact on our business.

Weak or unstable economic and political conditions generally, sustained uncertainty about global economic and political conditions, government spending cuts and the impact of new government policies (including the introduction of new or increased taxes, the imposition of minimum taxes or new or increased limitations on deductions, credits or other tax benefits), or a tightening of credit markets, including as a result of the COVID-19 pandemic, could cause our customers and potential customers to postpone or reduce spending on technology products or services or put downward pressure on prices, which could have an adverse effect on our business, results of operations or cash flows. For example, there continues to be uncertainty regarding the economic and other impacts of the UK's exit from the European Union ("EU") in 2020, referred to as "Brexit". Potential adverse consequences of Brexit and the uncertainties around the UK and EU's relationship include global market uncertainty, volatility in currency exchange rates, additional costs and operational burdens associated with increased operational restrictions on imports and exports between the UK and other countries, potential adverse effects on the mobility of personnel and potentially increased regulatory complexities, each of which could have a negative impact on our business, financial condition or results of operations. We have established a presence in the Netherlands to help address future developments, as needed, for Brexit, which could add complexity to our European operations as well as result in higher costs associated with serving our customers.

The interruption of the flow of products from suppliers could disrupt our supply chain.

Our business depends on the timely supply of products in order to meet the demands of our customers. Manufacturing interruptions or delays, including as a result of the financial instability or bankruptcy of manufacturers, significant labor disputes such as strikes, natural disasters (which may increase in number or severity as a result of climate change), political or social unrest, pandemics (such as the COVID-19 pandemic) or other public health crises, or other adverse occurrences affecting any of our suppliers' facilities, could disrupt our supply chain. We have experienced and could continue to experience product constraints due to the failure of suppliers to accurately forecast customer demand, or to manufacture sufficient quantities of product to meet customer demand (including as a result of shortages of product components), among other reasons. Additionally, the relocation of key distributors utilized in our purchasing model could increase our need for, and the cost of, working capital and have an adverse effect on our business, results of operations or cash flows.

Moreover, supply chain disruptions during the COVID-19 pandemic have caused and could continue to cause us to experience more volatility in our level of inventory and delays in completion of orders and installations for our customers.

Our supply chain is also exposed to risks related to international operations. While we purchase our products primarily in the markets we serve (for example, products for US customers are sourced in the US), our vendor partners manufacture or purchase a significant portion of the products we sell outside of the US, primarily in Asia. Political, social or economic instability in Asia, or in other regions in which our vendor partners purchase or manufacture the products we sell, could cause disruptions in trade, including exports to the US. Other events related to international operations that could cause disruptions to our supply chain include:

- the imposition of additional trade law provisions or regulations, including the adoption or expansion of trade restrictions;
- the imposition of additional duties, tariffs and other charges on imports and exports, including any resulting retaliatory tariffs or charges and any reductions in the production of products subject to such tariffs and charges;
- foreign currency fluctuations; and
- restrictions on the transfer of funds.

We cannot predict whether the countries in which the products we sell, or any components of those products, are purchased or manufactured will be subject to new or additional trade restrictions or sanctions imposed by the US or foreign governments, including the likelihood, type or effect of any such restrictions. Trade restrictions, including new or increased tariffs or quotas, embargoes, sanctions, safeguards and customs restrictions against the products we sell, could increase the cost or reduce the supply of product available to us and adversely affect our business, results of operations or cash flows. In addition, our exports are subject to regulations, some of which may be inconsistent, and noncompliance with these requirements could have a negative effect on our business, results of operations or cash flows.

Our financial performance could be adversely affected by decreases in spending on technology products and services by our public and private sector customers due to, among other things, customer spending decisions and government spending policies.

Our sales are impacted by customer spending decisions on technology, including refresh decisions, customer initiatives that drive technology spending and customer budget priorities. Our sales to our public sector customers, and our other customers that do business with our public sector customers in particular, are impacted by government spending policies, budget priorities and revenue levels. An adverse change in government spending policies (such as budget cuts or limitations or temporary shutdowns of government operations), shifts in budget priorities or reductions in revenue levels, could cause our impacted public sector customers or our other customers that do business with impacted public sector customers to reduce or delay their purchases or to terminate or not renew their contracts with us, which could adversely affect our business, results of operations or cash flows. Additionally, such adverse change in government spending policies, shifts in budget priorities or reductions in revenue levels could impact cash collections from contracts with our impacted public sector customers or other customers that do business with impacted public sector customers, which could adversely affect our business, results of operations or cash flows.

Legal and Regulatory Risks

The failure to comply with our public sector contracts or applicable laws and regulations could result in, among other things, termination, fines or other liabilities, and changes in procurement regulations could adversely impact our business, results of operations or cash flows.

Revenues from our public sector customers are derived from sales to governmental entities, educational institutions and healthcare customers through various contracts and open market sales of products and services. Sales to public sector customers are highly regulated and present risks and challenges not present in private commercial agreements. Noncompliance with contract provisions, government procurement regulations or other applicable laws or regulations (including the False Claims Act, the Medicare and Medicaid Anti-Kickback Statute or similar laws of the jurisdictions for our business activities outside of the US) or security clearance and confidentiality requirements could result in civil, criminal and administrative liability, including substantial monetary fines or damages, termination of government contracts or other public sector customer contracts, and suspension, debarment or ineligibility from doing business with governmental entities or other customers in the public sector. In addition, contracts in the public sector are generally terminable at any time for convenience of the contracting agency or group purchasing organization ("GPO") or upon default and public sector contracts may be subject to periodic funding approval, rejections or delays, which could adversely impact public sector demand for our products and services. Furthermore, our inability to enter into or retain contracts with GPOs may threaten our ability to sell to customers in those GPOs and compete effectively. The effect of any of these possible actions or failures could adversely affect our business, results of operations or cash flows. In addition, the adoption of new or modified procurement regulations and other requirements may increase our compliance costs and reduce our gross margins, which could have a negative effect on our business, results of operations or cash flows.

We are exposed to risks from legal proceedings and audits, including intellectual property infringement claims, which may result in substantial costs and expenses or interruption of our normal business operations.

We are party to various legal proceedings that arise in the ordinary course of our business, which include commercial, employment, tort and other litigation.

We are also subject to intellectual property infringement claims against us in the ordinary course of our business, either because of the products and services we sell or the business systems and processes we use to sell such products and services, in the form of cease-and-desist letters, licensing inquiries, lawsuits and other communications and demands. In our industry, such intellectual property claims have become more frequent as the complexity of technological products and the intensity of competition in our industry have increased. Increasingly, many of these assertions are brought by non-practicing entities whose principal business model is to secure patent licensing revenue, but we may also be subject to demands from inventors, competitors or other patent holders who may seek licensing revenue, lost profits and/or an injunction preventing us from engaging in certain activities, including selling certain products or services.

In addition, we are subject to proceedings, investigations and audits by federal, state, international, national, provincial and local authorities, including as a result of our significant sales to governmental entities. For example, a subsidiary of the Company received a Civil Investigative Demand dated September 20, 2021 from the US Department of Justice ("DOJ") in connection with a False Claims Act investigation. The DOJ has requested information related to teaming agreements with OEMs.

We also are subject to audits by various partners, group purchasing organizations and customers, including government agencies, relating to purchases and sales under various contracts. In addition, we are subject to indemnification claims under various contracts.

Current and future litigation, infringement claims, governmental proceedings and investigations, audits or indemnification claims that we face may result in substantial costs and expenses and significantly divert the attention of our management regardless of the outcome. In addition, these matters could lead to increased costs or interruptions of our normal business operations. Litigation, infringement claims, governmental proceedings and investigations, audits or indemnification claims involve uncertainties and the eventual outcome of any such matter could adversely affect our business, results of operations or cash flows.

Failure to comply with complex and evolving laws and regulations applicable to our operations or failure to meet stakeholder expectations on environmental sustainability and corporate responsibility matters could adversely affect our business, results of operations or cash flows.

Our global operations span a variety of legal regimes, subjecting us to numerous complex, diverse, evolving and at times potentially inconsistent laws and regulations in a number of areas, including labor and employment, advertising, e-commerce, tax, trade, import and export controls, economic and trade sanctions, anti-corruption, data privacy and security requirements, competition, climate, environmental and health and safety. The evaluation of and compliance with these laws, regulations and similar requirements may be onerous and expensive, and may have other adverse impacts on our business, results of operations or cash flows, the risk of which will be heightened as we expand the products and services we offer, expand into new markets and channels and expand internationally. For example, we may be subject to increased costs and use of operational resources associated with complying with any new climate-related laws and regulations. Additionally, the hardware, software and services we offer increasingly utilize new and evolving technologies such as artificial intelligence ("AI"), which presents risks and challenges that could result in legal liability.

We have implemented policies and procedures designed to help ensure compliance with applicable laws and regulations, but there can be no guarantee against coworkers, contractors or agents violating such laws and regulations or our policies and procedures. Additionally, there is increased focus by stakeholders on environmental sustainability and corporate responsibility matters, including climate change response, packaging and waste reduction, energy consumption, and diversity, equity and inclusion. Our disclosure on these matters and our failure, or perceived failure, to meet our commitments or otherwise effectively address these matters may erode customer trust or confidence, particularly if they receive considerable publicity or result in litigation, and could have a negative impact on our business.

As a public company, we also are subject to increasingly complex public disclosure, corporate governance and accounting requirements that increase compliance costs and require significant management focus.

Risks Related to Our Indebtedness

Our level of indebtedness could adversely affect our business.

As of December 31, 2021, we had \$6.9 billion of total debt outstanding and \$448 million of obligations outstanding under our inventory financing agreements, and the ability to borrow an additional \$1.0 billion under our senior unsecured revolving loan facility (the "Revolving Loan Facility"). Our level of indebtedness could have important consequences, including the following:

- making it more difficult for us to satisfy our obligations with respect to our indebtedness;
- requiring us to dedicate a substantial portion of our cash flow from operations to debt service payments on our and our subsidiaries' debt, which reduces the funds available for working capital, capital expenditures, acquisitions and other general corporate purposes;
- requiring us to comply with restrictive covenants in our senior credit facilities and indentures, which limit the manner in which we conduct our business;
- making it more difficult for us to obtain vendor financing from our vendor partners, including original equipment manufacturers and software publishers;
- limiting our flexibility in planning for, or reacting to, changes in the industry in which we operate;
- placing us at a competitive disadvantage compared to any of our less-leveraged competitors;
- increasing our vulnerability to both general and industry-specific adverse economic conditions; and

limiting our ability to obtain additional debt or equity financing to fund future working capital, capital expenditures, acquisitions or other general corporate requirements and increasing our cost of borrowing.

Restrictive covenants under our senior credit facilities and, to a lesser degree, our indentures may adversely affect our operations and liquidity.

Our senior credit facilities and, to a lesser degree, our indentures contain, and any future indebtedness of ours may contain, various covenants that limit our ability to, among other things:

- incur or guarantee additional debt;
- receive dividends or other payments from our subsidiaries;
- enter into transactions with affiliates;
- pledge our assets as collateral;
- merge or consolidate with other companies or transfer all or substantially all of our assets; and
- engage in sale leaseback transactions.

As a result of these covenants, we are limited in the manner in which we conduct our business and we may be unable to engage in favorable business activities or finance future operations or capital needs. A breach of any of these covenants or any of the other restrictive covenants would result in a default under our senior credit facilities. Upon the occurrence of an event of default under our senior credit facilities, the lenders:

- will not be required to lend any additional amounts to us;
- could elect to declare all borrowings outstanding thereunder, together with accrued and unpaid interest and fees, to be due and payable; or
- could require us to apply all of our available cash to repay these borrowings.

The acceleration of amounts outstanding under our senior credit facilities would likely trigger an event of default under our existing indentures.

If the lenders under our senior credit facilities accelerate the repayment of borrowings, we may not have sufficient assets to repay our senior credit facilities and our other indebtedness or the ability to borrow sufficient funds to refinance such indebtedness. Even if we were able to obtain new financing, it may not be on commercially reasonable terms, or terms that are acceptable to us.

We will be required to generate sufficient cash to service our indebtedness and, if not successful, we may be forced to take other actions to satisfy our obligations under our indebtedness.

Our ability to make scheduled payments on or to refinance our debt obligations depends on our financial and operating performance, which is subject to prevailing economic and competitive conditions and to certain financial, business and other factors beyond our control. Our outstanding long-term debt will impose significant cash interest payment obligations on us and, accordingly, we will have to generate significant cash flow from operating activities to fund our debt service obligations. We cannot assure you that we will maintain a level of cash flows from operating activities sufficient to permit us to pay the principal, premium, if any, and interest on our indebtedness. See "Management's Discussion and Analysis of Financial Condition and Results of Operations-Liquidity and Capital Resources" included elsewhere in this report.

If our cash flows and capital resources are insufficient to fund our debt service obligations, we may be forced to reduce or delay capital expenditures, sell assets or operations, seek additional debt or equity capital, restructure or refinance our indebtedness, or revise or delay our strategic plan. We cannot assure you that we would be able to take any of these actions on terms that are favorable to us or at all, that these actions would be successful and permit us to meet our scheduled debt service obligations or satisfy our capital requirements, or that these actions would be permitted under the terms of our existing or future debt agreements, including our senior credit facilities and indentures. In the absence of such operating results and resources, we could face substantial liquidity problems and might be required to dispose of material assets or operations to meet our debt service and other obligations. We may not be able to consummate those dispositions or to obtain the proceeds which we could realize from them and these proceeds may not be adequate to meet any debt service obligations then due.

Table of Contents

In addition, major debt rating agencies regularly evaluate our debt based on a number of factors. We may not be able to maintain our existing ratings, and the failure to do so could increase the cost of servicing certain of our existing indebtedness, and make it more difficult to raise debt financing on favorable terms in the future

If we cannot make scheduled payments on our debt, we will be in default and, as a result:

- our debt holders could declare all outstanding principal and interest to be due and payable;
- the lenders under our Revolving Loan Facility could terminate their commitments to lend us money; and
- we could be forced into bankruptcy or liquidation.

We and our subsidiaries may be able to incur substantially more debt, including secured debt. This could further increase the risks associated with our leverage.

We and our subsidiaries may be able to incur substantial additional indebtedness in the future. The terms of our senior credit facilities and indentures do not fully prohibit us or our subsidiaries from doing so. To the extent that we incur additional indebtedness, the risks associated with our level of indebtedness described above, including our possible inability to service our debt, will increase. As of December 31, 2021, we had \$1.0 billion available for additional borrowing under our Revolving Loan Facility.

Variable rate indebtedness subjects us to interest rate risk, which could cause our debt service obligations to increase significantly.

Certain of our borrowings, primarily borrowings under our senior credit facilities, are at variable rates of interest and expose us to interest rate risk. As of December 31, 2021, we had \$1.7 billion of variable rate debt outstanding. If interest rates increase, our debt service obligations on the variable rate indebtedness would increase even though the amount borrowed remained the same, and our net income would decrease. Although we have entered into interest rate cap agreements on our term loan facility to reduce interest rate volatility, we cannot assure you we will be able to enter into interest rate cap agreements in the future on acceptable terms or that such caps or the caps we have in place now will be effective.

The London Inter-bank Offered Rate ("LIBOR") is being discontinued as a floating rate benchmark, which may cause interest rates under our current or future debt agreements to perform differently than in the past or cause other unanticipated consequences.

Certain of our credit facilities, including our term loan facility and our Revolving Loan Facility, have variable interest rates using LIBOR as a benchmark rate, and we have entered into interest rate cap agreements with respect to the term loan facility that are based on LIBOR. As of December 31, 2021, \$1.7 billion of our total debt outstanding bears interest at variable interest rates using LIBOR as a benchmark rate. The LIBOR and certain other interest "benchmarks" are subject to regulatory guidance and/or reform that could cause interest rates under our current or future debt agreements to perform differently than in the past or cause other unanticipated consequences. The United Kingdom's Financial Conduct Authority, which regulates the LIBOR administrator, previously announced that all LIBOR settings will either cease to be provided or no longer be representative (i) after December 31, 2021, in the case of the one-week and two-month US dollar LIBOR tenors and all tenors of non-US dollar LIBOR, and (ii) after June 30, 2023, in the case of the overnight and one-, three-, six-, and 12-month US dollar LIBOR tenors. Additionally, the US Federal Reserve, in conjunction with the Alternative Reference Rates Committee, a steering committee comprised of large US financial institutions, announced the replacement of US dollar LIBOR with a new index calculated by short-term repurchase agreements, backed by US Treasury securities, called the Secured Overnight Financing Rate ("SOFR"). SOFR has a limited history, having been first published in April 2018. The future performance of SOFR, and SOFR-based reference rates, cannot be predicted based on SOFR's history or otherwise. Future levels of SOFR may bear little or no relation to historical levels of SOFR, LIBOR or other rates. If LIBOR ceases to exist, interest rates on our current or future debt obligations and hedging instruments may be adversely affected and we may need to renegotiate the agreements governing such obligations or instruments. Although the agreements governing our senior credit facilities contain provisions for transition to new "benchmark" rates if LIBOR is discontinued or cannot be determined, any new "benchmark" may perform differently than LIBOR or cause other unanticipated consequences, which could adversely affect our interest expense, related debt obligations and our interest rate cap agreements.

Risks Related to Ownership of Our Common Stock

Our common stock price may be volatile and may decline regardless of our operating performance, and holders of our common stock could lose a significant portion of their investment.

The market price for our common stock may be volatile. Our stockholders may not be able to resell their shares of common stock at or above the price at which they purchased such shares, due to fluctuations in the market price of our common stock, which may be caused by a number of factors, many of which we cannot control, including the risk factors described in this Annual Report on Form 10-K and the following:

- changes in financial estimates by any securities analysts who follow our common stock, our failure to meet these estimates or failure of securities analysts
 to maintain coverage of our common stock;
- downgrades by any securities analysts who follow our common stock;
- future sales of our common stock by our officers, directors and significant stockholders;
- market conditions or trends in our industry or the economy as a whole;
- investors' perceptions of our prospects;
- announcements by us or our competitors of significant contracts, acquisitions, joint ventures or capital commitments; and
- changes in key personnel.

In addition, the stock markets have experienced extreme price and volume fluctuations that have affected and continue to affect the market prices of equity securities of many companies, including companies in our industry. In the past, securities class action litigation has followed periods of market volatility. If we were involved in securities litigation, we could incur substantial costs, and our resources and the attention of management could be diverted from our business.

In the future, we may also issue our securities in connection with investments or acquisitions. The number of shares of our common stock issued in connection with an investment or acquisition could constitute a material portion of our then-outstanding shares of our common stock and depress our stock price.

Anti-takeover provisions in our charter documents and Delaware law might discourage or delay acquisition attempts for us that may be considered favorable.

Our amended and restated certificate of incorporation and amended and restated bylaws contain provisions that may make the acquisition of the Company more difficult without the approval of our Board of Directors. These provisions:

- authorize the issuance of undesignated preferred stock, the terms of which may be established and the shares of which may be issued without stockholder approval, and which may include super voting, special approval, dividend, or other rights or preferences superior to the rights of the holders of common stock;
- generally prohibit stockholder action by written consent, requiring all stockholder actions be taken at a meeting of our stockholders;
- provide that special meetings of the stockholders can only be called by or at the direction of our Board of Directors pursuant to a written resolution adopted by the affirmative vote of the majority of the total number of directors that the Company would have if there were no vacancies;
- establish advance notice requirements for nominations for elections to our Board of Directors or for proposing matters that can be acted upon by stockholder at stockholder meetings; and
- provide that our Board of Directors is expressly authorized to make, alter or repeal our amended and restated bylaws.

In addition, we are subject to Section 203 of the Delaware General Corporation Law, which will prevent us from engaging in a business combination with a person who acquires at least 15% of our common stock for a period of three years from the date such person acquired such common stock, unless Board or stockholder approval is obtained prior to the acquisition. These anti-takeover provisions and other provisions under Delaware law could discourage, delay or prevent a transaction involving a change in control of the Company, even if doing so would benefit our stockholders. These provisions could also discourage

proxy contests and make it more difficult for our stockholders to elect directors of their choosing and to cause us to take other corporate actions our stockholders desire.

We cannot assure you that we will continue to pay dividends on our common stock or repurchase any of our common stock under our share repurchase program, and our indebtedness and certain tax considerations could limit our ability to continue to pay dividends on, or make share repurchases of, our common stock. If we do not continue to pay dividends, you may not receive any return on investment unless you are able to sell your common stock for a price greater than your purchase price.

We expect to continue to pay a cash dividend on our common stock. However, any determination to pay dividends in the future will be at the discretion of our Board of Directors. Any determination to pay dividends on, or repurchase, shares of our common stock in the future will depend upon our results of operations, financial condition, business prospects, capital requirements, contractual restrictions, any potential indebtedness we may incur, our target leverage ratio, restrictions imposed by applicable law, tax considerations and other factors our Board of Directors deems relevant. In addition, our ability to pay dividends on, or repurchase, shares of our common stock will be limited by restrictions on our ability to pay dividends or make distributions to our stockholders and on the ability of our subsidiaries to pay dividends or make distributions to us, in each case, under the terms of our current and any future agreements governing our indebtedness. There can be no assurance that we will continue to pay a dividend at the current rate or at all or that we will continue to repurchase shares of our common stock. If we do not pay dividends in the future, realization of a gain on your investment will depend entirely on the appreciation of the price of our common stock, which may never occur.

We are a holding company and rely on dividends, distributions and other payments, advances and transfers of funds from our subsidiaries to meet our obligations.

We are a holding company that does not conduct any business operations of our own. As a result, we are largely dependent upon cash dividends and distributions and other transfers from our subsidiaries to meet our obligations. The agreements governing the indebtedness of our subsidiaries impose restrictions on our subsidiaries' ability to pay dividends or other distributions to us. The deterioration of the earnings from, or other available assets of, our subsidiaries for any reason could also limit or impair their ability to pay dividends or other distributions to us.

Item 1B. Unresolved Staff Comments

None.

Item 2. Properties

As of December 31, 2021, we owned or leased a total of 2.6 million square feet of space, primarily in the US, UK and Canada. We own two properties: a 513,000 square foot distribution center in North Las Vegas, Nevada, and a combined office and a 442,000 square foot distribution center in Vernon Hills, Illinois. In addition, we conduct sales, services and administrative activities in various locations primarily in the US, UK and Canada.

We believe our facilities are well maintained, suitable for our business and occupy sufficient space to meet our operating needs. As part of our normal business, we regularly evaluate sales center performance and site suitability. Leases covering our currently occupied leased properties expire at varying dates, all within the next 15 years.

We anticipate no difficulty in retaining occupancy through lease renewals, month-to-month occupancy or replacing the leased properties with equivalent properties. We believe that suitable additional or substitute leased properties will be available as required.

Item 3. Legal Proceedings

We are party to various legal proceedings that arise in the ordinary course of our business, which include commercial, intellectual property, employment, tort and other litigation matters. For additional information regarding legal proceedings, refer to Note 16 (Commitments and Contingencies) to the accompanying Consolidated Financial Statements.

Item 4. Mine Safety Disclosures

Not applicable.

Information about our Executive Officers

The following table lists the name, age as of February 28, 2022 and positions of each executive officer of the Company.

Name	Age	Position
Christine A. Leahy	57	President and Chief Executive Officer and member of our Board of Directors since January 2019; Chief Revenue Officer from July 2017 to December 2018; Senior Vice President - International, Chief Legal Officer, and Corporate Secretary from May 2016 to July 2017; Senior Vice President, General Counsel and Corporate Secretary from January 2007 to May 2016.
Sona Chawla	54	Chief Growth and Innovation Officer since January 2020; President, Kohl's Corporation (an omnichannel retailer) from May 2018 to October 2019 and Chief Operating Officer from November 2015 to May 2018.
Elizabeth H. Connelly	57	Chief Human Resources Officer and Senior Vice President, Coworker Services since December 2018; Managing Director and Head, Commercial Bank Healthcare, Higher Education and Not-for-Profit Banking at J.P. Morgan Chase & Company (a global financial services firm) from March 2012 to December 2018.
Christina M. Corley	54	Chief Commercial and Operating Officer since January 2020; Chief Operating Officer from January 2019 to January 2020; Senior Vice President, Commercial and International Markets from July 2017 to December 2018; Senior Vice President, Corporate Sales from September 2011 to July 2017.
Albert J. Miralles	52	Senior Vice President and Chief Financial Officer since September 2021; Executive Vice President and Chief Financial Officer, CNA Financial Corporation (a commercial property and casualty insurance company) from February 2020 to September 2021; President, CNA Warranty from October 2019 to September 2021; Executive Vice President and Chief Risk Officer of the CNA Insurance Companies from January 2018 to October 2019; President, Long-Term Care of the CNA Insurance Companies from March 2014 to December 2017.
Frederick J. Kulevich	56	Senior Vice President, General Counsel and Corporate Secretary since October 2017; Vice President and Deputy General Counsel from May 2016 to October 2017; Vice President and Assistant General Counsel from May 2014 to May 2016; Senior Director, Ethics and Compliance from July 2006 to May 2014.

PART II

Item 5. Market for Registrant's Common Equity, Related Stockholder Matters and Issuer Purchases of Equity Securities

Market Information

Our common stock has been listed on the Nasdaq Global Select Market since June 27, 2013 under the symbol "CDW."

Holders

As of February 24, 2022, there were 7 holders of record of our common stock. The number of beneficial stockholders is substantially greater than the number of holders of record because a portion of our common stock is held through brokerage firms.

Dividends

On February 9, 2022, we announced that our Board of Directors declared a quarterly cash dividend on our common stock of \$0.50 per share. The dividend will be paid on March 10, 2022 to all stockholders of record as of the close of business on February 25, 2022.

We expect to continue to pay quarterly cash dividends on our common stock in the future, but such payments remain at the discretion of our Board of Directors and will depend upon our results of operations, financial condition, business prospects, capital requirements, contractual restrictions, any potential indebtedness we may incur, restrictions imposed by applicable law, tax considerations and other factors that our Board of Directors deems relevant. In addition, our ability to pay dividends on our common stock will be limited by restrictions on our ability to pay dividends or make distributions to our stockholders and on the ability of our subsidiaries to pay dividends or make distributions to us, in each case, under the terms of our current and any future agreements governing our indebtedness. For additional information on our cash resources and needs and restrictions on our ability to pay dividends, see "Management's Discussion and Analysis of Financial Condition and Results of Operations—Liquidity and Capital Resources" included elsewhere in this report. For additional information on restrictions on our ability to pay dividends, see Note 9 (Debt) to the accompanying Consolidated Financial Statements.

Issuer Purchases of Equity Securities

On February 10, 2021, we announced that our Board of Directors authorized a \$1.25 billion increase to our share repurchase program under which we may repurchase shares of our common stock in the open market through privately negotiated or other transactions, depending on share price, market conditions and other factors

Information relating to the Company's purchases of its common stock during the quarter ended December 31, 2021 is as follows:

Period	Total Number of Shares Purchased (in millions)	verage Price Paid per Share	Total Number of Shares Purchased as Part of a Publicly Announced Program (in millions)	Maximum Dollar Value of Shares that May Yet be Purchased Under the Program ⁽¹⁾ (in millions)		
October 1 through October 31, 2021	0.6	\$	182.78	0.6	\$ 292.4	
November 1 through November 30, 2021	0.6	\$	191.15	0.6	\$ 182.3	
December 1 through December 31, 2021	0.5	\$	193.61	0.5	\$ 87.6	
Total	1.7			1.7		

(1) The amounts presented in this column are the remaining total authorized value to be spent after each month's repurchases.

Cumulative Total Shareholder Return

The information contained in this Cumulative Total Shareholder Return section shall not be deemed to be "soliciting material" or "filed" or incorporated by reference in future filings with the SEC, or subject to the liabilities of Section 18 of the Securities Exchange Act of 1934, except to the extent that we specifically incorporate it by reference into a document filed under the Securities Act of 1933 or the Securities Exchange Act of 1934.

The following graph compares the cumulative total shareholder return, calculated on a dividend reinvested basis, on \$100.00 invested at the closing of the market on December 31, 2016 through and including the market close on December 31, 2021, with the cumulative total return for the same time period of the same amount invested in the S&P 500 Index and a peer group

index. Our peer group index for 2021 consists of the following companies: Accenture plc, Arrow Electronics, Inc., Avnet, Inc., Best Buy Company, Inc., CGI Group Inc., Cognizant Technology Solutions Corporation, DXC Technology Company, Flex Ltd., Genuine Parts Company, Henry Schein, Inc., Hewlett Packard Enterprise Company, Insight Enterprises, Inc., Jabil, Inc., LKQ Corporation, TD SYNNEX Corporation, W.W. Grainger, Inc. and Wesco International, Inc. This peer group was selected based on a review of publicly available information about these companies and our determination that they met one or more of the following criteria: (i) similar size in terms of revenue and/or enterprise value (one-third to three times our revenue or enterprise value); (ii) operates in a business-to-business distribution environment; (iii) members of the technology industry; (iv) similar customers (*i.e.*, business, government, healthcare, and education); (v) companies that provide services and/or solutions; (vi) similar margins; (vii) comparable percentage of international sales; (viii) frequently identified as a peer by the other peer companies or Institutional Shareholder Services Inc.; or (ix) identified by the Company as a competitor.

The cumulative total shareholder returns over the indicated period are based on historical data and should not be considered indicative of future shareholder returns.





	December 31, 2016				December 31, 2018		December 31, 2019		December 31, 2020		December 31, 202	
CDW Corp	\$	100	\$	135	\$	159	\$	284	\$	265	\$	416
S&P 500 Index	\$	100	\$	119	\$	112	\$	144	\$	168	\$	213
CDW Peers	\$	100	\$	117	\$	100	\$	133	\$	146	\$	196

Recent Sales of Unregistered Securities

None.

Item 6. [RESERVED]

Item 7. Management's Discussion and Analysis of Financial Condition and Results of Operations

Unless otherwise indicated or the context otherwise requires, as used in this "Management's Discussion and Analysis of Financial Condition and Results of Operations," the terms "we," "us," "the Company," "our," "CDW" and similar terms refer to CDW Corporation and its subsidiaries. "Management's Discussion and Analysis of Financial Condition and Results of Operations" should be read in conjunction with the Consolidated Financial Statements and the related notes included elsewhere in this report. This discussion contains forward-looking statements that are subject to numerous risks and uncertainties. Actual results may differ materially from those contained in any forward-looking statements. See "Forward-Looking Statements" above.

Overview

CDW Corporation, a Fortune 500 company and member of the S&P 500 Index, is a leading multi-brand provider of information technology ("IT") solutions to small, medium and large business, government, education and healthcare customers in the US, the UK and Canada. Our broad array of offerings ranges from discrete hardware and software products to integrated IT solutions and services that include on-premise, hybrid and cloud capabilities across hybrid infrastructure, digital experience and security.

We are vendor, technology, and consumption model "agnostic", with a solutions portfolio including more than 100,000 products and services from more than 1,000 leading and emerging brands. Our solutions are delivered in physical, virtual and cloud-based environments through approximately 9,900 customer-facing coworkers, including sellers, highly-skilled technology specialists and advanced service delivery engineers. We are a leading sales channel partner for many original equipment manufacturers ("OEMs"), software publishers and cloud providers (collectively, our "vendor partners"), whose products we sell or include in the solutions we offer. We provide our vendor partners with a cost-effective way to reach customers and deliver a consistent brand experience through our established end-market coverage, technical expertise and extensive customer access.

On December 1, 2021, we completed the acquisition of Sirius Computer Solutions, Inc. ("Sirius"). The aggregate consideration paid, net of cash acquired, at the closing of the acquisition was approximately \$2.4 billion, which is subject to the finalization of customary closing adjustments. Sirius is a leading provider of secure, mission-critical technology-based solutions and is one of the largest IT solutions integrators in the United States, leveraging its services-led approach, broad portfolio of hybrid infrastructure solutions, and deep technical expertise of its 2,600 coworkers to support corporate and public customers. This strategic acquisition will enhance our breadth and depth of services and solutions offerings.

We have three reportable segments, Corporate, Small Business and Public. Our Corporate segment primarily serves US private sector business customers with more than 250 employees. Our Small Business segment primarily serves US private sector business customers with up to 250 employees. Our Public segment is comprised of government agencies and education and healthcare institutions in the US. We also have two other operating segments: CDW UK and CDW Canada, each of which do not meet the reportable segment quantitative thresholds and, accordingly, are included in an all other category ("Other"). The financial results of Sirius have been included in our Consolidated Financial Statements and the results of our Corporate, Small Business and Public segments since the date of the acquisition.

We may sell all or only select products that our vendor partners offer. Each vendor partner agreement provides for specific terms and conditions, which may include one or more of the following: product return privileges, price protection policies, purchase discounts and vendor incentive programs, such as purchase or sales rebates and cooperative advertising reimbursements. We also resell software for major software publishers. Our agreements with software publishers allow the end-user customer to acquire software or licensed products and services. In addition to helping our customers determine the best software solutions for their needs, we help them manage their software agreements, including warranties and renewals. A significant portion of our advertising and marketing expenses are reimbursed through cooperative advertising programs with our vendor partners. These programs are at the discretion of our vendor partners and are typically tied to sales or other commitments to be met by us within a specified period of time.

For a discussion of results for the year ended December 31, 2020, see "Item 7. Management's Discussion and Analysis of Financial Condition and Results of Operations" of our Annual Report on Form 10-K for the year ended December 31, 2020, filed with the Securities and Exchange Commission on February 26, 2021.

Trends and Key Factors Affecting our Financial Performance

We believe the following key factors may have a meaningful impact on our business performance, influencing our ability to generate sales and achieve our targeted financial and operating results:

- General economic conditions are a key factor affecting our results as they impact our customers' willingness to spend on information technology. This is
 particularly the case for our Corporate and Small Business customers, as their purchases tend to reflect confidence in their business prospects, which are
 driven by their discrete perceptions of business and general economic conditions. Additionally, changes in trade policy and product constraints from
 suppliers could have an adverse impact on our business.
- The global spread of the novel coronavirus ("COVID-19") pandemic continues to create macroeconomic uncertainty, volatility and disruption, including supply constraints. The supply constraints are being caused by component availability and labor and logistical disruptions, resulting in extended lead times, unpredictability and higher costs. In 2021, customer top priorities have been digital transformation, security, hybrid and cloud solutions, client devices, and preparing for workers to return to the office and enhancing remote enablement capabilities as hybrid environments become the future work model. We have orchestrated solutions by leveraging client devices, accessories, collaboration tools, security, software and hybrid and cloud offerings to help customers build these capabilities and achieve their objectives.
- Changes in spending policies, budget priorities and funding levels, including current and future stimulus packages, are key factors influencing the purchasing levels of Government, Healthcare and Education customers. In 2021, Education customers continued to prioritize investments towards equity and access for all students and enhancing the in-classroom and hybrid experiences. In addition, Healthcare customers resumed projects that were paused during the pandemic as budget certainty improved as more patients returned to elective procedures. Government customers focused on multiyear budget planning and had contracting delays in several large contracts. As the duration and ongoing economic impacts of the COVID-19 pandemic remain uncertain, current and future budget priorities and funding levels for Government, Healthcare and Education customers may be adversely affected.
- Technology trends drive customer purchasing behaviors in the market. Current technology trends are focused on delivering greater flexibility and efficiency, as well as designing IT securely. These trends are driving customer adoption of solutions such as those delivered via cloud, software defined architectures and hybrid on-premise and off-premise combinations, as well as the evolution of the IT consumption model to more "as a service" offerings, including Device as a Service and managed services. Technology trends could also change as customers consider the impact of the COVID-19 pandemic on their operations.

Key Business Metrics

We monitor a number of financial and non-financial measures and ratios on a regular basis in order to track the progress of our business and make adjustments as necessary. We believe that the most important of these measures and ratios include average daily sales, gross margin, operating margin, Net income, Non-GAAP operating income, Non-GAAP operating income margin, Non-GAAP income before income taxes, Non-GAAP net income, Net sales growth on a constant currency basis, Net income per diluted share, Non-GAAP net income per diluted share, free cash flow, return on working capital, Cash and cash equivalents, net working capital, cash conversion cycle and debt levels including available credit. These measures and ratios are closely monitored by management, so that actions can be taken, as necessary, in order to achieve set standards and objectives.

In this section, we discuss Non-GAAP operating income, Non-GAAP operating income margin, Non-GAAP income before income taxes, Non-GAAP net income and Net sales growth on a constant currency basis, which are non-GAAP financial measures.

We believe these measures provide analysts, investors and management with helpful information regarding the underlying operating performance of our business, as they remove the impact of items that management believes are not reflective of underlying operating performance. Management uses these measures to evaluate period-over-period performance as management believes they provide a more comparable measure of the underlying business. Certain non-GAAP financial measures are also used to determine certain components of performance-based compensation. For the definitions of Non-GAAP operating income, Non-GAAP operating income margin, Non-GAAP income before income taxes, Non-GAAP net income and Net sales growth on a constant currency basis and reconciliations to the most directly comparable US GAAP measure, see "Results of Operations - Non-GAAP Financial Measure Reconciliations."

Table of Contents

The results of certain key business metrics are as follows:

		Year Ended December 31,						
(dollars in millions)	2021	2020	2019					
Net sales	\$ 20,820.8	\$ 18,467.5	\$ 18,032.4					
Gross profit	3,568.5	3,210.1	3,039.9					
Operating income	1,419.0	1,179.2	1,133.6					
Net income	988.6	788.5	736.8					
Non-GAAP operating income	1,645.4	1,404.6	1,368.4					
Non-GAAP net income	1,118.9	954.4	902.1					
Average daily sales ⁽¹⁾	82.0	72.7	71.0					
Net debt ⁽²⁾	6,600.4	2,517.0	3,163.3					
Cash conversion cycle (in days) ⁽³⁾	24	17	18					

- (1) There were 254 selling days for each of the years ended December 31, 2021, 2020, and 2019.
- (2) Defined as Total debt minus Cash and cash equivalents.
- (3) Cash conversion cycle is defined as days of sales outstanding in Accounts receivable and certain receivables due from vendors plus days of supply in Merchandise inventory minus days of purchases outstanding in Accounts payable and Accounts payable-inventory financing, based on a rolling three-month average.

Results of Operations

Results of operations, in dollars and as a percentage of Net sales are as follows:

	Year Ended December 31,										
		2	2021			2020					
	Dollars in Millions		Percentage of Net Sales		Dollars in Millions	Percentage of Net Sales					
Net sales	\$	20,820.8	100.0 %	\$	18,467.5	100.0 %					
Cost of sales		17,252.3	82.9		15,257.4	82.6					
Gross profit		3,568.5	17.1		3,210.1	17.4					
Selling and administrative expenses		2,149.5	10.3		2,030.9	11.0					
Operating income		1,419.0	6.8		1,179.2	6.4					
Interest expense, net		(150.9)	(0.7)		(154.9)	(0.8)					
Other income (expense), net		29.7	0.1		(22.0)	(0.1)					
Income before income taxes		1,297.8	6.2		1,002.3	5.5					
Income tax expense		(309.2)	(1.5)		(213.8)	(1.2)					
Net income	\$	988.6	4.7 %	\$	788.5	4.3 %					

Net sales

Net sales by segment, in dollars and as a percentage of total Net sales, and the year-over-year dollar and percentage change in Net sales are as follows:

			Year Ended De						
		2	021	2020					
(dollars in millions)		Net Sales	Percentage of Total Net Sales		Net Sales	Percentage of Total Net Sales		Dollar Change	Percent Change ⁽¹⁾
Corporate	\$	8,179.7	39.3 %	\$	6,846.0	37.1 %	\$	1,333.7	19.5 %
Small Business		1,870.1	9.0		1,397.1	7.6		473.0	33.9
Public:									
Government		2,155.6	10.4		2,978.5	16.1		(822.9)	(27.6)
Education		4,108.7	19.7		3,458.1	18.7		650.6	18.8
Healthcare		1,919.3	9.2		1,701.1	9.2		218.2	12.8
Total Public		8,183.6	39.3		8,137.7	44.0		45.9	0.6
Other		2,587.4	12.4		2,086.7	11.3		500.7	24.0
Total Net sales	\$	20,820.8	100.0 %	\$	18,467.5	100.0 %	\$	2,353.3	12.7 %

(1) There were 254 selling days for both the years ended December 31, 2021 and 2020.

Total Net sales for the year ended December 31, 2021 increased \$2,353 million, or 12.7%, to \$20,821 million, compared to the prior year. This increase includes \$197 million of Net sales from the acquisition of Sirius which closed on December 1, 2021. The Net sales impact from the acquisition of Sirius is included in our Corporate, Small Business and Public segments. Net sales growth was primarily driven by Corporate, Education and Small Business customers and the results from the UK and Canadian operations included in Other, partially offset by lower Net sales to Government customers.

Corporate segment Net sales for the year ended December 31, 2021 increased \$1,334 million, or 19.5%, compared to the year ended December 31, 2020. The increase was primarily driven by hybrid work resulting in higher demand for notebooks/mobile devices, video and accessories. Additionally, Corporate customers continued to prioritize digital transformation, hybrid and cloud and security, driving growth in solutions categories, including servers and software.

Small Business segment Net sales for the year ended December 31, 2021 increased by \$473 million, or 33.9%, compared to the year ended December 31, 2020. Customers continued to focus on remote enablement as Net sales growth was driven by notebooks/mobile devices, video and accessories.

Public segment Net sales for the year ended December 31, 2021 increased \$46 million, or 0.6%, compared to the year ended December 31, 2020. The increase was primarily driven by growth in Education and Healthcare customers, offset by lower Net sales with Government customers. Net sales to Education customers increased 18.8% primarily driven by integrated solutions, including notebooks/mobile devices, video, accessories and services. Schools continued to prioritize equity and access to learning and investing in the interactive learning experience for both the classroom and dorm room. Net sales to Healthcare customers increased 12.8% primarily driven by desktops, software, notebooks/mobile devices, servers, video and services. Healthcare customers saw patients returning for elective procedures which increased confidence in budgets, enabling delayed projects to restart. Net sales to Government customers decreased 27.6%. Government decreased in most transactional and solutions categories primarily driven by several one-time activities in 2020 that did not reoccur in 2021, including the Census project, timebound stimulus funding and device refreshes related to large customer contracts. In addition, Government had contracting delays across certain large contracts in 2021.

Net sales in Other, which is comprised of results from our UK and Canadian operations, for the year ended December 31, 2021 increased \$501 million, or 24.0%, compared to the year ended December 31, 2020. UK and Canadian Net sales increased as a result of the economic recovery from 2020 and increased customer confidence. Customers in the UK and Canada remained focused on hybrid work and learning as Net sales growth was driven by notebooks/mobile devices, video and software. The impact of foreign currency exchange increased Other Net sales by approximately 810 basis points, primarily due to the favorable translation of the Canadian dollar and British pound to the US dollar.

Gross profit

Gross profit increased \$359 million, or 11.2%, to \$3,569 million for the year ended December 31, 2021, compared to \$3,210 million for the year ended December 31, 2020. As a percentage of Net sales, Gross profit margin decreased 30 basis points to 17.1% for the year ended December 31, 2021. This decrease in Gross profit margin was primarily due to lower product margin and higher margin configuration services in the prior year, partially offset by an increase in the mix of net service contract revenue, primarily Software as a Service, increase in Net sales and related margins on professional services.

Selling and administrative expenses

Selling and administrative expenses increased \$119 million, or 5.8%, to \$2,150 million for the year ended December 31, 2021, compared to \$2,031 million for the year ended December 31, 2020. The increase was primarily due to higher payroll expenses consistent with higher Gross profit, higher coworker count and higher performance-based compensation consistent with higher attainment against financial goals, and higher acquisition and integration costs, partially offset by lower intangible asset amortization and lower bad debt expense. Total coworker count was 13,924, up 3,942 from 9,982 at December 31, 2020 primarily due to an increase in customer-facing coworkers as a result of our recent acquisitions and an increase in new hires during 2021.

As a percentage of total Net sales, Selling and administrative expenses decreased 70 basis points to 10.3% for the year ended December 31, 2021, compared to 11.0% for the year ended December 31, 2020 primarily due to lower intangible asset amortization, lower bad debt expense and lower payroll expenses as a percentage of Net sales.

Operating income

Operating income by segment, in dollars and as a percentage of Net sales, and the year-over-year percentage change was as follows:

		2021	2	2020				
	Dollars in Millions	Operating Margin	Dollars in Millions	Operating Margin	Percent Change in Operating Income			
Segments:(1)								
Corporate	\$ 697.3	8.5 %	\$ 489.5	7.2 %	42.4 %			
Small Business	167.	9.0	99.0	7.1	69.4			
Public	606.	7.4	678.2	8.3	(10.5)			
Other ⁽²⁾	115.8	3 4.5	65.9	3.2	75.5			
Headquarters ⁽³⁾	(168.5	nm*	(153.4)	nm*	(9.7)			
Total Operating income	\$ 1,419.0	6.8 %	\$ 1,179.2	6.4 %	20.3 %			

^{*} Not meaningful

- (1) Segment operating income includes the segment's direct operating income, allocations for certain Headquarters' costs, allocations for income and expenses from logistics services, certain inventory adjustments and volume rebates and cooperative advertising from vendors.
- (2) Includes the financial results for our other operating segments, CDW UK and CDW Canada, which do not meet the reportable segment quantitative thresholds.
- (3) Includes Headquarters' function costs that are not allocated to the segments.

Operating income was \$1,419 million for the year ended December 31, 2021, an increase of \$240 million, or 20.3%, compared to \$1,179 million for the year ended December 31, 2020. Operating income increased primarily due to higher Gross profit dollars, lower intangible asset amortization and lower bad debt expense, partially offset by higher payroll expenses consistent with higher Gross profit, higher coworker count, higher performance-based compensation consistent with higher attainment against financial goals, and higher acquisition and integration expenses. Total operating margin percentage increased 40 basis points to 6.8% for the year ended December 31, 2021, from 6.4% for the year ended December 31, 2020 primarily due to lower intangible asset amortization, lower bad debt expense and lower payroll as a percentage of Net sales, partially offset by lower Gross profit margin and higher acquisition and integration expenses as a percentage of Net sales.

Corporate segment Operating income was \$697 million for the year ended December 31, 2021, an increase of \$207 million, or 42.4%, compared to \$490 million for the year ended December 31, 2020. Corporate segment Operating income increased primarily due to higher Gross profit and lower intangible asset amortization, partially offset by higher payroll expenses. Corporate segment operating margin percentage increased 130 basis points to 8.5% for the year ended December 31, 2021, from 7.2% for the year ended December 31, 2020 primarily due to lower intangible asset amortization and lower payroll expense as a percentage of Net sales.

Small Business segment Operating income was \$168 million for the year ended December 31, 2021, an increase of \$69 million, or 69.4%, compared to \$99 million for the year ended December 31, 2020. Small Business segment Operating income increased primarily due to higher Gross profit and lower intangible asset amortization, partially offset by higher payroll expenses. Small Business segment operating margin percentage increased 190 basis points to 9.0% for the year ended December 31, 2021, from 7.1% for the year ended December 31, 2020 primarily due to lower intangible asset amortization and lower payroll expenses as a percentage of Net sales.

Public segment Operating income was \$607 million for the year ended December 31, 2021, a decrease of \$71 million, or 10.5%, compared to \$678 million for the year ended December 31, 2020. Public segment Operating income decreased primarily due to higher payroll expenses and lower Gross profit dollars. Public segment operating margin percentage decreased 90 basis points to 7.4% for the year ended December 31, 2021, from 8.3% for the year ended December 31, 2020, primarily due to higher payroll expenses and higher margin configuration services in the prior year.

Other Operating income, which is comprised of results from our UK and Canadian operations, was \$116 million for the year ended December 31, 2021, an increase of \$50 million, or 75.5%, compared to \$66 million for the year ended December 31, 2020. Other Operating income increased primarily due to higher Gross profit and lower bad debt expense, partially offset by higher payroll expenses. Other operating margin percentage increased 130 basis points to 4.5% for the year ended December 31, 2021, from 3.2% for the year ended December 31, 2020, primarily due to lower expenses, including payroll expenses, bad debt expense, intangible asset amortization, integration costs and other selling and administrative expenses, partially offset by lower product margin.

Interest expense, net

Interest expense, net in 2021 was \$151 million, a decrease of \$4 million, compared to \$155 million in 2020. This decrease was primarily driven by lower effective interest rates in 2021 compared to 2020, partially offset by additional interest expense from the \$2.5 billion aggregate principal amount of senior notes issued on December 1, 2021, the net proceeds of which were used to fund the acquisition of Sirius.

Other income (expense), net

During the year ended December 31, 2021, we sold all ownership interests in an equity method investment and recognized a \$36 million gain. During the year ended December 31, 2020, we completed the August 2020 senior notes refinancing and recorded a \$27 million Net loss on extinguishment of long-term debt.

Income tax expense

Income tax expense was \$309 million in 2021, compared to \$214 million in 2020. The effective income tax rate, expressed by calculating income tax expense as a percentage of Income before income taxes, was 23.8% and 21.3% for 2021 and 2020, respectively.

For 2021, the effective tax rate differed from the US federal statutory rate primarily due to state and local income taxes and a discrete deferred tax expense as a result of an increase in the UK corporate tax rate effective in 2023, partially offset by excess tax benefits on equity-based compensation. For 2020, the effective tax rate differed from the US federal statutory rate primarily due to state and local income taxes and a discrete deferred tax expense as a result of an increase in the UK corporate tax rate, largely offset by excess tax benefits on equity-based compensation and tax benefits associated with global intangible low taxed income and nondeductible expenses.

The 2021 effective tax rate was higher than 2020 primarily due to certain tax benefits incurred in the prior year with no similar activity in the current year and a less favorable tax rate impact of excess tax benefits on equity-based compensation.

Non-GAAP Financial Measure Reconciliations

We have included reconciliations of Non-GAAP operating income, Non-GAAP operating income margin, Non-GAAP income before income taxes, Non-GAAP net income and Net sales growth on a constant currency basis for the years ended December 31, 2021 and 2020 below.

Non-GAAP operating income excludes, among other things, charges related to the amortization of acquisition-related intangible assets, equity-based compensation and the associated payroll taxes, and acquisition and integration expenses. Non-GAAP operating income margin is defined as Non-GAAP operating income as a percentage of Net sales. Non-GAAP income before income taxes and Non-GAAP net income exclude, among other things, charges related to acquisition-related intangible asset amortization, equity-based compensation, acquisition and integration expenses, and the associated tax effects of each. Net sales growth on a constant currency basis is defined as Net sales growth excluding the impact of foreign currency translation on Net sales compared to the prior period.

Non-GAAP operating income, Non-GAAP operating income margin, Non-GAAP income before income taxes, Non-GAAP net income and Net sales growth on a constant currency basis are considered non-GAAP financial measures. Generally, a non-GAAP financial measure is a numerical measure of a company's performance or financial condition that either excludes or includes amounts that are not normally included or excluded in the most directly comparable measure calculated and presented in accordance with US GAAP. Non-GAAP measures used by management may differ from similar measures used by other companies, even when similar terms are used to identify such measures.

We believe these measures provide analysts, investors and management with helpful information regarding the underlying operating performance of our business, as they remove the impact of items that management believes are not reflective of underlying operating performance. Management uses these measures to evaluate period-over-period performance as management believes they provide a more comparable measure of the underlying business. Certain non-GAAP financial measures are also used to determine certain components of performance-based compensation.

Non-GAAP operating income

Non-GAAP operating income was \$1,645 million for the year ended December 31, 2021, an increase of \$240 million, or 17.1%, compared to \$1,405 million for the year ended December 31, 2020. As a percentage of Net sales, Non-GAAP operating income was 7.9% and 7.6% for the years ended December 31, 2021 and 2020, respectively.

	 Year Ended December 31,							
(dollars in millions)	 2021	2020						
Operating income, as reported	\$ 1,419.0	\$ 1,1	179.2					
Amortization of intangibles ⁽¹⁾	94.9	1	158.1					
Equity-based compensation	72.6		42.5					
Acquisition and integration expenses	54.3		4.9					
Other adjustments	 4.6		19.9					
Non-GAAP operating income	1,645.4	1,4	404.6					
Non-GAAP operating income margin	7.9 %		7.6 %					

(1) Includes amortization expense for acquisition-related intangible assets, primarily customer relationships, customer contracts and trade names.

Non-GAAP net income

Non-GAAP net income was \$1,119 million for the year ended December 31, 2021, an increase of \$165 million, or 17.2%, compared to \$954 million for the year ended December 31, 2020.

Year Ended December 31, 2021				Year Ended December 31, 2020							
				1	Net income					Ne	t income
\$	1,297.8	\$	(309.2)	\$	988.6	\$	1,002.3	\$	(213.8)	\$	788.5
	94.9		(18.9)		76.0		158.1		(36.8)		121.3
	72.6		(42.6)		30.0		42.5		(37.0)		5.5
	54.3		(10.4)		43.9		4.9		(1.2)		3.7
	(36.0)		8.5		(27.5)		_		_		_
	6.0		(1.5)		4.5		27.3		(6.8)		20.5
	4.6		(1.2)		3.4		19.9		(5.0)		14.9
\$	1,494.2	\$	(375.3)	\$	1,118.9	\$	1,255.0	\$	(300.6)	\$	954.4
		Income before income taxes \$ 1,297.8 94.9 72.6 54.3 (36.0) 6.0 4.6	Income before income taxes	Income before income taxes Income tax expense(1) \$ 1,297.8 \$ (309.2) 94.9 (18.9) 72.6 (42.6) 54.3 (10.4) (36.0) 8.5 6.0 (1.5) 4.6 (1.2)	Income before income taxes Income tax expense(1) 1 \$ 1,297.8 \$ (309.2) \$ 94.9 (18.9) 72.6 (42.6) 54.3 (10.4) (36.0) 8.5 6.0 (1.5) 4.6 (1.2)	Income before income taxes Income tax expense(1) Net income \$ 1,297.8 \$ (309.2) \$ 988.6 94.9 (18.9) 76.0 72.6 (42.6) 30.0 54.3 (10.4) 43.9 (36.0) 8.5 (27.5) 6.0 (1.5) 4.5 4.6 (1.2) 3.4	Income before income taxes Income tax expense(1) Net income Income income \$ 1,297.8 \$ (309.2) \$ 988.6 \$ 94.9 (18.9) 76.0 72.6 (42.6) 30.0 54.3 (10.4) 43.9 (36.0) 8.5 (27.5) 6.0 (1.5) 4.5 4.6 (1.2) 3.4	Income before income taxes Income tax expense(1) Net income Income before income taxes \$ 1,297.8 \$ (309.2) \$ 988.6 \$ 1,002.3 94.9 (18.9) 76.0 158.1 72.6 (42.6) 30.0 42.5 54.3 (10.4) 43.9 4.9 (36.0) 8.5 (27.5) — 6.0 (1.5) 4.5 27.3 4.6 (1.2) 3.4 19.9	Income before income taxes Income tax expense(1) Net income Income before income taxes Income taxes Income before income taxes Income taxes	Income before income taxes Income taxe expense(1) Net income Income before income taxes Income tax expense(1) \$ 1,297.8 \$ (309.2) \$ 988.6 \$ 1,002.3 \$ (213.8) 94.9 (18.9) 76.0 158.1 (36.8) 72.6 (42.6) 30.0 42.5 (37.0) 54.3 (10.4) 43.9 4.9 (1.2) (36.0) 8.5 (27.5) — — 6.0 (1.5) 4.5 27.3 (6.8) 4.6 (1.2) 3.4 19.9 (5.0)	Income before income taxes Income tax expense(1) Net income Income before income taxes Income tax expense(1) Ne expense(1)

- (1) Income tax on non-GAAP adjustments includes excess tax benefits associated with equity-based compensation.
- (2) Includes amortization expense for acquisition-related intangible assets, primarily customer relationships, customer contracts and trade names.

Net sales growth on a constant currency basis

Net sales increased \$2,353 million, or 12.7%, to \$20,821 million for the year ended December 31, 2021, compared to \$18,468 million for the year ended December 31, 2020. Net sales on a constant currency basis, which excludes the impact of foreign currency translation, increased \$2,207 million, or 11.9%.

	 Year Ended I		
(dollars in millions)	2021	2020	% Change(1)
Net sales, as reported	\$ 20,820.8	\$ 18,467.5	12.7 %
Foreign currency translation ⁽²⁾	_	146.2	
Net sales, on a constant currency basis	\$ 20,820.8	\$ 18,613.7	11.9 %

- (1) There were 254 selling days for both the years ended December 31, 2021 and 2020.
- (2) Represents the effect of translating Net sales for the year ended December 31, 2020 of CDW UK and CDW Canada at the average exchange rates applicable in 2021.

Seasonality

While we have not historically experienced significant seasonality throughout the year, sales in our Corporate segment, which primarily serves US private sector business customers with more than 250 employees, are typically higher in the fourth quarter than in other quarters due to customers spending their remaining technology budget dollars at the end of the year. Additionally, sales in our Public segment have historically been higher in the third quarter than in other quarters primarily due to the buying patterns of the federal government and education customers. Since the onset of the pandemic, we have experienced variability compared to historic seasonality trends. As uncertainty due to COVID-19 remains, seasonality may continue to be different than historical experience.

Liquidity and Capital Resources

Overview

We finance our operations and capital expenditures with internally generated cash from operations and borrowings under our revolving loan facility. As of December 31, 2021, we had \$1.0 billion of availability for borrowings under our revolving loan facility. Our liquidity and borrowing plans are established to align with our financial and strategic planning processes and

Table of Contents

ensure we have the necessary funding to meet our operating commitments, which primarily include the purchase of inventory, payroll and general expenses. We also take into consideration our overall capital allocation strategy, which includes dividend payments, assessment of debt levels, acquisitions and share repurchases. We believe we have adequate sources of liquidity and funding available for at least the next year; however, there are a number of factors that may negatively impact our available sources of funds. The amount of cash generated from operations will be dependent upon factors such as the successful execution of our business plan, general economic conditions and working capital management.

Our material contractual obligations consist of debt and related interest payments and operating leases. See Note 9 (Debt) and Note 11 (Leases) to the accompanying Consolidated Financial Statements for additional information regarding future maturities of debt and operating leases.

Long-Term Debt and Financing Arrangements

During the fourth quarter of 2021, we entered into a commitment letter for a \$2.5 billion senior unsecured 364-day bridge loan facility (the "Bridge Facility"), which would have been used in the event permanent financing was not obtained on or before completing the acquisition of Sirius. In lieu of borrowing under the Bridge Facility, on December 1, 2021, we obtained permanent financing through the issuance of \$1.0 billion aggregate principal amount of 2.670% Senior Notes due 2026, \$500 million aggregate principal amount of 3.276% Senior Notes due 2028 and \$1.0 billion aggregate principal amount of 3.569% Senior Notes due 2031. The Bridge Facility was automatically terminated upon completing the acquisition of Sirius without using the Bridge Facility.

Also during the fourth quarter of 2021, we entered into the Revolving Loan Facility, a new five-year \$1.6 billion senior unsecured revolving loan facility, which replaced the senior secured asset-based revolving credit facility (the "ABL Facility"). On the same date, we also entered into the Term Loan Facility, a new five-year \$1.4 billion senior unsecured term loan facility, which replaced the senior secured term loan facility.

During the first quarter of 2021, we amended, extended and increased the size of the ABL Facility, prior to its extinguishment during the fourth quarter. Simultaneously, we paid off the remaining principal amount on the variable rate CDW UK term loan by drawing on the ABL Facility.

As of December 31, 2021, we had total unsecured indebtedness of \$6.9 billion. At December 31, 2021, we were in compliance with the covenants under our various credit agreements and indentures.

For additional information regarding our debt and refinancing activities, see Note 9 (Debt) to the accompanying Consolidated Financial Statements. For additional information regarding the acquisition of Sirius, see Note 3 (Acquisitions) to the accompanying Consolidated Financial Statements.

Inventory Financing Agreements

We have entered into agreements with certain financial intermediaries to obtain more favorable terms on purchases of inventory from various suppliers under certain terms and conditions. These amounts are classified separately as Accounts payable-inventory financing on the Consolidated Balance Sheets. We do not incur any interest expense associated with these agreements as balances are paid when they are due. For additional information, see Note 7 (Inventory Financing Agreements) to the accompanying Consolidated Financial Statements.

Share Repurchase Program

During 2021, we repurchased 8.7 million shares of our common stock for \$1,500 million under the previously announced share repurchase program. For additional information, refer to Note 12 (Stockholders' Equity) to the accompanying Consolidated Financial Statements.

Dividends

A summary of 2021 dividend activity for our common stock is as follows:

Dividend Amount	Declaration Date	Record Date	Payment Date
\$0.400	February 10, 2021	February 25, 2021	March 10, 2021
\$0.400	May 5, 2021	May 25, 2021	June 10, 2021
\$0.400	August 4, 2021	August 25, 2021	September 10, 2021
\$0.500	November 3, 2021	November 24, 2021	December 10, 2021
\$1.700			

On February 9, 2022, we announced that our Board of Directors declared a quarterly cash dividend on our common stock of \$0.500 per share. The dividend will be paid on March 10, 2022 to all stockholders of record as of the close of business on February 25, 2022.

The payment of any future dividends will be at the discretion of our Board of Directors and will depend upon our results of operations, financial condition, business prospects, capital requirements, contractual restrictions, any potential indebtedness we may incur, restrictions imposed by applicable law, tax considerations and other factors that our Board of Directors deems relevant. In addition, our ability to pay dividends on our common stock will be limited by restrictions on our ability to pay dividends or make distributions to us, in each case, under the terms of our current and any future agreements governing our indebtedness.

Cash Flows

Cash flows from operating, investing and financing activities are as follows:

	Year Ende	d December 31,
(dollars in millions)	2021	2020
Net cash provided by (used in):		
Operating activities	\$ 784.6	5 \$ 1,314.3
Investing activities		
Capital expenditures ⁽¹⁾	(100.0	(158.0)
Acquisitions of businesses, net of cash acquired	(2,705.6	(43.0)
Proceeds from sale of equity method investment	36.0	_
Cash flows used in investing activities	(2,769.6	(201.0)
Financing activities		
Net change in accounts payable - inventory financing	(161.8	93.0
Financing payments on revenue generating assets	(46.1	(18.1)
Other cash flows used in financing activities	1,040.7	63.9
Cash flows provided by financing activities	832.8	138.8
Effect of exchange rate changes on cash and cash equivalents	0.1	4.1
Net (decrease) increase in cash and cash equivalents	\$ (1,152.1) \$ 1,256.2

(1) Includes expenditures for revenue generating assets

Operating Activities

Cash flows from operating activities are as follows:

	Year Ended December 31,						
(dollars in millions)		2021	2021 2020			Change	
Net income	\$	988.6	\$	788.5	\$	200.1	
Adjustments for the impact of non-cash items(1)		227.6		520.9		(293.3)	
Net income adjusted for the impact of non-cash items		1,216.2		1,309.4		(93.2)	
Changes in assets and liabilities:							
Accounts receivable ⁽²⁾		(616.8)		(226.4)		(390.4)	
Merchandise inventory ⁽³⁾		(151.0)		(71.4)		(79.6)	
Accounts payable-trade ⁽⁴⁾		374.5		253.7		120.8	
Other ⁽⁵⁾		(38.3)		49.0		(87.3)	
Net cash provided by operating activities	\$	784.6	\$	1,314.3	\$	(529.7)	

- (1) Includes items such as depreciation and amortization, equity-based compensation expense, amortization of deferred financing costs, deferred income taxes and net loss on extinguishment of long-term debt.
- (2) The change is primarily due to higher Accounts receivable balance in Public segment.
- (3) The change is primarily due to higher customer-driven stocking positions in 2021.
- (4) The change is primarily due to mixing out of vendors with extended payment terms in 2021 and higher inventory purchases at the end of 2020, partially offset by timing of payments at the end of 2021.
- (5) The change is primarily due to higher contract liabilities in 2021, partially offset by a decrease in accrued compensation, a decrease in lease incentives and an increase in receivables from vendors in 2021.

In order to manage our working capital and operating cash needs, we monitor our cash conversion cycle, defined as days of sales outstanding in accounts receivable plus days of supply in inventory minus days of purchases outstanding in accounts payable, based on a rolling three-month average. Components of our cash conversion cycle are as follows:

	Decemb	per 31,
(in days)	2021	2020
Days of sales outstanding (DSO) ⁽¹⁾	65	57
Days of supply in inventory (DIO) ⁽²⁾	17	14
Days of purchases outstanding (DPO) ⁽³⁾	(58)	(54)
Cash conversion cycle	24	17

- (1) Represents the rolling three-month average of the balance of Accounts receivable, net at the end of the period, divided by average daily Net sales for the same three-month period. Also incorporates components of other miscellaneous receivables.
- (2) Represents the rolling three-month average of the balance of Merchandise inventory at the end of the period divided by average daily Cost of sales for the same three-month period.
- (3) Represents the rolling three-month average of the combined balance of Accounts payable-trade, excluding cash overdrafts, and Accounts payable-inventory financing at the end of the period divided by average daily Cost of sales for the same three-month period.

The cash conversion cycle increased to 24 days at December 31, 2021, compared to 17 days at December 31, 2020. DSO, DIO and DPO increased 8 days, 3 days and 4 days, respectively. The increase in DSO was primarily driven by higher Accounts receivable balance in Public segment and increased net service contract revenue, such as software as a service and warranties. The increase in net service contract revenue also results in a favorable impact on DPO. DPO further benefited from favorability in timing of payments at the end of 2021. Additionally, DIO increased due to higher customer and strategic stocking positions in 2021 relative to 2020.

Investing Activities

Net cash used in investing activities increased \$2,569 million in 2021 compared to 2020. The increase was primarily due to the acquisitions of Sirius, Amplified IT LLC and Focal Point Data Risk LLC, partially offset by lower capital expenditures and proceeds from the sale of an equity method investment. For additional information regarding the acquisitions, see Note 3 (Acquisitions) to the accompanying Consolidated Financial Statements.

Financing Activities

Net cash provided by financing activities increased \$694 million in 2021 compared to 2020. The increase was primarily due to the issuance of \$2.5 billion aggregate principal amount of senior notes issued on December 1, 2021 which was used to fund the acquisition of Sirius and increased borrowings under our revolving credit facilities, partially offset by higher share repurchases and the mixing out of vendors with extended payment terms under our inventory financing arrangements. For additional information regarding the inventory financing and debt activities, see Note 7 (Inventory Financing Agreements) and Note 9 (Debt) to the accompanying Consolidated Financial Statements.

Off-Balance Sheet Arrangements

We have no off-balance sheet arrangements that have or are reasonably likely to have a material current or future effect on our financial condition, results of operations or liquidity.

Issuers and Guarantors of Debt Securities

Each series of our outstanding unsecured senior notes (the "Notes") are issued by CDW LLC and CDW Finance Corporation (the "Issuers") and are guaranteed by CDW Corporation ("Parent") and certain of each CDW LLC's direct and indirect, 100% owned, domestic subsidiaries (the "Guarantor Subsidiaries" and, together with Parent, the "Guarantors"). All guarantees by Parent and the Guarantors are joint and several, and full and unconditional; provided that guarantees by the Guarantor Subsidiaries are subject to certain customary release provisions contained in the indentures governing the Notes.

The Notes and the related guarantees are the Issuers' and the Guarantors' senior unsecured obligations and are:

- · structurally subordinated to all existing and future indebtedness and other liabilities of our non-guarantor subsidiaries and
- rank equal in right of payment with all of the Issuers' and the Guarantors' existing and future unsecured senior debt.

The following tables set forth Balance Sheet information as of December 31, 2021 and December 31, 2020, and Statement of Operations information for the years ended December 31, 2021 and 2020 for the accounts of the Issuers and the accounts of the Guarantors (the "Obligor Group"). The financial information of the Obligor Group is presented on a combined basis and the intercompany balances and transactions between the Obligor Group have been eliminated.

Balance Sheet Information

(dollars in millions)		December 31,						
(dollars in millions)		2021		2020				
Current assets	\$	4,584.1	\$	5,161.3				
Goodwill		2,373.1		2,239.1				
Other assets		1,017.3		572.1				
Total Non-current assets		3,390.4		2,811.2				
Current liabilities		3,393.0		3,265.0				
Long-term debt		6,534.6		3,856.5				
Other liabilities		562.4		209.8				
Total Long-term liabilities		7,097.0		4,066.3				

Statements of Operations Information

		Year Ended De						
(dollars in millions)		2021		2020				
Net sales	\$	17,979.4	\$	16,380.8				
Gross profit		3,078.0		2,851.8				
Operating income		1,301.9		1,113.2				
Net income		921.3		738.8				

Commitments and Contingencies

The information set forth in Note 16 (Commitments and Contingencies) to the accompanying Consolidated Financial Statements included in Part II, Item 8 of this report is incorporated herein by reference.

Critical Accounting Policies and Estimates

The preparation of the Consolidated Financial Statements in accordance with US GAAP requires management to make use of certain estimates and assumptions that affect the reported amounts of assets, liabilities, revenues and expenses, as well as related disclosure of contingent assets and liabilities in the Consolidated Financial Statements and accompanying notes. We base our estimates on historical experience and on various other assumptions that we believe are reasonable under the circumstances. Historically, we have not made significant changes to the methods for determining these estimates as our actual results have not differed materially from our estimates. We do not believe it is reasonably likely that the estimates and related assumptions will change materially in the foreseeable future; however, actual results could differ from those estimates under different assumptions, judgments or conditions.

Critical accounting policies and estimates are those that are most important to the portrayal of our financial condition and results of operations, and which require us to make our most difficult and subjective judgments, often as a result of the need to make estimates of matters that are inherently uncertain. Based on this definition, we have identified the critical accounting policies and estimates addressed below. For additional information related to significant accounting policies used in the preparation of our Consolidated Financial Statements, see Note 1 (Description of Business and Summary of Significant Accounting Policies) to the accompanying Consolidated Financial Statements.

Revenue Recognition

We sell some of our products and services as part of bundled contract arrangements containing multiple deliverables, which may include a combination of different products and services. Significant judgment may be required when determining whether products and services are considered distinct performance obligations that should be accounted for separately versus together.

For contracts consisting of multiple performance obligations, the total transaction price is allocated to each performance obligation based upon its standalone selling price. Judgment is required to determine the standalone selling price for each distinct performance obligation. For certain performance obligations, we will use a combination of methods to estimate the standalone selling price based on recent transactions. When evidence from recent transactions is not available to confirm that the prices are representative of the standalone selling price, an expected cost plus margin approach is used.

Additional judgment is required in determining whether we are the principal, and report revenues on a gross basis, or agent, and report revenues on a net basis. For each identified performance obligation in a transaction, we evaluate the facts and circumstances present to determine whether or not we control the specified good or service prior to transfer to the customer. This evaluation includes, but is not limited to, assessing indicators such as whether: (i) we are primarily responsible for fulfilling the promise to provide the specified goods or service, (ii) we have inventory risk before the specified good or service has been transferred to a customer and (iii) we have discretion in establishing the price for the specified good or service. When the evaluation indicates we control the specified good or service prior to transfer to the customer, we are acting as a principal. When the evaluation indicates we do not control the specified good or service prior transfer to the customer, we are acting as an agent.

The nature of our contracts give rise to variable consideration in the form of volume rebates and sales returns and allowances. We estimate variable consideration at the most likely amount to which we expect to be entitled. The estimates of variable consideration and determination of whether to include estimated amounts in the transaction price are based on an assessment of our anticipated performance and all information that is reasonably available.

We recognize revenue on performance obligations when the customer obtains control over the specified good or service. That is, when the customer has the ability to direct the use of and obtain substantially all of the benefits from the good or service. For the sale of hardware and software, this is generally upon delivery to the customer. As a result, we perform an analysis to estimate the amount of Net sales in-transit at the end of the period and adjust revenue and the related costs to reflect only what has been delivered to the customer. This analysis requires judgment whereby we perform an analysis of the estimated number of days of sales intransit to customers at the end of each reporting period based on a weighted-average analysis of commercial delivery terms that include drop-shipment arrangements. Changes in delivery patterns may result in a different number of business days estimated to make this adjustment.

Goodwill

Goodwill is allocated to reporting units expected to benefit from the business combination. Goodwill is subject to periodic testing for impairment at the reporting unit level on an annual basis during the fourth quarter, or more frequently if events or changes in circumstances indicate that the asset may be impaired. These events or circumstances could include a significant change in the business climate, legal factors, operating performance indicators, competition or sale or disposition of a significant portion of a reporting unit.

We may elect to utilize a qualitative assessment to determine whether it is more likely than not that the fair value of a reporting unit is less than its carrying value. As part of our qualitative assessment, judgment is required in weighing the effect of various positive and negative factors that may affect the fair value. We consider various factors, including the excess of fair value over carrying value from the last quantitative test, macroeconomic conditions, industry and market considerations, the projected financial performance and actual financial performance compared to prior year projected financial performance.

If we elect to bypass the qualitative assessment, or if indicators of impairment exist, a quantitative impairment test is performed. As part of the quantitative assessment, application of the goodwill impairment test requires judgment, including the identification of reporting units, assignment of assets and liabilities to reporting units, assignment of goodwill to reporting units, and determination of the fair value of each reporting unit. Fair value of a reporting unit is determined by using a weighted combination of an income approach and a market approach, as this combination is considered the most indicative of our fair value in an orderly transaction between market participants. This analysis requires significant judgments, including estimation of future cash flows, which is dependent on internal forecasts, estimation of the long-term rate of growth for our business, estimation of the useful life over which cash flows will occur, determination of our weighted average cost of capital, future market conditions and profitability of future business strategies. The estimates used to calculate the fair value of a reporting unit change from year to year based on operating results, market conditions and other factors. Changes in these estimates and assumptions could materially affect the determination of fair value and goodwill impairment for each reporting unit. However, our past estimates of fair value would not have indicated an impairment when revised to include subsequent years' actual results.

We completed our annual impairment analysis during the fourth quarter of 2021. We performed a qualitative analysis for all reporting units and concluded that it was more likely than not that the fair values of all reporting units exceeded their respective carrying values and, therefore, did not result in an impairment. In 2020, we performed a quantitative analysis of goodwill impairment and determined that no impairment existed.

Business combinations

We allocate purchase price consideration to the assets acquired and liabilities assumed based on their fair values as of the acquisition date. Determining the fair value of these assets and liabilities requires the use of significant estimates, particularly in valuing acquired intangible assets and Goodwill.

Purchased intangible assets other than goodwill are initially recognized at fair value and amortized over their useful lives. We determine the fair value of purchased intangible using an income approach on an individual asset basis. The fair value measurements were primarily based on significant inputs that are not observable, which are categorized as a Level 3 measurement in the fair value hierarchy. The values assigned to consideration transferred, assets acquired and liabilities assumed may be adjusted during the measurement period as new information arises.

We use the multi-period excess earnings method to determine the fair value of customer relationships. This method identifies the portion of revenue expected to be generated through repeat customers existing as of the valuation date and includes an attrition rate to account for the loss of customers over time. Critical estimates utilized in valuing customer relationships include estimated forecasted future revenue and EBITDA margin growth rates, customer attrition rates and market-participant discount rates. The assumptions we apply in forecasting future revenue and customer attrition rates is based on analysis of historical data, assessment of current and anticipated market conditions, estimated growth rates, and management plans.

Recent Accounting Pronouncements

The information set forth in Note 2 (Recent Accounting Pronouncements) to the accompanying Consolidated Financial Statements included in Part II, Item 8 of this report is incorporated herein by reference.

Item 7A. Quantitative and Qualitative Disclosures of Market Risks

Interest Rate Risk

Our market risks relate primarily to changes in interest rates. The interest rates on borrowings under our senior unsecured revolving loan facility and our senior unsecured term loan facility are floating and, therefore, are subject to fluctuations. In order to manage the risk associated with changes in interest rates on borrowings under our senior unsecured term loan facility, we have entered into interest rate caps to add stability to interest expense and to manage our exposure to interest rate fluctuations.

As of December 31, 2021, we have an interest rate cap agreement in effect with a notional amount of \$1.3 billion. For additional information, see Note 8 (Financial Instruments) to the accompanying Consolidated Financial Statements.

Foreign Currency Risk

We transact business in foreign currencies other than the US dollar, primarily the British pound and the Canadian dollar, which exposes us to foreign currency exchange rate fluctuations. Revenue and expenses generated from our international operations are generally denominated in the local currencies of the corresponding countries. The functional currency of our international operating subsidiaries is the same as the corresponding local currency. Upon consolidation, as results of operations are translated, operating results may differ from expectations. The direct effect of foreign currency fluctuations on our results of operations has not been material as the majority of our results of operations are denominated in US dollars.

Table of Contents

Item 8. Financial Statements and Supplementary Data

Index to Consolidated Financial Statements

	<u>Page</u>
Report of Independent Registered Public Accounting Firm (PCAOB ID: 42)	<u>43</u>
Consolidated Balance Sheets as of December 31, 2021 and 2020	<u>46</u>
Consolidated Statements of Operations for the years ended December 31, 2021, 2020 and 2019	<u>47</u>
Consolidated Statements of Comprehensive Income for the years ended December 31, 2021, 2020 and 2019	48
Consolidated Statements of Stockholders' Equity for the years ended December 31, 2021, 2020 and 2019	<u>49</u>
Consolidated Statements of Cash Flows for the years ended December 31, 2021, 2020 and 2019	<u>50</u>
Notes to Consolidated Financial Statements	<u>51</u>

Report of Independent Registered Public Accounting Firm

To the Stockholders and the Board of Directors of CDW Corporation and subsidiaries

Opinion on the Financial Statements

We have audited the accompanying consolidated balance sheets of CDW Corporation and subsidiaries (the Company) as of December 31, 2021 and 2020, the related consolidated statements of operations, comprehensive income, stockholders' equity, and cash flows for each of the three years in the period ended December 31, 2021, and the related notes and financial statement schedule listed in the Index at Item 15(a) (2) (collectively referred to as the "consolidated financial statements"). In our opinion, the consolidated financial statements present fairly, in all material respects, the financial position of the Company at December 31, 2021 and 2020, and the results of its operations and its cash flows for each of the three years in the period ended December 31, 2021, in conformity with U.S. generally accepted accounting principles.

We also have audited, in accordance with the standards of the Public Company Accounting Oversight Board (United States) (PCAOB), the Company's internal control over financial reporting as of December 31, 2021, based on criteria established in Internal Control-Integrated Framework issued by the Committee of Sponsoring Organizations of the Treadway Commission (2013 framework) and our report dated February 28, 2022 expressed an unqualified opinion thereon.

Basis for Opinion

These financial statements are the responsibility of the Company's management. Our responsibility is to express an opinion on the Company's financial statements based on our audits. We are a public accounting firm registered with the PCAOB and are required to be independent with respect to the Company in accordance with the U.S. federal securities laws and the applicable rules and regulations of the Securities and Exchange Commission and the PCAOB.

We conducted our audits in accordance with the standards of the PCAOB. Those standards require that we plan and perform the audit to obtain reasonable assurance about whether the financial statements are free of material misstatement, whether due to error or fraud. Our audits included performing procedures to assess the risks of material misstatement of the financial statements, whether due to error or fraud, and performing procedures that respond to those risks. Such procedures included examining, on a test basis, evidence regarding the amounts and disclosures in the financial statements. Our audits also included evaluating the accounting principles used and significant estimates made by management, as well as evaluating the overall presentation of the financial statements. We believe that our audits provide a reasonable basis for our opinion.

Critical Audit Matters

The critical audit matters communicated below are matters arising from the current period audit of the financial statements that were communicated or required to be communicated to the audit committee and that: (1) relate to accounts or disclosures that are material to the financial statements and (2) involved our especially challenging, subjective or complex judgments. The communication of critical audit matters does not alter in any way our opinion on the consolidated financial statements, taken as a whole, and we are not, by communicating the critical audit matters below, providing a separate opinion on the critical audit matters or on the accounts or disclosures to which they relate.

Description of the Matter

Revenue recognition

As described in Note 1 to the consolidated financial statements, the Company recognizes revenue upon transfer of control of promised products or services to customers. The Company applies judgment in determining whether it is the principal and reports revenue on a gross basis, or an agent and reports revenue on a net basis. The Company also sells some of its products and services as part of bundled contract arrangements containing multiple performance obligations.

Significant judgment may be required when determining whether products and services are considered distinct performance obligations that should be accounted for separately versus together. For each distinct performance obligation, judgment is required to determine the relative standalone selling price to allocate the transaction price, such as using an expected cost plus margin approach.

Auditing the Company's contracts with customers was challenging given the significant audit effort required to analyze the Company's various products, services and contract arrangements. For example, certain customer contracts contain multiple parties and there can be subjective judgment in assessing the Company's role as principal or agent in the contract arrangement. For certain other customer contracts, there can be judgment in the identification of the distinct performance obligations along with the determination of the associated relative standalone selling prices.

How We Addressed the Matter in Our Audit

We obtained an understanding of the revenue process, evaluated the design and tested the operating effectiveness of the Company's internal controls over the relevant terms of the customer contracts, including the determination of principal versus agent, the identification of distinct performance obligations and the determination of the relative standalone selling price for separate performance obligations.

To test revenue recognition, our audit procedures included among others, examination of executed customer contracts for a sample of sales transactions, and evaluating the Company's determination of principal versus agent, identifying products and services in the contract and assessing separate distinct performance obligations. To test management's determination of relative standalone selling price for separate performance obligations, we performed audit procedures that included, among others, assessing the appropriateness of the methodology applied, testing the mathematical accuracy of the underlying data and calculations and inspecting the underlying data information on a sample basis.

Accounting for the Acquisition of Sirius - Valuation of Intangible Assets

Description of the Matter

As described in Note 1 and Note 3 to the consolidated financial statements, the Company acquired Granite Parent, Inc. (also referred to as "Sirius") for net consideration of \$2.4 billion during the year ended December 31, 2021. The transaction was accounted for as a business combination and the Company preliminarily allocated \$1.1 billion of the purchase price to the fair value of identified intangible assets.

Auditing the Company's accounting for its acquisition of Sirius was complex due to the significant estimation uncertainty in the Company's preliminary determination of the fair value of identified intangible assets of \$1.1 billion, which principally consisted of customer relationships of \$1,090.0 million. The significant estimation uncertainty was primarily due to the sensitivity of the fair value of customer relationships to underlying assumptions about the future performance of the acquired business and the expectations of market participant synergies on which those assumptions were based. The Company used the income approach to measure customer relationships. The significant assumptions used to estimate the value of customer relationships included the long-term growth rate, customer attrition rate and discount rate. These significant assumptions are forward looking and could be affected by future economic and market conditions.

Table of Contents

How We Addressed the Matter in Our Audit

We obtained an understanding of the Company's process for accounting for the acquisition. We tested the design and operating effectiveness of the Company's controls over the estimation process supporting the recognition and measurement of customer relationships. We also tested controls regarding management's review of assumptions used in the valuation model.

To test the fair value of the Company's customer relationships, we performed, with the assistance of our valuation specialists, audit procedures that included evaluating the Company's selection of the valuation methodology, significant assumptions used and completeness and accuracy of the underlying data. For example, we compared the significant assumptions to historical and current industry, market and economic trends. We also tested the underlying source information used and verified the mathematical accuracy of the calculations within the valuation model.

/s/ Ernst & Young LLP We have served as the Company's auditor since 2011. Chicago, Illinois February 28, 2022

CDW CORPORATION AND SUBSIDIARIES CONSOLIDATED BALANCE SHEETS

(dollars in millions, except per share amounts)

	December 31			,	
		2021		2020	
Assets					
Current assets:					
Cash and cash equivalents	\$	258.1	\$	1,410.2	
Accounts receivable, net of allowance for credit losses of \$20.4 and \$29.6, respectively		4,499.4		3,212.6	
Merchandise inventory		927.6		760.0	
Miscellaneous receivables		435.5		379.5	
Prepaid expenses and other		357.5		191.2	
Total current assets		6,478.1		5,953.5	
Operating lease right-of-use assets		155.6		130.8	
Property and equipment, net		195.8		175.5	
Goodwill		4,382.9		2,595.9	
Other intangible assets, net		1,628.1		445.1	
Other assets		358.9		43.9	
Total Assets	\$	13,199.4	\$	9,344.7	
Liabilities and Stockholders' Equity					
Current liabilities:					
Accounts payable-trade	\$	3,114.2	\$	2,088.4	
Accounts payable-inventory financing		448.3		524.6	
Current maturities of long-term debt		102.7		70.9	
Contract liabilities		402.9		243.7	
Accrued expenses and other current liabilities:					
Compensation		361.7		288.3	
Advertising		145.5		153.4	
Sales and income taxes		65.9		104.2	
Other		454.8		424.8	
Total current liabilities		5,096.0		3,898.3	
Long-term liabilities:		-,		,0,0,0	
Debt		6,755.8		3,856.3	
Deferred income taxes		222.3		55.3	
Operating lease liabilities		184.2		169.0	
Other liabilities		235.4		68.7	
Total long-term liabilities		7,397.7		4,149.3	
Commitments and contingencies		1,571.1		1,1 17.5	
Stockholders' equity:					
Preferred stock, \$0.01 par value, 100.0 shares authorized; no shares issued or outstanding for both periods				_	
Common stock, \$0.01 par value, 1,000.0 shares authorized; 134.8 and 141.9 shares outstanding, respectively		1.3		1.4	
Paid-in capital		3,369.5		3,204.9	
Accumulated deficit		(2,570.7)		(1,813.4)	
Accumulated other comprehensive loss		(94.4)		(95.8)	
Total stockholders' equity		705.7		1,297.1	
	<u></u>		¢.		
Total Liabilities and Stockholders' Equity	\$	13,199.4	\$	9,344.7	

CDW CORPORATION AND SUBSIDIARIES CONSOLIDATED STATEMENTS OF OPERATIONS

(dollars in millions, except per share amounts)

	 Y	0,820.8 \$ 18,467.5 \$ 18,032.4 7,252.3 15,257.4 14,992.5 3,568.5 3,210.1 3,039.9 2,149.5 2,030.9 1,906.3			
	2021		2020		2019
Net sales	\$ 20,820.8	\$	18,467.5	\$	18,032.4
Cost of sales	17,252.3		15,257.4		14,992.5
Gross profit	3,568.5		3,210.1		3,039.9
Selling and administrative expenses	2,149.5		2,030.9		1,906.3
Operating income	1,419.0		1,179.2		1,133.6
Interest expense, net	(150.9)		(154.9)		(159.4)
Other income (expense), net	 29.7		(22.0)		(24.5)
Income before income taxes	 1,297.8		1,002.3		949.7
Income tax expense	 (309.2)		(213.8)		(212.9)
Net income	\$ 988.6	\$	788.5	\$	736.8
Net income per common share:					
Basic	\$ 7.14	\$	5.53	\$	5.08
Diluted	\$ 7.04	\$	5.45	\$	4.99
Weighted-average common shares outstanding:					
Basic	138.5		142.6		145.1
Diluted	140.5		144.8		147.8

CDW CORPORATION AND SUBSIDIARIES CONSOLIDATED STATEMENTS OF COMPREHENSIVE INCOME

(dollars in millions)

	Year Ended December 31,				
	2021		2020		2019
Net income	\$ 988.6	\$	788.5	\$	736.8
Other comprehensive income:					
Unrealized loss from cash flow hedge, net of tax	_		(0.6)		(11.3)
Reclassification of cash flow hedge loss to net income, net of tax	2.5		6.0		1.7
Foreign currency translation, net of tax	(1.1)		16.6		22.4
Other comprehensive income	1.4		22.0		12.8
Comprehensive income	\$ 990.0	\$	810.5	\$	749.6

CDW CORPORATION AND SUBSIDIARIES CONSOLIDATED STATEMENTS OF STOCKHOLDERS' EQUITY

(dollars in millions)

	Commo	on Stock					
	Shares	Amoun		Paid-in Capital	Accumulated Deficit	Accumulated Other Comprehensive Loss	Total Stockholders' Equity
Balance as of December 31, 2018	147.7	\$ 1	.5	\$ 2,996.9	\$ (1,892.6)	\$ (130.6)	\$ 975.2
Net income	_		_	_	736.8	_	736.8
Equity-based compensation expense	_		_	47.7	_	_	47.7
Stock option exercises	1.3		_	34.9	_	_	34.9
Coworker Stock Purchase Plan	0.1		_	14.9	_	_	14.9
Repurchases of common stock	(6.1)	(0	.1)	_	(657.1)	_	(657.2)
Dividend payments (\$1.265 per share)	_		_	0.9	(184.3)	_	(183.4)
Incentive compensation plan stock withheld for taxes	_		_	_	(21.4)	_	(21.4)
Unrealized loss from hedge accounting	_		_	_	_	(11.3)	(11.3)
Reclassification of cash flow hedge loss to net income	_		_	_	_	1.7	1.7
Foreign currency translation				 _		22.4	22.4
Balance as of December 31, 2019	143.0	\$ 1	.4	\$ 3,095.3	\$ (2,018.6)	\$ (117.8)	\$ 960.3
Net income	_		_	_	788.5	_	788.5
Equity-based compensation expense	_		_	42.5	_	_	42.5
Stock option exercises	1.4		_	49.2	_	_	49.2
Coworker Stock Purchase Plan	0.1		_	16.8	_	_	16.8
Repurchases of common stock	(2.6)		_	_	(340.6)	_	(340.6)
Dividend payments (\$1.540 per share)	_		_	1.1	(220.7)	_	(219.6)
Incentive compensation plan stock withheld for taxes	_		_	_	(22.5)	_	(22.5)
Unrealized loss from hedge accounting	_		_	_	_	(0.6)	(0.6)
Reclassification of cash flow hedge loss to net income	_		_	_	_	6.0	6.0
Foreign currency translation	_		_	_	_	16.6	16.6
Adoption of Credit Losses ASU 2016-13				 _	0.5		0.5
Balance as of December 31, 2020	141.9	\$ 1	.4	\$ 3,204.9	\$ (1,813.4)	\$ (95.8)	\$ 1,297.1
Net income	_		_	_	988.6	_	988.6
Equity-based compensation expense	_		_	72.6	_	_	72.6
Stock option exercises	1.5		_	69.9	_	_	69.9
Coworker Stock Purchase Plan	0.1		_	20.6	_	_	20.6
Repurchases of common stock	(8.7)	(0	1)	_	(1,500.3)	_	(1,500.4)
Dividend payments (\$1.700 per share)	_		_	1.5	(236.3)	_	(234.8)
Incentive compensation plan stock withheld for taxes	_		_	_	(28.5)	_	(28.5)
Reclassification of cash flow hedge loss to net income	_		_	_	_	2.5	2.5
Foreign currency translation	_		_	_	_	(1.1)	(1.1)
Adoption of Income Tax ASU 2019-12				_	19.2		19.2
Balance as of December 31, 2021	134.8	\$ 1	.3	\$ 3,369.5	\$ (2,570.7)	\$ (94.4)	\$ 705.7

CDW CORPORATION AND SUBSIDIARIES CONSOLIDATED STATEMENTS OF CASH FLOWS

(dollars in millions)

		Year Ended December 31,			
		2021	2020		2019
Cash flows from operating activities:					
Net income	\$	988.6	\$ 788.5	\$	736.8
Adjustments to reconcile net income to net cash provided by operating activities:					
Depreciation and amortization		191.2	425.6		267.1
Equity-based compensation expense		72.6	42.5		48.5
Deferred income taxes		(6.7)	(20.2)		(87.9)
Provision for credit losses		(5.4)	30.9		0.8
Other		(24.1)	42.1		28.2
Changes in assets and liabilities:					
Accounts receivable		(616.8)	(226.4)		(244.8)
Merchandise inventory		(151.0)	(71.4)		(153.0)
Other assets		(134.8)	18.6		(10.9)
Accounts payable-trade		374.5	253.7		194.1
Other liabilities		96.5	30.4		248.3
Net cash provided by operating activities		784.6	1,314.3		1,027.2
Cash flows used in investing activities:					
Capital expenditures		(100.0)	(158.0)		(236.3)
Acquisitions of businesses, net of cash acquired		(2,705.6)	(43.0)		(95.1)
Proceeds from the sale of equity method investment		36.0	_		_
Net cash used in investing activities		(2,769.6)	(201.0)		(331.4)
Cash flows from financing activities:			· · · · · · ·		
Proceeds from borrowings under revolving credit facilities		1,619.7	1,024.0		2,445.5
Repayments of borrowings under revolving credit facilities		(1,300.5)	(1,075.0)		(2,394.5)
Proceeds from issuance of long-term debt		3,917.5	1,300.0		600.0
Payments to extinguish long-term debt		(1,469.2)	(622.5)		(539.0)
Payments of debt financing fees		(38.1)	(16.2)		(9.5)
Net change in accounts payable-inventory financing		(161.8)	93.0		(1.3)
Financing payments for revenue generating assets		(46.1)	(18.1)		
Repurchases of common stock		(1,500.4)	(340.6)		(657.2)
Proceeds from stock option exercises		69.9	49.2		34.9
Payment of incentive compensation plan withholding taxes		(28.5)	(22.5)		(21.4)
Dividend payments		(234.8)	(219.6)		(183.4)
Other		5.1	(12.9)		(23.9)
Net cash provided by (used in) financing activities		832.8	138.8		(749.8)
Effect of exchange rate changes on cash and cash equivalents		0.1	4.1		2.2
Net (decrease) increase in cash and cash equivalents		(1,152.1)	1,256.2		(51.8)
Cash and cash equivalents – beginning of period		1,410.2	154.0		205.8
Cash and cash equivalents – end of period	•	258.1	\$ 1,410.2	\$	154.0
	\$	230.1	ψ 1,710.2	Ψ	134.0
Supplementary disclosure of cash flow information:	Φ.	(124.2)	e (120 t)	e.	(154.2)
Interest paid	\$	(134.3)	· · · · · · · · · · · · · · · · · · ·		(154.2)
Income taxes paid, net	\$	(351.0)	\$ (245.6)	\$	(272.2)

(dollars in millions, except per share data, unless otherwise noted)

1. Description of Business and Summary of Significant Accounting Policies

Description of Business

CDW Corporation ("Parent"), a Fortune 500 company and member of the S&P 500 Index, is a leading multi-brand provider of information technology ("IT") solutions to small, medium and large business, government, education and healthcare customers in the United States ("US"), the United Kingdom ("UK") and Canada. The Company's broad array of offerings ranges from discrete hardware and software products to integrated IT solutions and services that include on-premise, hybrid and cloud capabilities across hybrid infrastructure, digital experience and security.

Throughout this report, the terms "the Company" and "CDW" refer to Parent and its 100% owned subsidiaries.

Parent has two 100% owned subsidiaries, CDW LLC and CDW Finance Corporation. CDW LLC is an Illinois limited liability company that, together with its 100% owned subsidiaries, holds all material assets and conducts all business activities and operations of the Company. CDW Finance Corporation is a Delaware corporation formed for the sole purpose of acting as co-issuer of certain debt obligations and does not hold any material assets or engage in any business activities or operations.

Significant Accounting Policies

Basis of Presentation

The Consolidated Financial Statements have been prepared in conformity with accounting principles generally accepted in the United States of America ("US GAAP") and the rules and regulations of the US Securities and Exchange Commission ("SEC"). The Company's Consolidated Financial Statements are based on a fiscal year ended December 31.

Principles of Consolidation

The Consolidated Financial Statements include the accounts of Parent and its 100% owned subsidiaries. All intercompany transactions and accounts are eliminated in consolidation.

On October 15, 2021, the Company entered into a Purchase and Sale Agreement (the "Purchase Agreement") to acquire all issued and outstanding equity interests in Granite Parent, Inc., the parent company of Sirius Computer Solutions, Inc. ("Sirius"), for a base purchase price of \$2.5 billion in cash, subject to customary closing adjustments. On December 1, 2021, the Company completed its acquisition of Sirius. The Company included the financial results of Sirius in its Consolidated Financial Statements from the date of the acquisition. For additional information on the acquisition of Sirius, refer to Note 3 (Acquisitions).

Reclassifications

Certain prior period amounts included in the Financing activities of the Consolidated Statements of Cash Flows have been reclassified to conform with the current period presentation.

Use of Estimates

The preparation of the Consolidated Financial Statements in accordance with US GAAP requires management to make certain estimates and assumptions that affect the reported amounts of assets and liabilities and disclosure of contingent assets and liabilities as of the date of the Consolidated Financial Statements and the reported amounts of revenue and expenses during the reported periods. The Company bases its estimates on historical experience and on various other assumptions that management believes are reasonable under the circumstances, the results of which form the basis for making judgments about carrying values of assets and liabilities that are not readily apparent from other sources. Actual results and outcomes could differ from those estimates.

Except as noted within Note 2 (Recent Accounting Pronouncements) for the adoption of Accounting Standards Update ("ASU") 2019-12, Income Taxes (Topic 740): Simplifying the Accounting for Income Taxes, there have been no changes to the Company's significant accounting policies and estimates during the year ended December 31, 2021.

(dollars in millions, except per share data, unless otherwise noted)

Business Combinations

The Company accounts for business combinations using the acquisition method of accounting, which allocates the fair value of the purchase consideration to the tangible and intangible assets acquired and liabilities assumed based on their estimated fair values. The excess of the purchase consideration over the fair values of these identifiable assets and liabilities is recorded as goodwill. When determining the fair values of assets acquired and liabilities assumed, management makes significant estimates and assumptions. The Company may utilize third-party valuation specialists to assist the Company in the allocation. Initial purchase price allocations are subject to revision within the measurement period, not to exceed one year from the date of acquisition. Acquisition-related expenses and transaction costs associated with business combinations are expensed as incurred.

Cash and Cash Equivalents

Cash and cash equivalents include deposits in banks and short-term (original maturities of three months or less at the time of purchase), highly liquid investments that are readily convertible to known amounts of cash and are so near maturity that there is insignificant risk of changes in value due to interest rate changes.

Accounts Receivable

Trade accounts receivable are recorded at the invoiced amount and typically do not bear interest. The Company estimates an allowance for credit losses related to accounts receivable for future expected credit losses by using relevant information such as historical information, current conditions, and reasonable and supportable forecasts. The allowance is measured on a pool basis when similar risk characteristics exist, and a loss-rate for each pool is determined using historical credit loss experience as the basis for the estimation of expected credit losses. Adjustments to historical loss information are made for differences in current conditions as well as changes in forecasted macroeconomic conditions, such as changes in the unemployment rate or gross domestic product growth rate. The Company has typically observed a higher loss-rate experience with customers in pools associated with the Company's Corporate and Small Business segments, as compared to the pools associated with the Public segment.

The Company occasionally may transfer certain accounts receivable, without recourse, to third-party financial companies as a method to accelerate cash collections and reduce the Company's credit exposure. Under these agreements, the Company may transfer certain accounts receivable in exchange for cash less a discount, as defined by the agreements. The Company's ability to sell receivables is dependent on the financial institutions' willingness to purchase such receivables. In addition, certain of these agreements also require that the Company continue to service, administer and collect the sold accounts receivable. Such transfers are recognized as a sale and the related accounts receivable is derecognized from the Consolidated Balance Sheet upon receipt of the third-party financing company's payment. During the years ended December 31, 2021 and 2020, the Company sold approximately \$139 million and \$83 million, respectively, of accounts receivable.

Merchandise Inventory

Inventory is valued at the lower of cost and net realizable value. Cost is determined using a weighted-average cost method. Price protection is recorded when earned as a reduction to the cost of inventory. The Company decreases the value of inventory for estimated obsolescence equal to the difference between the cost of inventory and the net realizable value, based upon an aging analysis of the inventory on hand, specifically known inventory-related risks and assumptions about future demand and market conditions.

Miscellaneous Receivables

Miscellaneous receivables primarily consist of amounts due from vendors. The Company receives incentives from vendors related to cooperative advertising, volume rebates, bid programs, price protection and other programs. These incentives generally relate to written vendor agreements with specified performance requirements and are generally recorded as adjustments to Cost of sales or Merchandise inventory, depending on the nature of the incentive.

Property and Equipment

Property and equipment are stated at cost, less accumulated depreciation. The Company calculates depreciation expense using the straight-line method over the estimated useful lives of the assets. For revenue generating assets, the Company calculates depreciation expense using the straight-line method to the estimated residual value over the

(dollars in millions, except per share data, unless otherwise noted)

estimated useful life of the assets. Property and equipment are reviewed for impairment when events or changes in circumstances indicate that the carrying amount may not be recoverable. Determination of recoverability is based on an estimate of undiscounted future cash flows resulting from the use of the asset and its eventual disposition. If the carrying amount of an asset exceeds its estimated future undiscounted cash flows, an impairment loss is recorded for the excess of the asset's carrying amount over its fair value. Leasehold improvements are amortized over the shorter of their estimated useful lives or the remaining lease term. Expenditures for major renewals and improvements that extend the useful life of property and equipment are capitalized. Expenditures for maintenance and repairs are charged to expense as incurred.

Leases

The Company enters into operating lease contracts, as assessed at contract inception, primarily for real estate, data centers and equipment. On the lease commencement date, the Company records operating lease liabilities based on the present value of the future lease payments. In determining the present value of future lease payments, the Company uses its incremental borrowing rate based on the information available at the commencement date. For real estate and data center contracts, the Company accounts for the lease and non-lease components as a single lease component. For certain equipment leases, the Company applies a portfolio approach to account for the right-of-use asset and operating lease liability. In assessing the lease term, the Company includes options to renew only when it is reasonably certain that it will be exercised; a determination which is at the sole discretion of the Company. For leases with an initial term of 12 months or less, the Company has elected to not record a right-of-use asset and lease liability. For equipment leases used in revenue generating activities, the Company records a right-of-use asset and lease liability for leases with a term of 12 months or less. The Company records lease expense on a straight-line basis over the lease term beginning on the commencement date.

Goodwill

The Company performs an evaluation of goodwill at the reporting unit level, utilizing either a qualitative or quantitative impairment test. A qualitative assessment is performed at least on an annual basis to determine whether it is more likely than not that the fair value of a reporting unit is less than its carrying value. The Company performs a quantitative impairment test for each reporting unit every three years, or more frequently if circumstances indicate a potential impairment. The annual test for impairment is conducted as of December 1. The Company's reporting units included in the assessment of potential goodwill impairment are the same as its operating segments.

Under a qualitative assessment, the most recent quantitative assessment is used to determine if it is more likely than not that the reporting unit's goodwill is impaired. As part of this qualitative assessment, the Company assesses relevant events and circumstances including macroeconomic conditions, industry and market conditions, cost factors, overall financial performance, changes in share price and entity-specific events to determine if there is an indication of impairment.

Under a quantitative assessment, goodwill impairment is identified by comparing the fair value of a reporting unit to its carrying amount, including goodwill. If the carrying amount of a reporting unit exceeds its fair value, goodwill is considered impaired and an impairment charge is recognized in an amount equal to that excess, not to exceed the carrying amount of goodwill. Fair value of a reporting unit is determined by using a weighted combination of an income approach (75%) and a market approach (25%), as this combination is considered the most indicative of the Company's fair value in an orderly transaction between market participants.

Under the income approach, the Company determines fair value based on estimated future cash flows of a reporting unit, discounted by an estimated weighted-average cost of capital, which reflects the overall level of inherent risk of a reporting unit and the rate of return an outside investor would expect to earn. The estimated future cash flows of each reporting unit are based on internally generated forecasts for the remainder of the respective reporting period and the next five years.

Under the market approach, the Company utilizes valuation multiples derived from publicly available information for guideline companies to provide an indication of how much a knowledgeable investor in the marketplace would be willing to pay for a company. The valuation multiples are applied to the reporting units.

Determining the fair value of a reporting unit is judgmental in nature and requires the use of significant estimates and assumptions, including Net sales growth rates, gross profit margins, operating margins, discount rates and future

(dollars in millions, except per share data, unless otherwise noted)

market conditions, among others. Any changes in the judgments, estimates or assumptions used could produce significantly different results.

Intangible Assets

Intangible assets with determinable lives are amortized on a straight-line basis over their respective estimated useful lives. The cost of computer software developed or obtained for internal use is capitalized and amortized on a straight-line basis over the estimated useful life of the software. Intangible assets are reviewed for impairment when events or changes in circumstances indicate that the carrying amount of such assets may not be recoverable. Determination of recoverability is based on an estimate of undiscounted future cash flows resulting from the use of the asset and its eventual disposition. If the carrying amount of an asset exceeds its estimated future undiscounted cash flows, an impairment loss is recorded for the excess of the asset's carrying amount over its fair value. In addition, each quarter, the Company evaluates whether events and circumstances warrant a revision to the remaining estimated useful life of each of these intangible assets. If the Company were to determine that a change to the remaining estimated useful life of an intangible asset was necessary, then the remaining carrying amount of the intangible asset would be amortized prospectively over that revised remaining useful life.

Deferred Financing Costs

Deferred financing costs, such as underwriting, financial advisory, professional fees and other similar fees are capitalized and recognized in Interest expense, net over the estimated life of the related debt instrument using the effective interest method or straight-line method, as applicable. The Company classifies deferred financing costs as a direct deduction from the carrying value of the Long-term debt liability on the Consolidated Balance Sheets, except for deferred financing costs associated with revolving credit facilities which are presented as an asset, within Other assets on the Consolidated Balance Sheets.

Derivative Instruments

The Company has interest rate cap agreements for the purpose of hedging its exposure to fluctuations in interest rates. The interest rate cap agreements are designated as cash flow hedges of interest rate risk and recorded at fair value in Other assets on the Consolidated Balance Sheets. Changes in fair value of the derivative instruments, along with the change in the fair value of the hedged item, are reported as a component of Accumulated other comprehensive loss until reclassified to Interest expense, net in the same period the hedge transaction affects earnings.

Fair Value Measurements

Fair value is defined under US GAAP as the price that would be received to sell an asset or paid to transfer a liability in an orderly transaction between market participants at the measurement date. A fair value hierarchy has been established for valuation inputs to prioritize the inputs into three levels based on the extent to which inputs used in measuring fair value are observable in the market. Each fair value measurement is reported in one of the three levels which is determined by the lowest level input that is significant to the fair value measurement in its entirety. These levels are:

Level 1 – observable inputs such as quoted prices for identical instruments traded in active markets.

Level 2 – inputs are based on quoted prices for similar instruments in active markets, quoted prices for identical or similar instruments in markets that are not active and model-based valuation techniques for which all significant assumptions are observable in the market or can be corroborated by observable market data for substantially the full term of the assets or liabilities.

Level 3 – inputs are generally unobservable and typically reflect management's estimates of assumptions that market participants would use in pricing the asset or liability. The fair values are therefore determined using model-based techniques that include option pricing models, discounted cash flow models and similar techniques.

Revenue Recognition

The Company is a primary distribution channel for a large group of vendors and suppliers, including original equipment manufacturers ("OEMs"), software publishers and wholesale distributors.

(dollars in millions, except per share data, unless otherwise noted)

The Company accounts for a contract when it has approval and commitment from both parties, the rights of the parties are identified, payment terms are established, the contract has commercial substance and collectability of consideration is probable. The Company evaluates the following indicators amongst others when determining whether it is acting as a principal in the transaction and recording revenue on a gross basis: (i) the Company is primarily responsible for fulfilling the promise to provide the specified goods or service, (ii) the Company has inventory risk before the specified good or service has been transferred to a customer or after transfer of control to the customer and (iii) the Company has discretion in establishing the price for the specified good or service. If the terms of a transaction do not indicate the Company is acting as a principal in the transaction, then the Company is acting as an agent in the transaction and the associated revenues are recognized on a net basis.

The Company recognizes revenue once control has passed to the customer. The following indicators are evaluated in determining when control has passed to the customer: (i) the Company has a right to payment for the product or service, (ii) the customer has legal title to the product, (iii) the Company has transferred physical possession of the product to the customer, (iv) the customer has the significant risk and rewards of ownership of the product and (v) the customer has accepted the product. The Company's products can be delivered to customers in a variety of ways, including (i) as physical product shipped from the Company's warehouse, (ii) via drop-shipment by the vendor or supplier or (iii) via electronic delivery of keys for software licenses. The Company's shipping terms typically allow for the Company to recognize revenue when the product reaches the customer's location.

The Company leverages drop-shipment arrangements with many of its vendors and suppliers to deliver products to its customers without having to physically hold the inventory at its warehouses. The Company is the principal in the transaction and recognizes revenue for drop-shipment arrangements on a gross basis.

Revenue Recognition for Hardware

Revenues from sales of hardware products are recognized on a gross basis as the Company is acting as a principal in these transactions, with the selling price to the customer recorded as Net sales and the acquisition cost of the product recorded as Cost of sales. The Company recognizes revenue from these transactions when control has passed to the customer, which is usually upon delivery of the product to the customer.

In some instances, the customer agrees to buy the product from the Company but requests delivery at a later date, commonly known as bill-and-hold arrangements. For these transactions, the Company deems that control passes to the customer when the product is ready for delivery. The Company views products ready for delivery when the customer has a signed agreement, significant risk and rewards for the products and the ability to direct the assets, and the products have been set aside specifically for the customer, cannot be redirected to another customer and for customer orders that include configuration services, when such services have been completed.

The Company's vendor partners warrant most of the products the Company sells. These manufacturer warranties are assurance-type warranties and are not considered separate performance obligations. The warranties are not sold separately and only provide assurance that products will conform with the manufacturer's specifications. In some transactions, a third party will provide the customer with an extended warranty. These extended warranties are sold separately and provide the customer with a service in addition to assurance that the product will function as expected. The Company considers these warranties to be separate performance obligations from the underlying product. For extended warranties, the Company is arranging for those services to be provided by the third party and therefore is acting as an agent in the transaction and records revenue on a net basis at the point of sale.

The Company sells cloud computing solutions which include Infrastructure as a Service ("IaaS"). IaaS solutions utilize third-party partners to enable customers to access data center functionality in a cloud-based solution, including storage, computing and networking. The Company recognizes revenue for cloud computing solutions for arrangements with one-time invoicing to the customer at the time of invoice on a net basis as the Company is acting as an agent in the transaction. For monthly subscription-based arrangements, the Company is acting as an agent in the transaction and recognizes revenue as it invoices the customer for its monthly usage on a net basis.

Revenue Recognition for Software

Revenues from most software license sales are recognized as a single performance obligation on a gross basis as the Company is acting as a principal in these transactions at the point the software license is delivered to the customer. Generally, software licenses are sold with accompanying third-party delivered software assurance, which is a product that allows customers to upgrade, at no additional cost, to the latest technology if new capabilities are introduced

(dollars in millions, except per share data, unless otherwise noted)

during the period that the software assurance is in effect. The Company evaluates whether the software assurance is a separate performance obligation by assessing if the third-party delivered software assurance is critical or essential to the core functionality of the software itself. This involves considering if the software provides its original intended functionality to the customer without the updates, if the customer would ascribe a higher value to the upgrades versus the up-front deliverable, if the customer would expect frequent intelligence updates to the software (such as updates that maintain the original functionality), and if the customer chooses to not delay or always install upgrades. If the Company determines that the accompanying third-party delivered software assurance is critical or essential to the core functionality of the software license, the software license and the accompanying third-party delivered software assurance are recognized as a single performance obligation. The value of the product is primarily the accompanying support delivered by a third party and therefore the Company is acting as an agent in these transactions and recognizes them on a net basis at the point the associated software license is delivered to the customer. For software licenses where the accompanying third-party delivered software assurance is not critical or essential to the core functionality, the software assurance is recognized as a separate performance obligation, with the associated revenue recognized on a net basis at the point the related software license is delivered to the customer. For additional information regarding the accounting for bundled arrangements, see "Revenue Recognition for Bundled Arrangements" below.

The Company sells cloud computing solutions which include Software as a Service ("SaaS"). SaaS solutions utilize third-party partners to offer the Company's customers access to software in the cloud that enhances office productivity, provides security or assists in collaboration. The Company recognizes revenue for cloud computing solutions for arrangements with one-time invoicing to the customer at the time of invoice on a net basis as the Company is acting as an agent in the transaction. For monthly subscription-based arrangements, the Company is acting as an agent in the transaction and recognizes revenue as it invoices the customer for its monthly usage on a net basis.

The Company's customers are offered the opportunity by certain of its vendors to purchase software licenses and software assurance under enterprise agreements ("EAs"). For most EA transactions, the Company's obligation to the customer is that of a distributor or sales agent of the services, where all obligations for providing the services to customers are passed to the Company's vendors. The Company's performance obligations are satisfied at the time of the sale. In other EA transactions, the Company is responsible for fulfilling the promised services to the customer and providing remedy or refund for work if the customer is not satisfied with the delivered services, has inventory risk in the arrangement and has full control to set the price for the customer. With most EAs, the Company's vendors will transfer the license and invoice the customer directly, paying resellers an agency fee or commission on these sales. The Company records these fees as a component of Net sales as earned and there is no corresponding Cost of sales amount.

Revenue Recognition for Services

The Company provides professional services, which include project managers and consultants recommending, designing and implementing IT solutions. Revenue from professional services is recognized either on a time and materials basis or proportionally as costs are incurred for fixed fee project work. Revenue is recognized on a gross basis each month as work is performed and the Company transfers those services.

Revenues from the sale of data center services, such as managed and remote managed services, server co-location, internet connectivity and data backup and storage provided by the Company, are recognized over the period the service is provided. Most hosting and managed service obligations are based on the quantity and pricing parameters established in the agreement. As the customer receives the benefit of the service each month, the Company recognizes the respective revenue on a gross basis as the Company is acting as a principal in the transaction. Additionally, the Company's managed services team provides project support to customers that are billed on a fixed fee basis. The Company is acting as the principal in the transaction and recognizes revenue on a gross basis based on the total number of hours incurred for the period over the total expected hours for the project. Total expected hours to complete the project is updated for each period and best represents the transfer of control of the service to the customer.

Revenue Recognition for Bundled Arrangements

The Company also sells some of its products and services as part of bundled contract arrangements containing multiple deliverables, which may include a combination of products and services. For each deliverable that represents a distinct performance obligation, total arrangement consideration is allocated based upon the standalone selling prices of each performance obligation.

(dollars in millions, except per share data, unless otherwise noted)

Sales In-Transit

The Company performs an analysis of the estimated number of days of sales in-transit to customers at the end of each reporting period based on a weighted-average analysis of commercial delivery terms that include drop-shipment arrangements. This analysis is the basis upon which the Company estimates the amount of Net sales in-transit at the end of the period and adjusts revenue and the related costs to reflect only what has been delivered to the customer. Changes in delivery patterns may result in a different number of business days estimated to make this adjustment.

Freight Costs

The Company records freight billed to its customers as Net sales and the related freight costs as Cost of sales when the underlying product revenue is recognized. For freight not billed to its customers, the Company records the freight costs as Cost of sales. The Company's typical shipping terms result in shipping being performed before the customer obtains control of the product. The Company considers shipping to be a fulfillment activity and not a separate performance obligation.

Other

The nature of the Company's contracts give rise to variable consideration in the form of volume rebates and sales returns and allowances, which are estimated at contract inception. The Company estimates variable consideration at the most likely amount to which it is expected to be entitled. This estimated amount is included in the transaction price to the extent it is probable that a significant reversal of cumulative revenue recognized will not occur when the uncertainty associated with the variable consideration is resolved. The estimates of variable consideration and determination of whether to include estimated amounts in the transaction price are based on an assessment of the Company's anticipated performance and all information that is reasonably available. At the time of sale, the Company records a liability for estimated sales returns and allowances and an associated right of return asset. The Company also records a provision for volume rebates based on the evaluation of contract terms and historical experience.

The Company excludes amounts collected on behalf of third-parties, such as sales taxes, when determining the transaction price.

When a contract results in revenue being recognized in excess of the amount the Company has the right to invoice to the customer, a contract asset is recorded on the Consolidated Balance Sheets. Contract assets are comprised primarily of professional services with fixed fee arrangements.

Contract liabilities consist of payments received from customers, or such consideration that is contractually due, in advance of providing the product or performing services. Contract liabilities are comprised primarily of professional services with fixed fee arrangements, bill-and-hold transactions where control has not passed to the customer and certain governmental contracts.

Trade accounts receivable are recorded at the point of sale (or in accordance with the Statement of Work for services) for the total amount payable by the customer to the Company for sale of goods and services. Taxes to be collected from the customer as part of the sale are included in Accounts receivable.

Any incremental direct costs of obtaining a contract, primarily sales commissions, are deferred on the Consolidated Balance Sheets and amortized over the period of contract performance.

The Company generally does not enter into long-term contracts. The Company has elected to use the practical expedient for its performance obligations table to include only those contracts that are longer than 12 months at the time of contract inception and those contracts that are non-cancelable. Additionally, for certain governmental contracts where there are annual renewals, the Company has excluded these contracts since there is only a one-year legal obligation. Typically, the only contracts that are longer than 12 months in duration are related to the Company's managed services business.

The Company requests payments for its products and services at the point of sale. The Company generally does not enter into any long-term financing arrangements or payment plans with customers or contracts with customers that have non-cash consideration.

(dollars in millions, except per share data, unless otherwise noted)

Sales Taxes

Sales tax amounts collected from customers for remittance to governmental authorities are presented on a net basis in the Consolidated Statements of Operations.

Advertising

Advertising costs are generally charged to expense in the period incurred and are recorded in Selling and administrative expenses in the Consolidated Statements of Operations. Cooperative reimbursements from vendors are recorded in the period the related advertising expenditure is incurred. The Company classifies vendor consideration as a reduction to Cost of sales. During the years ended December 31, 2021, 2020 and 2019, the Company had advertising costs of \$199 million, \$191 million and \$193 million, respectively.

Equity-Based Compensation

The Company measures all equity-based payments using a fair-value-based method and records compensation expense over the requisite service period using the straight-line method in its Consolidated Financial Statements. The expense calculation includes estimated forfeiture rates, which have been developed based upon historical experience.

Interest Expense

Interest expense is recognized in the period incurred at the applicable interest rate in effect.

Foreign Currency Translation

The Company's reporting currency is the US dollar. The functional currency of the Company's international operating subsidiaries is generally the same as the corresponding local currency. Assets and liabilities of the international operating subsidiaries are translated at the spot rate in effect at the applicable reporting date. Revenues and expenses of the international operating subsidiaries are translated at the average exchange rates in effect during the applicable period. The resulting foreign currency translation adjustment is recorded as Accumulated other comprehensive loss, which is reflected as a separate component of Stockholders' equity.

Income Taxes

Deferred income taxes are provided to reflect the differences between the tax bases of assets and liabilities and their reported amounts in the Consolidated Financial Statements using enacted tax rates in effect for the year in which the differences are expected to reverse. The Company performs an evaluation of the realizability of deferred tax assets on a quarterly basis. This evaluation requires management to make use of estimates and assumptions and considers all positive and negative evidence and factors, such as the scheduled reversal of temporary differences, the mix of earnings in the jurisdictions in which the Company operates, and prudent and feasible tax planning strategies.

The Company accounts for unrecognized tax benefits based upon its assessment of whether a tax benefit is more likely than not to be sustained upon examination by tax authorities. The Company reports a liability for unrecognized tax benefits resulting from unrecognized tax benefits taken or expected to be taken in a tax return and recognizes interest and penalties, if any, related to its unrecognized tax benefits in income tax expense.

2. Recent Accounting Pronouncements

Accounting for Contract Assets and Contract Liabilities

In October 2021, the Financial Accounting Standards Board ("FASB") issued ASU 2021-08, Business Combinations (Topic 805): Accounting for Contract Assets and Contract Liabilities from Contracts with Customers. This ASU requires entities to recognize contract liabilities and contract assets acquired in a business combination to be recognized in accordance with ASC 606, Revenue from Contracts with Customers ("Topic 606") as if the acquirer had originated the contracts, subject to certain considerations. As a result, the recognition and measurement of those contract liabilities and contract assets will likely be comparable to the acquiree's book value under Topic 606. This ASU is effective for the Company beginning in the first quarter of 2023 and allows for early adoption upon issuance. The Company early adopted this standard, and the impact of adoption was not significant to the Company's Consolidated Financial Statements.

(dollars in millions, except per share data, unless otherwise noted)

Accounting for Income Taxes

On January 1, 2021, the Company adopted and applied ASU 2019-12, Income Taxes (Topic 740): Simplifying the Accounting for Income Taxes ("Topic 740") in accordance with the applicable transition methods. Among the various updates, the Company adopted the accounting for ownership changes when transitioning from equity method to consolidation on a modified retrospective basis, which resulted in a \$19 million adjustment to retained earnings as of January 1, 2021 for the cumulative effect of derecognizing the deferred tax liability related to the UK acquisition. The adoption of the remaining components of Topic 740 did not have an impact to the Company's Consolidated Financial Statements.

Reference Rate Reform

In March 2020, the FASB issued ASU 2020-04, Reference Rate Reform (Topic 848): Facilitation of the Effects of Reference Rate Reform on Financial Reporting, followed by an amendment issued in January 2021. This ASU temporarily provides optional expedients and exceptions for applying existing guidance to contract modifications, hedging relationships and other transactions that are expected to be affected by reference rate reform. This ASU, as amended, was effective upon issuance and will remain in effect for all contract modifications and hedging relationships entered into through December 31, 2022. The adoption of this ASU along with the related expedients did not have an impact to the Company's Consolidated Financial Statements.

3. Acquisitions

Sirius

On December 1, 2021, the Company completed its previously announced acquisition of all issued and outstanding equity interests of Sirius, as described within Note 1 (Description of Business and Summary of Significant Accounting Policies). The aggregate consideration paid, net of cash acquired, at the closing of the acquisition was approximately \$2.4 billion, which is subject to the finalization of customary closing adjustments. Transaction costs related to the acquisition were \$35 million, which are included in Selling and administrative expenses for the year ended December 31, 2021. The Company used the net proceeds from the issuance of the \$2.5 billion aggregate principal amount of senior unsecured notes to finance the acquisition and related transaction costs. For additional information on the issuance of the senior notes, see Note 9 (Debt).

Sirius is a leading provider of secure, mission-critical technology-based solutions and is one of the largest IT solutions integrators in the United States, leveraging its services-led approach, broad portfolio of hybrid infrastructure solutions, and deep technical expertise of its 2,600 coworkers to support corporate and public customers. This strategic acquisition will enhance the Company's breadth and depth of services and solutions offerings.

Following the close of the acquisition, the Company issued a mix of cash and equity-based retention awards to certain Sirius coworkers, which vest over a required service period and will be recorded as expense over the required service period. The results of operations of Sirius are included in the consolidated financial statements of the Company beginning on the acquisition date. These amounts are presented within the Corporate, Small Business and Public reportable segments. For the year ended December 31, 2021, the Company's consolidated financial statements included \$197 million of net sales and \$9 million of net income from the results of operations of Sirius.

The acquisition of Sirius has been accounted for as a business combination. The Company is currently assessing the identification and measurement of the assets acquired and liabilities assumed as of the date of the acquisition. As the values of certain of these assets and liabilities are preliminary, they are subject to adjustment as additional information is obtained about the facts and circumstances that existed as of the acquisition date. The valuations will be finalized within twelve months following the close of the acquisition. When valuations are finalized, any changes to the preliminary valuation of assets acquired and liabilities assumed may result in adjustments to the preliminary fair value of the net identifiable assets acquired and goodwill.

(dollars in millions, except per share data, unless otherwise noted)

The preliminary purchase price allocation is as follows:

	Acquisiti	on-Date Fair Value
Cash and cash equivalents	\$	52.8
Accounts receivable		646.1
Intangible assets, net		1,140.5
Goodwill		1,572.4
Other assets		444.6
Total assets acquired		3,856.4
Accounts payable-trade		643.8
Debt		170.1
Deferred tax liabilities		208.6
Other liabilities		415.6
Total liabilities assumed		1,438.1
Total preliminary purchase price	\$	2,418.3

The Company used the income approach to value the intangible assets, consisting of acquired customer relationships and trade name. The fair value measurements were primarily based on significant inputs that are not observable, which are categorized as a Level 3 measurement in the fair value hierarchy. Significant inputs used to value these intangible assets include projection of all future cash flows, long-term growth rates, customer attrition rates, discount rates, royalty rates and applicable income tax rates. The excess purchase price recorded to goodwill primarily represents the future economic benefits the Company expects to achieve as a result of combining operations and Sirius' workforce.

The amount of goodwill expected to be deductible for income tax purposes is estimated to be \$160 million. The Company has preliminarily allocated the goodwill to the reportable segments. For additional information on goodwill allocation, see Note 6 (Goodwill and Other Intangible Assets).

The table below summarizes the preliminary estimated fair value of identifiable intangible assets acquired.

	Useful Lives (Years)	Acquisition-D	ate Fair Value
Customer relationships	12	\$	1,090.0
Trade name	3		50.5
		\$	1,140.5

The following unaudited pro forma financial information presents the combined results of operations as if the acquisition of Sirius had been consummated on January 1, 2020. The pro forma adjustments are based on historical results of operations and financial condition of the Company and Sirius and do not include any anticipated synergies or other expected benefits of the acquisition. The unaudited pro forma financial information is not necessarily indicative of the actual consolidated results of operations had the acquisition actually consummated on January 1, 2020, nor are they indicative of future consolidated results of operations of the combined company.

	 Year Ended December 31,				
	2021	2020			
Pro forma net sales	\$ 22,793.0	\$ 20,659.0			
Pro forma net income	\$ 977.4	\$ 771.1			

The pro forma adjustments include, among other things:

- Estimated impact to conform Sirius' classification to the Company's financial statement presentation.
- Estimated amortization expense of intangible assets acquired.

(dollars in millions, except per share data, unless otherwise noted)

- Estimated compensation expense for the cash and equity retention awards.
- Interest expense for the additional indebtedness incurred to fund the acquisition.
- Transaction costs that have been incurred in connection with the acquisition.

Focal Point Data Risk LLC and certain affiliates ("Focal Point")

On July 30, 2021, the Company completed the acquisition of Focal Point through a purchase of all issued and outstanding equity interests. Focal Point is a leading US-based cybersecurity services firm that brings a team skilled in identity and access management as well as the ability to serve customers across the full cybersecurity landscape. This strategic acquisition expands the Company's services and capabilities to help customers address risks posed by malicious cyber threats and cyber workforce shortages, while helping customers navigate shifting data protection laws. The acquisition of Focal Point was not material to the Company's results of operations and financial condition. The financial results of Focal Point have been included in the Company's Consolidated Financial Statements since the date of the acquisition. These amounts are presented within the Public reportable segment and are insignificant during the year ended December 31, 2021. The purchase price allocation is preliminary and subject to customary closing adjustments and revision as additional information about fair value of assets and liabilities become available. Preliminarily, the Company recorded \$36 million of intangible assets related to customer relationships.

Amplified IT LLC ("Amplified IT")

On March 15, 2021, the Company completed the acquisition of Amplified IT through a purchase of all issued and outstanding membership interests. Amplified IT is a Google Premium education partner and leading provider of Google Cloud services, solutions and software for education partners. This strategic acquisition expands the Company's services and solutions capabilities to help schools leverage technology to achieve greater educational outcomes. The acquisition of Amplified IT was not material to the Company's results of operations and financial condition. The financial results of Amplified IT have been included in the Company's Consolidated Financial Statements since the date of the acquisition. These amounts are presented within the Public reportable segment and are insignificant during the year ended December 31, 2021. The purchase price allocation is preliminary and subject to customary closing adjustments and revision as additional information about fair value of assets and liabilities become available. Preliminarily, the Company recorded approximately \$88 million of intangible assets, which primarily consisted of customer relationships.

4. Accounts Receivable and Contract Balances

Accounts Receivable

The timing of revenue recognition may differ from the time of billing to customers. Accounts receivable presented on the Consolidated Balance Sheets represent an unconditional right to consideration, which includes unbilled receivables. Unbilled receivables represent revenues that are not currently billable where payment is unconditional and solely subject to the passage of time. These items are expected to be billed and collected in the normal course of business. The balance of the Company's accounts receivable is classified as current for amounts expected to be collected within twelve months and noncurrent for amounts to be collected beyond twelve months. The following table details the total accounts receivable recognized and the related classification on the Consolidated Balance Sheets:

	 December 31,				
	2021		2020		
Accounts receivable, current	\$ 4,499.4	\$	3,212.6		
Accounts receivable, noncurrent	197.4		_		
Total accounts receivable	\$ 4,696.8	\$	3,212.6		

- (1) Accounts receivable, current are presented within Accounts receivable, net of allowance for credit losses on the Consolidated Balance Sheets.
- (2) Accounts receivable, noncurrent are presented within Other assets on the Consolidated Balance Sheets.

(dollars in millions, except per share data, unless otherwise noted)

Accounts receivable increased during the year ended December 31, 2021 primarily due to the acquisition of Sirius. For additional information on the acquisition of Sirius, refer to Note 3 (Acquisitions).

The Company recognizes an allowance for credit losses at inception and reassesses quarterly on a pool basis based on expected collectability. The following table details the changes in the allowance for credit losses related to accounts receivable:

Balance as of December 31, 2019	\$ 7.9
Increase to provision for credit losses	30.9
Write-offs charged against the allowance for credit losses	(10.8)
Other	1.6
Balance as of December 31, 2020	29.6
Decrease to provision for credit losses	(5.4)
Write-offs charged against the allowance for credit losses	(5.0)
Other	1.2
Balance as of December 31, 2021	\$ 20.4

During the year ended December 31, 2021, the Company recognized a \$5 million decrease to the provision within the Corporate and Public segments. While the overall impact and duration of the COVID-19 pandemic remains uncertain, the Company has observed improved collections for certain pools throughout the year 2021 and loss rates across certain pools are approaching pre-pandemic levels. The Company's estimates and assumptions may continue to evolve as conditions change.

Contract Balances

Contract assets and liabilities represent the difference in the timing of revenue recognition from receipt of cash from customers. Contract assets represent revenue recognized on performance obligations satisfied or partially satisfied for which the Company has no unconditional right to consideration. Contract liabilities consist of payments received from customers, or such consideration that is contractually due, in advance of providing the product or performing services. The following table details information about the Company's contract balances recognized on the Consolidated Balance Sheets:

	Dece	December 31,				
	2021		2020			
Contract assets ⁽¹⁾	\$ 134.7	\$	39.1			
Contract liabilities ⁽²⁾⁽³⁾	\$ 423.3	\$	255.3			

- (1) Contract assets are presented within Prepaid expenses and other on the Consolidated Balance Sheets.
- (2) Includes \$20 million and \$12 million of long-term contract liabilities that are presented within Other liabilities on the Consolidated Balance Sheets as of December 31, 2021 and 2020, respectively.
- (3) For the years ended December 31, 2021 and 2020, the Company recognized revenue of \$171 million and \$203 million, respectively, related to its contract liabilities that were included in the beginning balance of the respective periods.

Contract assets and contract liabilities increased \$96 million and \$168 million, respectively, primarily due to the acquisition of Sirius.

(dollars in millions, except per share data, unless otherwise noted)

A contract's transaction price is allocated to each distinct performance obligation and recognized as revenue when, or as, the performance obligation is satisfied. For additional information regarding the Company's performance obligations, see Note 1 (Description of Business and Summary of Significant Accounting Policies). The following table represents the total transaction price for the remaining performance obligations as of December 31, 2021 related to non-cancelable contracts longer than 12 months in duration that is expected to be recognized over future periods.

	Within 1 Year		Within 1 Year Years 1-2			ears 2-3	Thereafter		
Remaining performance obligations	\$	57.2	\$	26.3	\$	7.5	\$	2.8	

5. Property and Equipment

Property and equipment consist of the following:

		 Decem	ber 31,	
	Useful Lives (Years)	2021		2020
Computer and data processing equipment	3 - 5	\$ 162.1	\$	126.5
Building and leasehold improvements	5 - 25	151.3		126.8
Machinery and equipment	5 - 10	44.4		43.3
Computer software	3 - 5	32.9		22.9
Furniture and fixtures	5 - 10	31.0		21.2
Land	_*	27.7		27.7
Construction in progress	_*	12.0		50.8
Property and equipment, gross		 461.4		419.2
Less: accumulated depreciation		(265.6)		(243.7)
Property and equipment, net		\$ 195.8	\$	175.5

^{*}Asset is not depreciated.

During 2021, 2020 and 2019, the Company recorded disposals of \$20 million, \$54 million and \$3 million, respectively, to derecognize Property and equipment that were no longer in use.

Depreciation expense for the years ended December 31, 2021, 2020, and 2019 was \$42 million, \$213 million and \$41 million, respectively.

6. Goodwill and Other Intangible Assets

Goodwill

The changes in goodwill by reportable segment are as follows:

	Corporate Small Business		Public		Other ⁽¹⁾		C	onsolidated	
Balances as of December 31, 2019 ⁽²⁾	\$	1,090.6	\$ 185.9	\$	929.6	\$	346.9	\$	2,553.0
IGNW, Inc. acquisition		33.0	_		_		_		33.0
Foreign currency translation		_	_		_		9.9		9.9
Balances as of December 31, 2020 ⁽²⁾		1,123.6	185.9		929.6		356.8		2,595.9
Amplified IT acquisition ⁽³⁾		_	_		133.8		_		133.8
Focal Point acquisition ⁽³⁾		_	_		82.7		_		82.7
Sirius acquisition ⁽³⁾		900.6	80.2		591.6		_		1,572.4
Other acquisition adjustments		0.2			_		_		0.2
Foreign currency translation			 				(2.1)		(2.1)
Balances as of December 31, 2021 ⁽²⁾	\$	2,024.4	\$ 266.1	\$	1,737.7	\$	354.7	\$	4,382.9

(1) Other is comprised of CDW UK and CDW Canada operating segments.

(dollars in millions, except per share data, unless otherwise noted)

- (2) Goodwill is net of accumulated impairment losses of \$1,571 million, \$354 million and \$28 million related to the Corporate, Public and Other segments, respectively, recorded in 2008 and 2009.
- (3) For additional information regarding the Company's acquisitions, see Note 3 (Acquisitions).

Other Intangible Assets

A summary of intangible assets is as follows:

December 31, 2021	Useful Lives (Years)	Gross Carrying Amount		Accumulated Amortization	Ne	et Carrying Amount
Customer relationships	3 - 14	\$	3,330.9	\$ (1,987.8)	\$	1,343.1
Trade name	1 - 20		472.7	(302.0)		170.7
Internally developed software	3 - 5		352.0	(239.8)		112.2
Other	1 - 10		2.5	(0.4)		2.1
Total		\$	4,158.1	\$ (2,530.0)	\$	1,628.1
December 31, 2020						
Customer relationships	3 - 14	\$	2,131.5	\$ (1,927.9)	\$	203.6
Trade name	1 - 20		422.8	(280.1)		142.7
Internally developed software	3 - 5		280.6	(186.0)		94.6
Other	1 - 10		9.6	(5.4)		4.2
Total		\$	2,844.5	\$ (2,399.4)	\$	445.1

During the years ended December 31, 2021, 2020 and 2019, the Company recorded disposals of \$2 million, \$25 million and \$11 million, respectively, to remove fully amortized intangible assets that were no longer in use.

During the years ended December 31, 2021, 2020 and 2019, the Company recorded amortization expense related to intangible assets of \$149 million, \$212 million and \$219 million, respectively.

Estimated future amortization expense related to intangible assets is as follows:

Years ending December 31,	nense
2022	\$ 218.7
2023	197.9
2024	178.1
2025	149.9
2026	147.9
Thereafter	 735.6
Total future amortization expense	\$ 1,628.1

(dollars in millions, except per share data, unless otherwise noted)

7. Inventory Financing Agreements

The Company has entered into agreements with certain financial intermediaries to facilitate the purchase of inventory from various suppliers under certain terms and conditions, as described below. These amounts are classified separately as Accounts payable-inventory financing on the Consolidated Balance Sheets. The Company does not incur any interest expense associated with these agreements as balances are paid when they are due.

Amounts included in accounts payable-inventory financing are as follows:

	 December 31,			
	 2021		2020	
Revolving Loan inventory financing agreement(1)	\$ 310.1	\$	470.1	
Other inventory financing agreements	138.2		54.5	
Accounts payable-inventory financing	\$ 448.3	\$	524.6	

(1) The revolving credit facilities include an inventory floorplan sub-facility that enables the Company to maintain an inventory financing agreement with a financial intermediary.

8. Financial Instruments

The Company does not hold or issue derivative financial instruments for trading or speculative purposes. The Company's indebtedness creates interest rate risk on its variable-rate debt. The Company uses derivative financial instruments to manage its exposure to interest rate risk.

The Company has interest rate cap agreements that entitle it to payments from the counterparty of the amount, if any, by which three-month London Interbank Offered Rate ("LIBOR") exceeds the strike rates of the caps during the agreement period in exchange for an upfront premium. During 2021, the Company did not enter into new interest rate cap agreements.

As of December 31, 2021 and December 31, 2020, the Company had interest rate cap agreements with a fair value of less than \$1 million which were classified within Other assets on the Consolidated Balance Sheets. The total notional value of the interest rate cap agreements was \$1.3 billion and \$1.4 billion as of December 31, 2021 and December 31, 2020, respectively, of which \$100 million matured on December 31, 2021 and \$1.3 billion will mature on December 31, 2022.

The fair value of the Company's interest rate cap agreements is classified as Level 2 in the fair value hierarchy. The valuation of the interest rate cap agreements is derived by using a discounted cash flow analysis on the expected cash receipts that would occur if variable interest rates rise above the strike rates of the caps. This analysis reflects the contractual terms of the interest rate cap agreements, including the period to maturity, and uses observable market-based inputs, including LIBOR curves and implied volatilities. The Company also incorporates insignificant credit valuation adjustments to appropriately reflect the respective counterparty's nonperformance risk in the fair value measurements. The counterparty credit spreads are based on publicly available credit information obtained from a third-party credit data provider. For additional information, see Note 9 (Debt).

The interest rate cap agreements are designated as cash flow hedges. The changes in the fair value of derivatives that qualify as cash flow hedges are recorded in Accumulated other comprehensive loss ("AOCL") and are subsequently reclassified into Interest expense in the period when the hedged forecasted transaction affects earnings. The following tables provide the activity in AOCL, net of tax, for the years ended December 31, 2021, 2020 and 2019

	 Year Ended December 31,						
	 2021		2020		2019		
Change in fair value recorded to AOCL	\$ 	\$	(0.6)	\$	(11.3)		
Reclassification from AOCL to Interest expense, net	\$ 2.5	\$	6.0	\$	1.7		

The Company expects to reclassify \$5 million from Accumulated other comprehensive loss into Interest expense, net during the next 12 months.

(dollars in millions, except per share data, unless otherwise noted)

9. Debt

		As of December	er 31, 2021	As of December 31, 2020			
	Maturity Date	Interest Rate	Amount	Interest Rate	Amount		
Credit Facility							
Senior unsecured revolving loan facility	December 2026	Variable	\$ 316.4	Variable	\$ —		
Term Loans							
CDW UK term loan	March 2021	— %	_	1.445 %	56.0		
Senior unsecured term loan facility	December 2026	Variable	1,420.0	— %	_		
Senior secured term loan facility	December 2021	— %		1.900 %	1,423.4		
Total term loans			1,420.0		1,479.4		
Unsecured Senior Notes							
Senior notes due 2024	December 2024	5.500 %	575.0	5.500 %	575.0		
Senior notes due 2025	May 2025	4.125 %	600.0	4.125 %	600.0		
Senior notes due 2028	April 2028	4.250 %	600.0	4.250 %	600.0		
Senior notes due 2029	February 2029	3.250 %	700.0	3.250 %	700.0		
Senior notes due 2026	December 2026	2.670 %	1,000.0	— %	_		
Senior notes due 2028	December 2028	3.276 %	500.0	— %	_		
Senior notes due 2031	December 2031	3.569 %	1,000.0	— %	_		
Total unsecured senior notes			4,975.0		2,475.0		
Receivable financing liability			179.5		_		
Other long-term obligations			13.6		_		
Unamortized deferred financing fees			(46.0)		(27.2)		
Current maturities of long-term debt			(102.7)		(70.9)		
Total long-term debt			\$ 6,755.8		\$ 3,856.3		

As of December 31, 2021, the Company is in compliance with the covenants under its credit agreements and indentures.

Credit Facility

The Company has a variable rate senior unsecured revolving loan facility (the "Revolving Loan Facility") from which it may draw tranches denominated in US dollars, British pounds or Euros. The interest rate is based on LIBOR plus a margin or an alternate base rate plus a margin, where the margin is based on the Company's senior unsecured rating. The Revolving Loan Facility is used by the Company for borrowings, issuances of letters of credit and floorplan financing. As of December 31, 2021, the Company could have borrowed up to an additional \$1.0 billion under the Revolving Loan Facility. As of December 31, 2021, the Revolving Loan Facility had less than \$1 million of undrawn letters of credit and \$296 million reserved for the floorplan subfacility.

Term Loan

The senior unsecured term loan facility (the "Term Loan Facility") has a variable interest rate, which has effectively been capped through the use of interest rate caps. The interest rate is based on LIBOR plus a margin, where the margin is determined by the Company's senior unsecured credit rating. The Company is required to pay quarterly principal installments of \$9 million in 2022 and of \$18 million in 2023 and thereafter, with the remaining principal amount due at the maturity date.

Unsecured Senior Notes

The senior notes have a fixed interest rate, which is paid semi-annually.

(dollars in millions, except per share data, unless otherwise noted)

Receivable Financing

The majority of the receivable financing liabilities were assumed in connection with the acquisition of Sirius, which had a balance of \$160 million as of December 31, 2021. Such amounts relate to pre-acquisition transfers of certain accounts receivable to third-party financing companies that did not qualify as a sale under the terms of the agreement. While the terms of such agreements are on a nonrecourse basis, the transfers of accounts receivable could not achieve certain criteria that would allow derecognition of the accounts receivable. The proceeds from these arrangements are recognized as a liability and the associated accounts receivable remains on the Consolidated Balance Sheet until the liability is settled.

Debt Issuances and Extinguishments

In connection with the execution of the Purchase Agreement to acquire Sirius, on October 15, 2021, the Company entered into a commitment letter for a \$2.5 billion senior unsecured 364-day bridge loan facility ("Bridge Facility"), which would have been used in the event permanent financing was not obtained on or before completing the acquisition of Sirius. In lieu of borrowing under the Bridge Facility, on December 1, 2021, the Company obtained permanent financing through the issuance of \$1.0 billion aggregate principal amount of 2.670% Senior Notes due 2026, \$500 million aggregate principal amount of 3.276% Senior Notes due 2028 and \$1.0 billion aggregate principal amount of 3.569% Senior Notes due 2031. Interest on each note is payable semi-annually on June 1 and December 1 of each year, and payments commence on June 1, 2022. The net proceeds from the issuance were used to fund the Sirius acquisition and related transaction costs. The Bridge Facility was automatically terminated upon completing the acquisition of Sirius without using the Bridge Facility.

Also on December 1, 2021, the Company entered into the Revolving Loan Facility, a new five-year \$1.6 billion senior unsecured revolving loan facility. The Revolving Loan Facility replaced the senior secured asset-based revolving credit facility (the "ABL Facility"). On the same date, the Company also entered into the Term Loan Facility, a new five-year \$1.4 billion senior unsecured term loan facility. The Term Loan Facility replaced the senior secured term loan facility. The net loss recognized on extinguishment of the senior secured facilities was insignificant for the year ended December 31, 2021.

On March 26, 2021, the Company amended, extended and increased the size of the ABL Facility, prior to its extinguishment on December 1, 2021. On the same day, the Company early extinguished the remaining principal amount on the CDW UK term loan by drawing on the ABL Facility. The net loss recognized on extinguishment of CDW UK term loan was insignificant for the year ended December 31, 2021.

On August 13, 2020, the Company completed the issuance of \$700 million aggregate principal amount of 3.250% Senior Notes due 2029 at par ("2029 Senior Notes"). Interest on the 2029 Senior Notes is payable semi-annually on February 15 and August 15 of each year, and payments commenced on February 15, 2021. The net proceeds from the issuance were primarily used to redeem all of the remaining \$600 million aggregate principal amount of the 5.000% Senior Notes due September 2025 at a redemption price of 103.75% of the principal amount redeemed, plus accrued and unpaid interest to the date of redemption, to pay fees and expenses related to the issuance and redemption, and for general corporate purposes.

On April 21, 2020, the Company completed the issuance of \$600 million aggregate principal amount of 4.125% Senior Notes due 2025 at par ("2025 Senior Notes"). Interest on the 2025 Senior Notes is payable semi-annually on May 1 and November 1 of each year, and payments commenced on November 1, 2020.

(dollars in millions, except per share data, unless otherwise noted)

Total Debt Maturities

A summary of total debt maturities is as follows:

Years ending December 31,	Deb	Debt Maturities		
2022	\$	102.7		
2023		131.0		
2024		689.3		
2025		693.4		
2026		2,488.1		
Thereafter		2,800.0		
Total debt maturities	\$	6,904.5		

Fair Value

The fair values of the Senior Notes were estimated using quoted market prices for identical liabilities that are traded in over-the-counter secondary markets. The fair value of the Term Loan was estimated using dealer quotes for identical liabilities in markets that are not considered active. The Senior Notes and Term Loan were classified as Level 2 within the fair value hierarchy. The carrying value of the Revolving Loan approximates fair value.

The approximate fair values and related carrying values of the Company's long-term debt, including current maturities and excluding unamortized discount and unamortized deferred financing costs, were as follows:

	 December 31,				
	2021				
Fair value	\$ 6,996.0	\$	4,077.9		
Carrying value	6,904.5		3,954.4		

10. Income Taxes

Income before income taxes was taxed under the following jurisdictions:

		Year Ended December 31,					
		2021	2020			2019	
mestic	\$	1,186.7	\$	934.3	\$	854.1	
reign		111.1		68.0		95.6	
tal	\$	1,297.8	\$	1,002.3	\$	949.7	
	Ψ	1,	271.0	271.0 Φ	277.0 \$ 1,002.3	277.0 \$ 1,002.5 \$	

Components of Income tax expense (benefit) consist of the following:

	Year Ended December 31,					
		2021		2020		2019
Current:						
Federal	\$	235.6	\$	166.5	\$	224.7
State		52.9		49.2		56.1
Foreign		27.4		18.3		20
Total current		315.9		234.0		300.8
Deferred:						
Domestic		(8.7)		(18.8)		(83.0)
Foreign		2.0		(1.4)		(4.9)
Total deferred		(6.7)		(20.2)		(87.9)
Income tax expense	\$	309.2	\$	213.8	\$	212.9

(dollars in millions, except per share data, unless otherwise noted)

The reconciliation between the statutory tax rate expressed as a percentage of income before income taxes and the effective tax rate was as follows:

			Year Ended	December 31,		
	20:	21	2	020	2	019
Statutory federal income tax rate	\$ 272.5	21.0 %	\$ 210.5	21.0 %	\$ 199.4	21.0 %
State taxes, net of federal effect	50.3	3.9	36.0	3.6	35.4	3.7
Excess tax benefit of equity awards	(30.1)	(2.3)	(28.8)	(2.9)	(26.8)	(2.8)
Tax on foreign earnings	1.7	0.1	1.0	0.1	2.1	0.2
Effect of tax law changes	4.8	0.4	(6.8)	(0.7)	_	_
Other	10.0	0.7	1.9	0.2	2.8	0.3
Effective tax rate	\$ 309.2	23.8 %	\$ 213.8	21.3 %	\$ 212.9	22.4 %

The tax effect of temporary differences that give rise to net deferred income tax liabilities is presented below. Reclassifications have been made to conform to current year presentation.

	Dece	mber 31,
	2021	2020
Deferred tax assets:		
Contract liabilities	\$ 45.3	\$ 13.2
Equity compensation plans	22.7	20.1
Net operating loss and credit carryforwards, net	28.9	22.9
Payroll and benefits	37.6	21.8
Operating lease liabilities	51.6	47.5
Accounts receivable	18.0	26.0
Other	20.5	15.9
Total deferred tax assets	224.6	167.4
Deferred tax liabilities:		
Acquisition-related intangibles	322.2	76.5
Property and equipment	47.6	39.9
International investments	_	19.2
Operating lease right-of-use assets	35.6	32.5
Other	26.5	23.3
Total deferred tax liabilities	431.9	191.4
Deferred tax asset valuation allowance	17.0	16.9
Net deferred tax liabilities	\$ 224.3	\$ 40.9

The Company has income tax net operating losses and other carryforwards of \$39 million that do not expire and state and international tax credit carryforwards of \$20 million, which expire at various dates from 2026 through 2027.

The Company is indefinitely reinvested in its UK business, and therefore will not provide for any US deferred taxes on the earnings of the UK business. The Company is not permanently reinvested in its Canadian business and therefore has recognized deferred tax liabilities of \$2 million as of December 31, 2021 related to Canada withholding taxes on earnings of its Canadian business.

In the ordinary course of business, the Company is subject to review by domestic and foreign taxing authorities, including the Internal Revenue Service ("IRS"). In general, the Company is no longer subject to audit by the IRS or state, local, or foreign taxing authorities for tax years through 2014. Various taxing authorities are in the process of auditing income tax returns of the Company and its subsidiaries. The Company does not anticipate that any adjustments from the audits would have a material impact on its Consolidated Financial Statements.

(dollars in millions, except per share data, unless otherwise noted)

Changes in the Company's unrecognized tax benefits as of December 31, 2021, 2020 and 2019 were as follows:

	Year Ended December 31,					
		2021		2020		2019
Balance as of January 1	\$	18.3	\$	17.7	\$	15.1
Additions for tax positions related to current year		0.1		0.1		2.6
Additions for tax positions related to prior year				0.5		_
Balance as of December 31	\$	18.4	\$	18.3	\$	17.7

As of December 31, 2021, the Company had \$18 million of unrecognized tax benefits that, if recognized, would have decreased income taxes and the corresponding effective income tax rate and increased net income. The impact of recognizing these tax benefits, net of the federal income tax benefit related to unrecognized state income tax benefits, would be approximately \$15 million.

11. Leases

The Company has operating leases primarily for real estate, data centers and equipment. Remaining lease terms range from 1 to 15 years.

Supplemental Consolidated Balance Sheets information related to the Company's operating leases is as follows:

		December 31,			
Classification on the Consolidated Balance	Sheets	 2021		2020	
Assets					
	Operating lease right-of-use assets	\$ 155.6	\$	130.8	
Liabilities					
Current	Accrued expenses and other current liabilities - Other	\$ 31.7	\$	25.6	
Long-term	Long-term operating lease liabilities	 184.2		169.0	
Total lease liabilities		\$ 215.9	\$	194.6	
		 December 31,			
Lease term and discount rate		2021		2020	
Weighted average remaining lease te	rm (years)	9.0)	10.3	

Operating lease expense for the years ended December 31, 2021, 2020 and 2019 was \$50 million, \$53 million and \$93 million, respectively.

3.81 %

3.98 %

Maturities of operating lease liabilities are as follows:

Weighted average discount rate

	Decem	iber 31, 2021
2022	\$	39.4
2023		34.3
2024		28.9
2025		27.4
2026		24.1
Thereafter		106.3
Total lease payments	\$	260.4
Less: Interest		(44.5)
Present value of lease liabilities	\$	215.9

(dollars in millions, except per share data, unless otherwise noted)

Supplemental cash flow information related to operating leases is as follows:

		Year Ended December 31,				
	2	2021		2020	2019	
Cash paid for amounts included in the measurement of lease liabilities						
Operating cash flows from operating leases	\$	35.9	\$	35.8 \$	88.0	
Right-of-use assets obtained in exchange for lease obligations						
Operating leases ⁽¹⁾	\$	49.8	\$	26.7 \$	110.2	

(1) In 2021, primarily includes right-of-use assets acquired as a result of the Sirius acquisition.

12. Stockholders' Equity

Share Repurchase Program

The Company has a share repurchase program under which it may repurchase shares of its common stock in the open market or through privately negotiated other transactions, depending on share price, market conditions and other factors. The share repurchase program does not obligate the Company to repurchase any dollar amount or number of shares, and repurchases may be commenced or suspended from time to time without prior notice.

During 2021, the Company repurchased 8.7 million shares of its common stock for \$1,500 million. These repurchases occurred under the programs announced on February 7, 2019 and February 10, 2021 by which the Company's Board of Directors authorized an increase to the Company's share repurchase program by \$1.0 billion and \$1.25 billion, respectively. As of December 31, 2021, the Company has \$88 million remaining under this program.

13. Equity-Based Compensation

Equity-based compensation expense, which is recorded in Selling and administrative expenses in the Consolidated Statements of Operations was as follows:

	Year Ended December 31,					
		2021		2020		2019
Equity-based compensation expense	\$	72.6	\$	42.5	\$	48.5
Income tax benefit ⁽¹⁾		(12.2)		(7.7)		(9.8)
Equity-based compensation expense, net of tax	\$	60.4	\$	34.8	\$	38.7

(1) Represents equity-based compensation tax expense at the statutory tax rates. Excess tax benefits associated with equity awards are excluded from this disclosure and separately disclosed in Note 10 (Income Taxes).

The total unrecognized compensation cost related to non-vested awards was \$111 million as of December 31, 2021 and is expected to be recognized over a weighted-average period of 2.1 years.

2021 Long-Term Incentive Plan

During May 2021, the Company adopted the 2021 Long-Term Incentive Plan ("2021 LTIP"), which replaced the former 2013 Long-Term Incentive Plan in connection with the issuance of new equity awards ("2013 LTIP" and, together with the 2021 LTIP, the "LTIPs"). The 2021 LTIP provides for the grant of incentive stock options, nonqualified stock options, stock appreciation rights, restricted stock, restricted stock units, bonus stock and performance awards. The 2021 LTIP allows the Company to grant 6.6 million new shares of the Company's common stock in addition to the number of shares that remained available for issuance under the 2013 LTIP, and undelivered shares subject to outstanding awards granted under the 2013 LTIP that become available for future awards under the 2021 LTIP. As of December 31, 2021, 8.1 million shares were available for issuance under the 2021 LTIP. Authorized but unissued shares are reserved for issuance in connection with equity-based awards.

(dollars in millions, except per share data, unless otherwise noted)

Stock Options

The exercise price of a stock option granted is equal to the fair value of the underlying stock on the date of the grant. Stock options granted under the LTIPs have a contractual term of ten years and generally vest ratably over three years. To estimate the fair value of options granted, the Company uses the Black-Scholes option pricing model. The weighted-average assumptions used to value the stock options granted were as follows:

	 Year Ended December 31,							
	2021	2020		2019				
Grant date fair value	\$ 40.83	\$ 20.46	\$	19.26				
Volatility (1)	30.00 %	25.50 %		20.00 %				
Risk-free rate (2)	0.93 %	0.51 %		2.53 %				
Expected dividend yield	1.03 %	1.52 %		1.23 %				
Expected term (in years) (3)	5.6	6.0		6.0				

- (1) Based upon an assessment of the two-year and five-year historical and implied volatility for the Company's selected peer group, adjusted for the Company's leverage.
- (2) Based on a composite US Treasury rate.
- (3) Based on contractual term length and on historical experience of both exercised and unexercised options.

Stock option activity for the year ended December 31, 2021 was as follows:

		Av	Weighted- erage Exercise	Weighted-Average Remaining Contractual		Aggregate
Options	Number of Options		Price	Term (years)	Int	trinsic Value
Outstanding at January 1, 2021	3,965,452	\$	73.71			
Granted	545,359		156.30			
Forfeited/Expired	(58,446)		109.68			
Exercised ⁽¹⁾	(1,097,099)		63.76			
Outstanding at December 31, 2021	3,355,266	\$	89.76	6.66	\$	385.9
W	1.050.076	ď	((07	5 15	ø	270.0
Vested and exercisable at December 31, 2021	1,959,076	\$	66.97	5.45	Э	270.0
Expected to vest after December 31, 2021	1,377,326	\$	121.56	8.35	\$	114.6

(1) The total intrinsic value of stock options exercised during the years ended December 31, 2021, 2020 and 2019 was \$117 million, \$94 million and \$83 million, respectively.

Restricted Stock Units ("RSUs")

Restricted stock units represent the right to receive unrestricted shares of the Company's stock at the time of vesting. RSUs granted under the LTIPs vest either ratably over three years or cliff-vest at the end of three years. The fair value of RSUs is equal to the closing price of the Company's common stock on date of grant.

RSU activity for the year ended December 31, 2021 was as follows:

	Number of Units	Weighted- Grant-Date I	
Non-vested at January 1, 2021	92,436	\$	107.88
Granted (1)	373,530		172.96
Vested (2)	(20,340)		101.06
Forfeited	(8,102)		160.90
Non-vested at December 31, 2021	437,524	\$	163.82

(dollars in millions, except per share data, unless otherwise noted)

- (1) The weighted-average grant date fair value of RSUs granted during the years ended December 31, 2021, 2020 and 2019 was \$172.96, \$112.55 and \$103.24, respectively.
- (2) The aggregate fair value of RSUs that vested during the years ended December 31, 2021, 2020 and 2019 was \$2 million, \$12 million and \$4 million, respectively.

Performance Share Units ("PSUs")

Performance share units represent the right to receive unrestricted shares of the Company's stock at the time of vesting. PSUs granted under the LTIPs cliff-vest at the end of three years. The percentage of PSUs that shall vest will range from 0% to 200% of the number of PSUs granted based on the Company's performance against a cumulative adjusted free cash flow measure and cumulative non-GAAP net income per diluted share measure over a three-year performance period.

PSU activity for the year ended December 31, 2021 was as follows:

	Number of Units	Weight Grant-Da	ed-Average ite Fair Value
Non-vested at January 1, 2021	421,164	\$	102.07
Granted (1)	147,133		154.37
Attainment adjustment (2)	163,807		73.68
Vested (3)	(324,323)		86.17
Forfeited	(20,577)		105.34
Non-vested at December 31, 2021	387,204	\$	123.05

- (1) The weighted-average grant date fair value of PSUs granted during the years ended December 31, 2021, 2020 and 2019 was \$154.37, \$102.96 and \$101.33, respectively.
- (2) During the year ended December 31, 2021, the attainment on PSUs vested at December 31, 2020 was adjusted to reflect actual performance.
- (3) The aggregate fair value of PSUs that vested during the years ended December 31, 2021, 2020 and 2019 was \$28 million, \$24 million and \$18 million, respectively.

14. Earnings Per Share

The numerator for both basic and diluted earnings per share is Net income. The denominator for basic earnings per share is the weighted-average shares outstanding during the period.

A reconciliation of basic weighted-average shares outstanding to diluted weighted-average shares outstanding is as follows:

	Year Ended December 31,					
	2021	2020	2019			
Basic weighted-average shares outstanding	138.5	142.6	145.1			
Effect of dilutive securities (1)	2.0	2.2	2.7			
Diluted weighted-average shares outstanding (2)	140.5	144.8	147.8			

- (1) The dilutive effect of outstanding stock options, restricted stock units, performance share units and Coworker Stock Purchase Plan ("CSPP") units is reflected in the diluted weighted-average shares outstanding using the treasury stock method.
- (2) There were fewer than 0.1 million potential common shares excluded from diluted weighted-average shares outstanding for the years ended December 31, 2021, 2020 and 2019, respectively, as their inclusion would have had an anti-dilutive effect.

(dollars in millions, except per share data, unless otherwise noted)

15. Coworker Retirement and Other Compensation Benefits

Profit Sharing Plan and Other Savings Plans

The Company has a profit-sharing plan that includes a salary reduction feature established under the Internal Revenue Code Section 401(k) covering substantially all coworkers in the US. In addition, coworkers outside the US participate in other savings plans. Company contributions to the profit sharing and other savings plans are made in cash and determined at the discretion of the Board of Directors. For the years ended December 31, 2021, 2020 and 2019, the amounts expensed for these plans were \$46 million, \$28 million and \$38 million, respectively.

CSPP

The Company has a CSPP that provides the opportunity for eligible coworkers to acquire shares of the Company's common stock at a 5% discount from the closing market price on the final day of the offering period. There is no compensation expense associated with the CSPP.

16. Commitments and Contingencies

The Company is party to various legal proceedings that arise in the ordinary course of its business, which include commercial, intellectual property, employment, tort and other litigation matters. The Company is also subject to audit by federal, state, international, national, provincial and local authorities, and by various partners, group purchasing organizations and customers, including government agencies, relating to purchases and sales under various contracts. In addition, the Company is subject to indemnification claims under various contracts. From time to time, certain customers of the Company file voluntary petitions for reorganization or liquidation under the US bankruptcy laws or similar laws of the jurisdictions for the Company's business activities outside of the US. In such cases, certain pre-petition payments received by the Company could be considered preference items and subject to return to the bankruptcy administrator.

As of December 31, 2021, the Company does not believe that there is a reasonable possibility that any material loss exceeding the amounts already recognized for these proceedings and matters, if any, has been incurred. However, the ultimate resolutions of these proceedings and matters are inherently unpredictable. As such, the Company's consolidated financial statements could be adversely affected in any particular period by the unfavorable resolution of one or more of these proceedings or matters.

A subsidiary of the Company received a Civil Investigative Demand dated September 20, 2021 from the Department of Justice ("DOJ") in connection with a False Claims Act Investigation. The DOJ has requested information related to teaming agreements with OEMs. The Company is cooperating with the request and, given the early stage of the matter, the Company is currently unable to assess the probability of any outcome or the range of possible loss, if any.

17. Segment Information

The Company's segment information reflects the way the chief operating decision maker uses internal reporting to evaluate business performance, allocate resources and manage operations.

The Company has three reportable segments: Corporate, which is comprised primarily of private sector business customers with more than 250 employees in the US, Small Business, primarily servicing private sector business customers with up to 250 employees in the US, and Public, which is comprised of government agencies and education and healthcare institutions in the US. The Company has two other operating segments: CDW UK and CDW Canada, both of which do not meet the reportable segment quantitative thresholds and, accordingly, are included in an all other category ("Other").

The Company has centralized logistics and headquarters functions that provide services to the segments. The logistics function includes purchasing, distribution and fulfillment services to support the Corporate, Small Business and Public segments. As a result, costs and intercompany charges associated with the logistics function are fully allocated to all of these segments based on a percent of Net sales. The centralized headquarters function provides services in areas such as accounting, information technology, marketing, legal and coworker services. Headquarters function costs that are not allocated to the segments are included under the heading of "Headquarters" in the tables below.

Segment information for Total assets and capital expenditures is not presented, as such information is not used in measuring segment performance or allocating resources between segments.

(dollars in millions, except per share data, unless otherwise noted)

The segment information presented below includes the results of operations from Sirius since its acquisition on December 1, 2021.

Selected Segment Financial Information

Information about the Company's segments for the years ended December 31, 2021, 2020 and 2019 is as follows:

	Small Corporate Business			Public	c Other			Headquarters		Total		
2021:		orporate	_	Business		1 40110		- CLIEF		aquarters		70141
Net sales	\$	8,179.7	\$	1,870.1	\$	8,183.6	\$	2,587.4	\$	_	\$	20,820.8
Operating income (loss)		697.3		167.7		606.7		115.8		(168.5)		1,419.0
Depreciation and amortization expense		(22.9)		(4.1)		(57.2)		(34.4)		(72.6)		(191.2)
2020:												
Net sales	\$	6,846.0	\$	1,397.1	\$	8,137.7	\$	2,086.7	\$	_	\$	18,467.5
Operating income (loss)		489.5		99.0		678.2		65.9		(153.4)		1,179.2
Depreciation and amortization expense		(73.2)		(18.3)		(229.7)		(32.5)		(71.9)		(425.6)
2019:												
Net sales	\$	7,499.0	\$	1,510.3	\$	6,864.8	\$	2,158.3	\$	_	\$	18,032.4
Operating income (loss)		585.1		107.5		475.0		101.6		(135.6)		1,133.6
Depreciation and amortization expense		(86.9)		(22.5)		(56.3)		(31.2)		(70.2)		(267.1)

(dollars in millions, except per share data, unless otherwise noted)

Geographic Areas and Revenue Mix

	Year Ended December 31, 2021									
	(Corporate	Sm	all Business		Public	Other		Total	
Geography ⁽¹⁾										
United States	\$	8,165.4	\$	1,870.1	\$	8,183.6	\$ 19.7	\$	18,238.8	
Rest of World		14.3		_		_	2,567.7		2,582.0	
Total Net sales		8,179.7		1,870.1		8,183.6	2,587.4		20,820.8	
			-			_				
Major Product and Services										
Hardware		6,427.9		1,587.9		6,827.1	1,926.7		16,769.6	
Software		1,172.4		211.0		1,017.3	401.7		2,802.4	
Services		510.1		49.1		321.5	245.4		1,126.1	
Other ⁽²⁾		69.3		22.1		17.7	13.6		122.7	
Total Net sales		8,179.7		1,870.1		8,183.6	2,587.4		20,820.8	
	-				_	·	·	==	;	
Sales by Channel										
Corporate		8,179.7		_			_		8,179.7	
Small Business		_		1,870.1		_	_		1,870.1	
Government		_		_		2,155.6	_		2,155.6	
Education		_		_		4,108.7	_		4,108.7	
Healthcare		_		_		1,919.3	_		1,919.3	
Other		_		_		_	2,587.4		2,587.4	
Total Net sales		8,179.7		1,870.1		8,183.6	2,587.4		20,820.8	
Timing of Revenue Recognition										
Transferred at a point in time where CDW is										
principal		7,332.3		1,734.7		7,634.3	2,288.7		18,990.0	
Transferred at a point in time where CDW is agent		517.5		112.3		336.6	83.2		1,049.6	
Transferred over time where CDW is principal		329.9		23.1		212.7	215.5		781.2	
Total Net sales	\$	8,179.7	\$	1,870.1	\$	8,183.6	\$ 2,587.4	\$	3 20,820.8	

⁽¹⁾ Net sales by geography is generally based on the ship-to address with the exception of certain services that may be performed at, or on behalf of, multiple locations. Such service arrangements are categorized based on the bill-to address.

⁽²⁾ Includes items such as delivery charges to customers.

(dollars in millions, except per share data, unless otherwise noted)

Year Ended December 31, 2020 Corporate Small Business Public Other Total Geography⁽¹⁾ United States \$ 6,823.6 \$ 1,397.1 \$ 8,137.7 \$ 20.8 16,379.2 Rest of World 22.4 2,065.9 2,088.3 Total Net sales 6,846.0 1,397.1 8,137.7 2,086.7 18,467.5 **Major Product and Services** Hardware 6,844.0 1,544.1 14,833.4 5,289.2 1,156.1 Software 1,088.3 189.3 982.8 320.6 2,581.0 Services 31.5 269.8 211.8 400.8 913.9 Other(2) 41.1 67.7 20.2 10.2 139.2 6,846.0 1,397.1 8,137.7 2,086.7 18,467.5 Total Net sales Sales by Channel 6,846.0 Corporate 6,846.0 1,397.1 **Small Business** 1,397.1 Government 2,978.5 2,978.5 Education 3,458.1 3,458.1 Healthcare 1,701.1 1,701.1 Other 2,086.7 2,086.7 Total Net sales 1,397.1 8,137.7 2,086.7 6,846.0 18,467.5 **Timing of Revenue Recognition** Transferred at a point in time where CDW is 1,301.3 1,835.5 16,754.9 6,140.7 7,477.4 principal Transferred at a point in time where CDW is 457.4 84.5 292.5 61.6 896.0 agent Transferred over time where CDW is principal 247.9 11.3 367.8 189.6 816.6 Total Net sales 6,846.0 1,397.1 8,137.7 2,086.7 18,467.5

⁽¹⁾ Net sales by geography is generally based on the ship-to address with the exception of certain services that may be performed at, or on behalf of, multiple locations. Such service arrangements are categorized based on the bill-to address.

⁽²⁾ Includes items such as delivery charges to customers.

(dollars in millions, except per share data, unless otherwise noted)

	Year Ended December 31, 2019									
	Corporate	Small Business	Public	Other	Total					
Geography ⁽¹⁾										
United States	\$ 7,485.7	\$ 1,510.3	\$ 6,864.8	\$ 32.5	\$ 15,893.3					
Rest of World	13.3			2,125.8	2,139.1					
Total Net sales	7,499.0	1,510.3	6,864.8	2,158.3	18,032.4					
Major Product and Services										
Hardware	5,963.7	1,264.7	5,624.9	1,628.9	14,482.2					
Software	1,069.2	196.0	1,019.6	300.2	2,585.0					
Services	395.8	28.5	199.0	217.6	840.9					
Other ⁽²⁾	70.3	21.1	21.3	11.6	124.3					
Total Net sales	7,499.0	1,510.3	6,864.8	2,158.3	18,032.4					
Sales by Channel										
Corporate	7,499.0	_	_	_	7,499.0					
Small Business	_	1,510.3	_	_	1,510.3					
Government	_	_	2,519.3	_	2,519.3					
Education	_	_	2,411.6	_	2,411.6					
Healthcare	_	_	1,933.9	_	1,933.9					
Other	_	_	_	2,158.3	2,158.3					
Total Net sales	7,499.0	1,510.3	6,864.8	2,158.3	18,032.4					
Timing of Revenue Recognition										
Transferred at a point in time where CDW is principal	6,818.7	1,423.1	6,410.2	1,900.6	16,552.6					
Transferred at a point in time where CDW is agent	446.1	80.0	248.5	59.6	834.2					
Transferred over time where CDW is principal	234.2	7.2	206.1	198.1	645.6					
Total Net sales	\$ 7,499.0	\$ 1,510.3	\$ 6,864.8	\$ 2,158.3	\$ 18,032.4					

⁽¹⁾ Net sales by geography is generally based on the ship-to address with the exception of certain services that may be performed at, or on behalf of, multiple locations. Such service arrangements are categorized based on the bill-to address.

⁽²⁾ Includes items such as delivery charges to customers.

(dollars in millions, except per share data, unless otherwise noted)

The following table presents Net sales by major category for the years ended December 31, 2021, 2020 and 2019. Categories are based upon internal classifications.

			Year Ended	December 31,			
	 20)21	 20	020	2019		
	Net Sales	Percentage of Total Net Sales	Net Sales	Percentage of Total Net Sales		Net Sales	Percentage of Total Net Sales
Notebooks/Mobile Devices	\$ 6,659.4	32.0 %	\$ 5,486.2	29.7 %	\$	4,344.9	24.1 %
Netcomm Products	1,950.9	9.4	1,955.0	10.6		2,189.1	12.1
Desktops	1,203.6	5.8	1,132.4	6.1		1,547.3	8.6
Video	1,605.0	7.7	1,190.8	6.4		1,272.9	7.1
Enterprise and Data Storage (Including Drives)	992.1	4.8	947.4	5.1		1,147.6	6.4
Other Hardware	4,358.6	20.9	4,121.6	22.3		3,980.4	22.1
Total Hardware	16,769.6	80.6	14,833.4	80.2		14,482.2	80.4
Software ⁽¹⁾	 2,802.4	13.5	2,581.0	14.0		2,585.0	14.3
Services ⁽¹⁾	1,126.1	5.4	913.9	4.9		840.9	4.7
Other ⁽²⁾	122.7	0.5	139.2	0.9		124.3	0.6
Total Net sales	\$ 20,820.8	100.0 %	\$ 18,467.5	100.0 %	\$	18,032.4	100.0 %

⁽¹⁾ Certain software and services revenues are recorded on a net basis for accounting purposes. As a result, the category percentage of net revenues is not representative of the category percentage of gross profits.

⁽²⁾ Includes items such as delivery charges to customers.

SCHEDULE II – VALUATION AND QUALIFYING ACCOUNTS Years ended December 31, 2021, 2020 and 2019

(dollars in millions)

Allowance for credit losses:	Balance at Beginning of Period	Charged to Costs and Expenses	Deductions (1)	Balance at End of Period
Year Ended December 31, 2021	\$ 29.6	\$ (5.4)	\$ (3.8)	\$ 20.4
Year Ended December 31, 2020	7.9	30.9	(9.2)	29.6
Year Ended December 31, 2019	7.0	2.2	(1.3)	7.9

(1) Primarily includes write-offs of uncollectible accounts.

Item 9. Changes in and Disagreements with Accountants on Accounting and Financial Disclosure

None.

Item 9A. Controls and Procedures

Evaluation of Disclosure Controls and Procedures

The Company's management, with the participation of the Company's

Chief Executive Officer and Chief Financial Officer, has evaluated the effectiveness of the Company's disclosure controls and procedures (as such term is defined in Rule 13a-15(e) or Rule 15d-15(e) under the Securities Exchange Act of 1934, as amended (the "Exchange Act")) as of the end of the period covered by this report. Based on such evaluation, the Company's management, including the Company's Chief Executive Officer and Chief Financial Officer, has concluded that, as of the end of such period, the Company's disclosure controls and procedures were effective in recording, processing, summarizing, and reporting, on a timely basis, information required to be disclosed by the Company in the reports that it files or submits under the Exchange Act, and that information is accumulated and communicated to the Company's management, including the Company's Chief Executive Officer and Chief Financial Officer, as appropriate to allow timely discussions regarding required disclosure.

Management's Annual Report on Internal Control over Financial Reporting

Management is responsible for establishing and maintaining adequate internal control over financial reporting, as defined in Rule 13a-15(f) and 15d-15(f) under the Exchange Act. Because of its inherent limitations, internal control over financial reporting may not prevent or detect misstatements and can provide only reasonable assurance with respect to financial statement preparation and presentation. Also, projections of any evaluation of effectiveness to future periods are subject to the risk that controls may become inadequate because of changes in conditions, or that the degree of compliance with policies or procedures may deteriorate.

Management assessed the effectiveness of the Company's internal control over financial reporting as of December 31, 2021. Management based this assessment on the criteria set forth by the Committee of Sponsoring Organizations of the Treadway Commission (COSO) in "Internal Control — Integrated Framework (2013 framework)."

As permitted by the Securities and Exchange Commission guidance for newly acquired businesses, management excluded its assessment of internal control over financial reporting for Sirius Computer Solutions, Inc., which was acquired on December 1, 2021 and accounts for approximately 30% of consolidated total assets and 1% of consolidated net sales as of and for the year ended December 31, 2021.

Based on its assessment, which excluded an internal control assessment of Sirius Computer Solutions, Inc., management concluded that, as of December 31, 2021, the Company's internal control over financial reporting is effective.

Ernst & Young LLP, independent registered public accounting firm, has audited the Consolidated Financial Statements of the Company and the Company's internal control over financial reporting and has included their reports herein.

Changes in Internal Control over Financial Reporting

There have been no changes in our internal control over financial reporting during the quarter ended December 31, 2021 that have materially affected or are reasonably likely to materially affect, our internal control over financial reporting.

Report of Independent Registered Public Accounting Firm

To the Stockholders and the Board of Directors of CDW Corporation and subsidiaries

Opinion on Internal Control Over Financial Reporting

We have audited CDW Corporation and subsidiaries' internal control over financial reporting as of December 31, 2021, based on criteria established in Internal Control—Integrated Framework issued by the Committee of Sponsoring Organizations of the Treadway Commission (2013 framework) (the COSO criteria). In our opinion, CDW Corporation and subsidiaries (the Company) maintained, in all material respects, effective internal control over financial reporting as of December 31, 2021, based on the COSO criteria.

As indicated in the accompanying Management's Annual Report on Internal Control over Financial Reporting, management's assessment of and conclusion on the effectiveness of internal control over financial reporting did not include the internal controls of Sirius Computer Solutions, Inc., which is included in the 2021 consolidated financial statements of the Company and constituted 30% of total assets as of December 31, 2021. Our audit of internal control over financial reporting of the Company also did not include an evaluation of the internal control over financial reporting of Sirius Computer Solutions, Inc.

We also have audited, in accordance with the standards of the Public Company Accounting Oversight Board (United States) (PCAOB), the consolidated balance sheets of the Company as of December 31, 2021 and 2020, and the related consolidated statements of operations, comprehensive income, stockholders' equity and cash flows for each of the three years in the period ended December 31, 2021, and the related notes and the financial statement schedule listed in the Index at Item 15 (a) (2) and our report dated February 28, 2022 expressed an unqualified opinion thereon.

Basis for Opinion

The Company's management is responsible for maintaining effective internal control over financial reporting and for its assessment of the effectiveness of internal control over financial reporting included in the accompanying Management's Annual Report on Internal Control Over Financial Reporting. Our responsibility is to express an opinion on the Company's internal control over financial reporting based on our audit. We are a public accounting firm registered with the PCAOB and are required to be independent with respect to the Company in accordance with the U.S. federal securities laws and the applicable rules and regulations of the Securities and Exchange Commission and the PCAOB.

We conducted our audit in accordance with the standards of the PCAOB. Those standards require that we plan and perform the audit to obtain reasonable assurance about whether effective internal control over financial reporting was maintained in all material respects.

Our audit included obtaining an understanding of internal control over financial reporting, assessing the risk that a material weakness exists, testing and evaluating the design and operating effectiveness of internal control based on the assessed risk, and performing such other procedures as we considered necessary in the circumstances. We believe that our audit provides a reasonable basis for our opinion.

Definition and Limitations of Internal Control Over Financial Reporting

A company's internal control over financial reporting is a process designed to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles. A company's internal control over financial reporting includes those policies and procedures that (1) pertain to the maintenance of records that, in reasonable detail, accurately and fairly reflect the transactions and dispositions of the assets of the company; (2) provide reasonable assurance that transactions are recorded as necessary to permit preparation of financial statements in accordance with generally accepted accounting principles, and that receipts and expenditures of the company are being made only in accordance with authorizations of management and directors of the company; and (3) provide reasonable assurance regarding prevention or timely detection of unauthorized acquisition, use, or disposition of the company's assets that could have a material effect on the financial statements.

Because of its inherent limitations, internal control over financial reporting may not prevent or detect misstatements. Also, projections of any evaluation of effectiveness to future periods are subject to the risk that controls may become inadequate because of changes in conditions, or that the degree of compliance with the policies or procedures may deteriorate.

/s/ Ernst & Young LLP Chicago, Illinois February 28, 2022

Table of Contents

Item 9B. Other Information

None.

Item 9C. Disclosure Regarding Foreign Jurisdictions that Prevent Inspections

None.

PART III

Item 10. Directors, Executive Officers and Corporate Governance

We have adopted The CDW Way Code, our code of business conduct and ethics, that is applicable to all of our coworkers and directors. A copy of The CDW Way Code is available on our website at www.cdw.com. Within The CDW Way Code is a Financial Integrity Code of Ethics that sets forth an even higher standard applicable to our executives, officers, members of our internal disclosure committee and all managers and above in our finance department. We intend to disclose any substantive amendments to, or waivers from, The CDW Way Code by posting such information on our website or by filing a Form 8-K, in each case to the extent such disclosure is required by the rules of the SEC or Nasdaq.

See Part I - "Information about our Executive Officers" for the biographical information of our executive officers, which is incorporated by reference in this Item 10. Other information required under this Item 10 is incorporated herein by reference to our definitive proxy statement for our 2022 annual meeting of stockholders on May 19, 2022 ("2022 Proxy Statement"), which we will file with the SEC on or before April 30, 2022.

Item 11. Executive Compensation

Information required under this Item 11 is incorporated herein by reference to the 2022 Proxy Statement.

Item 12. Security Ownership of Certain Beneficial Owners and Management and Related Stockholder Matters

Information required under this Item 12 is incorporated herein by reference to the 2022 Proxy Statement.

Item 13. Certain Relationships and Related Transactions, and Director Independence

Information required under this Item 13 is incorporated herein by reference to the 2022 Proxy Statement.

Item 14. Principal Accountant Fees and Services

Information required under this Item 14 is incorporated herein by reference to the 2022 Proxy Statement.

PART IV

Item 15. Exhibits and Financial Statement Schedules

(a) Financial Statements and Schedules

The following documents are filed as part of this report:

(1) Consolidated Financial Statements:

	Page
Report of Independent Registered Public Accounting Firm	<u>43</u>
Consolidated Balance Sheets as of December 31, 2021 and 2020	<u>46</u>
Consolidated Statements of Operations for the years ended December 31, 2021, 2020 and 2019	<u>47</u>
Consolidated Statements of Comprehensive Income for the years ended December 31, 2021, 2020 and 2019	<u>48</u>
Consolidated Statements of Stockholders' Equity for the years ended December 31, 2021, 2020 and 2019	<u>49</u>
Consolidated Statements of Cash Flows for the years ended December 31, 2021, 2020 and 2019	<u>50</u>
Notes to Consolidated Financial Statements	<u>51</u>
(2) Financial Statement Schedules:	
	Page

<u>Schedule II – Valuation and Qualifying Accounts</u>

All other schedules are omitted since the required information is not present or is not present in amounts sufficient to require submission of the schedule, or because the information required is included in the Consolidated Financial Statements or notes thereto.

<u>80</u>

(b) Exhibits

Exhibit Number	Description
2.1	Purchase and Sale Agreement, dated as of October 15, 2021, by and between Sirius Computer Solutions Holdco, LP and CDW LLC previously filed as Exhibit 2.1 with CDW Corporation's Form 8-K filed on October 18, 2021 and incorporated herein by reference.
3.1	Sixth Restated Certificate of Incorporation of CDW Corporation, previously filed as Exhibit 3.2 with CDW Corporation's Form 8-K filed on May 21, 2021 and incorporated herein by reference.
3.2	Amended and Restated By-Laws of CDW Corporation, previously filed as Exhibit 3.1 with CDW Corporation's Form 8-K filed on December 23, 2019 and incorporated herein by reference.
3.3	Articles of Organization of CDW LLC, previously filed as Exhibit 3.3 with CDW Corporation's Form S-4 filed on September 7, 2010 and incorporated herein by reference.
3.4	Amended and Restated Limited Liability Company Agreement of CDW LLC, previously filed as Exhibit 3.4 with CDW Corporation's Form S-4 filed on September 7, 2010 and incorporated herein by reference.
3.5	Certificate of Incorporation of CDW Finance Corporation, previously filed as Exhibit 3.5 with CDW Corporation's Form S-4 filed on September 7, 2010 and incorporated herein by reference.
3.6	Amended and Restated By-Laws of CDW Finance Corporation, previously filed as Exhibit 3.1 with CDW Corporation's Form 10-Q filed on May 8, 2015 and incorporated herein by reference.
3.7	Articles of Organization of CDW Technologies LLC, previously filed as Exhibit 3.7 with CDW Corporation's Form 10-K filed on February 25, 2016 and incorporated herein by reference.
3.8	Operating Agreement of CDW Technologies LLC, previously filed as Exhibit 3.8 with CDW Corporation's Form 10-K filed on February 25, 2016 and incorporated herein by reference.
3.9	Articles of Organization of CDW Direct, LLC, previously filed as Exhibit 3.9 with CDW Corporation's Form S-4 filed on September 7, 2010 and incorporated herein by reference.

Exhibit Number	Description
3.10	Amended and Restated Limited Liability Company Agreement of CDW Direct, LLC, previously filed as Exhibit 3.10 with CDW Corporation's Form S-4 filed on September 7, 2010 and incorporated herein by reference.
3.11	Articles of Organization of CDW Government LLC, previously filed as Exhibit 3.11 with CDW Corporation's Form S-4 filed on September 7, 2010 and incorporated herein by reference.
3.12	Amended and Restated Limited Liability Company Agreement of CDW Government LLC, previously filed as Exhibit 3.12 with CDW Corporation's Form S-4 filed on September 7, 2010 and incorporated herein by reference.
3.13	Articles of Organization of CDW Logistics LLC, previously filed as Exhibit 3.13 with CDW Corporation's Form 10-K filed on February 28, 2020 and incorporated herein by reference.
3.14	<u>Limited Liability Company Agreement of CDW Logistics LLC, previously filed as Exhibit 3.14 with CDW Corporation's Form 10-K filed on February 28, 2020 and incorporated herein by reference.</u>
3.15	Articles of Organization of Amplified IT LLC, previously filed as Exhibit 3.15 with CDW Corporation's Post-Effective Amendment No. 1 to Form S-3 filed on November 23, 2021 and incorporated herein by reference.
3.16	Operating Agreement of Amplified IT LLC, previously filed as Exhibit 3.15 with CDW Corporation's Post-Effective Amendment No. 1 to Form S-3 filed on November 23, 2021 and incorporated herein by reference.
4.1*	Description of CDW Corporation's Common Stock.
4.2	Specimen Common Stock Certificate, previously filed as Exhibit 4.1 with CDW Corporation's Amendment No. 3 to Form S-1 filed on June 25, 2013 and incorporated herein by reference.
4.3	Base Indenture, dated as of December 1, 2014, by and among CDW LLC, CDW Finance Corporation, CDW Corporation, the other guarantors party thereto and U.S. Bank National Association as trustee, previously filed as Exhibit 4.1 with CDW Corporation's Form 8-K filed on December 1, 2014 and incorporated herein by reference.
4.4	First Supplemental Indenture, dated as of December 1, 2014, by and among CDW LLC, CDW Finance Corporation, CDW Corporation, the other guarantors party thereto and U.S. Bank National Association as trustee, previously filed as Exhibit 4.2 with CDW Corporation's Form 8-K filed on December 1, 2014 and incorporated herein by reference.
4.5	Form of 5.5% Senior Note (included as Exhibit A to Exhibit 4.4), previously filed as Exhibit 4.3 with CDW Corporation's Form 8-K filed on December 1, 2014 and incorporated herein by reference.
4.6	Fourth Supplemental Indenture, dated as of September 26, 2019, by and among the CDW LLC, CDW Finance Corporation, CDW Corporation, the other guarantors party thereto and U.S. Bank National Association as trustee, previously filed as Exhibit 4.2 with CDW Corporation's Form 8-K filed on September 26, 2019 and incorporated herein by reference.
4.7	Form of 4.250% Senior Note (included as Exhibit A to Exhibit 4.6) previously filed as Exhibit 4.3 with CDW Corporation's Form 8-K filed on September 26, 2019 and incorporated herein by reference.
4.8	Fifth Supplemental Indenture, dated as of April 21, 2020, by and among CDW LLC, CDW Finance Corporation, CDW Corporation, the other guarantors party thereto and U.S. Bank National Association as trustee, previously filed as Exhibit 4.2 with CDW Corporation's Form 8-K filed on April 21, 2020 and incorporated herein by reference.
4.9	Form of 4.125% Senior Note (included as Exhibit A to Exhibit 4.8), previously filed as Exhibit 4.3 with CDW Corporation's Form 8-K filed on April 21, 2020 and incorporated herein by reference.
4.10	Sixth Supplemental Indenture, dated as of August 13, 2020, by and among CDW LLC, CDW Finance Corporation, CDW Corporation, the other guarantors party thereto and U.S. Bank National Association as trustee, previously filed as Exhibit 4.2 with CDW Corporation's Form 8-K filed on August 13, 2020 and incorporated herein by reference.

Exhibit Number	Description			
4.11	Form of 3.25% Senior Note (included as Exhibit A to Exhibit 4.10), previously filed as Exhibit 4.3 with CDW Corporation's Form 8-K filed on August 13, 2020 and incorporated herein by reference.			
4.12	Seventh Supplemental Indenture, dated as of December 1, 2021, by and among CDW LLC, CDW Finance Corporation, CDW Corporation, the other guarantors party thereto and U.S. Bank National Association, previously filed as Exhibit 4.2 with CDW Corporation's Form 8-K filed on December 1, 2021 and incorporated herein by reference.			
4.13	Form of 2.670% Senior Note (included as Exhibit A to Exhibit 4.12) previously filed as Exhibit 4.3 with CDW Corporation's Form 8-K filed on December 1, 2021 and incorporated herein by reference.			
4.14	Eighth Supplemental Indenture, dated as of December 1, 2021, by and among CDW LLC, CDW Finance Corporation, CDW Corporation, the other guarantors party thereto and U.S. Bank National Association, previously filed as Exhibit 4.4 with CDW Corporation's Form 8-K filed on December 1, 2021 and incorporated herein by reference.			
4.15	Form of 3.276% Senior Note (included as Exhibit A to Exhibit 4.14) previously filed as Exhibit 4.5 with CDW Corporation's Form 8-K filed on December 1, 2021 and incorporated herein by reference.			
4.16	Ninth Supplemental Indenture, dated as of December 1, 2021, by and among CDW LLC, CDW Finance Corporation, CDW Corporation, the other guarantors party thereto and U.S. Bank National Association, previously filed as Exhibit 4.6 with CDW Corporation's Form 8-K filed on December 1, 2021 and incorporated herein by reference.			
4.17	Form of 3.569% Senior Note (included as Exhibit A to Exhibit 4.16) previously filed as Exhibit 4.7 with CDW Corporation's Form 8-K filed on December 1, 2021 and incorporated herein by reference.			
10.1	Credit Agreement, dated as of December 1, 2021, by and among CDW LLC, the lenders from time to time party thereto, JPMorgan Chase Bank, N.A., as administrative agent, and the joint lead arrangers, joint bookrunners, co-syndication agents and co-documentation agents party thereto, previously filed as Exhibit 10.1 with CDW Corporation's Form 8-K filed on December 2, 2021 and incorporated herein by reference.			
10.2	Revolving Credit Agreement, dated as of December 1, 2021, by and among CDW LLC, CDW Finance Holdings Limited, the guarantors party thereto, the lenders party thereto, JPMorgan Chase Bank, N.A., as administrative agent, Wells Fargo Commercial Distribution Finance, LLC, as floorplan funding agent, and the joint lead arrangers, joint bookrunners, co-syndication agents and co-documentation agents party thereto, previously filed as Exhibit 10.2 with CDW Corporation's Form 8-K filed on December 2, 2021 and incorporated herein by reference.			
10.3§	Compensation Protection Agreement, effective as of January 1, 2020, by and among CDW Corporation, CDW LLC and Christine A. Leahy, previously filed as Exhibit 10.1 with CDW Corporation's Form 8-K filed on March 11, 2019 and incorporated herein by reference.			
10.4§	Form of Compensation Protection Agreement (executive officers other than Christine A. Leahy), previously filed as Exhibit 10.1 with CDW Corporation's Form 10-Q filed on November 3, 2021 and incorporated herein by reference.			
10.5§	Form of Noncompetition Agreement under the Compensation Protection Agreement, previously filed as Exhibit 10.3 with CDW Corporation's Form 8-K filed on March 14, 2016 and incorporated herein by reference.			
10.6§	Letter Agreement, dated as of September 13, 2011, by and between CDW Direct, LLC and Christina M. Corley, previously filed as Exhibit 10.31 with CDW Corporation's Form 10-K filed on March 9, 2012 and incorporated herein by reference.			
10.7§	Form of Indemnification Agreement by and between CDW Corporation and its directors and executive officers, previously filed as Exhibit 10.32 with CDW Corporation's Amendment No. 2 to Form S-1 filed on June 14, 2013 and incorporated herein by reference.			
10.8§	CDW Corporation Senior Management Incentive Plan, as Amended and Restated Effective January 1, 2020, previously filed as Exhibit 10.1 with CDW Corporation's Form 10-Q filed on August 5, 2020 and incorporated herein by reference.			
10.9§	CDW Corporation Amended and Restated 2013 Long-Term Incentive Plan, previously filed as Exhibit 10.1 with CDW Corporation's Form 8-K filed on May 19, 2016 and incorporated herein by reference.			

Exhibit Number	Description		
10.10§	CDW Corporation 2021 Long-Term Incentive Plan, previously filed as Exhibit 10.1 with CDW Corporation's Form 8-K filed on May 19, 2021 and incorporated herein by reference.		
10.11§	CDW Corporation Coworker Stock Purchase Plan (As Amended and Restated, Effective May 20, 2021), previously filed as Exhibit 10.2 with CDW Corporation's Form 10-Q filed on August 4, 2021 and incorporated herein by reference.		
10.12§	Form of Stock Option Agreement (executive officers) under the CDW Corporation Amended and Restated 2013 Long-Term Incentive Plan, previously filed as Exhibit 10.22 with CDW Corporation's Form 10-K filed on March 1, 2017 and incorporated herein by reference.		
10.13§	Form of Stock Option Agreement (other than executive officers) under the CDW Corporation Amended and Restated 2013 Long-Term Incentive Plan, previously filed as Exhibit 10.22 with CDW Corporation's Form 10-K filed on March 1, 2018 and incorporated herein by reference.		
10.14§*	Form of Stock Option Agreement (executive officers) under the CDW Corporation 2021 Long-Term Incentive Plan.		
10.15§	Form of Performance Share Unit Award Agreement (executive officers) under the CDW Corporation Amended and Restated 2013 Long-Term Incentive Plan, previously filed as Exhibit 10.2 with CDW Corporation's Form 10-Q filed on May 5, 2021 and incorporated herein by reference.		
10.16§	Form of Performance Share Unit Award Agreement (other than executive officers) under the CDW Corporation Amended and Restated 2013 Long-Term Incentive Plan, previously filed as Exhibit 10.24 with CDW Corporation's Form 10-K filed on March 1, 2018 and incorporated herein by reference.		
10.17§*	Form of Performance Share Unit Award Agreement (executive officers) under the CDW Corporation 2021 Long-Term Incentive Plan.		
10.18§	Form of Restricted Stock Unit Award Agreement under the CDW Corporation Amended and Restated 2013 Long-Term Incentive Plan, previously filed as Exhibit 10.20 with CDW Corporation's Form 10-K filed on February 28, 2020 and incorporated herein by reference.		
10.19§*	Form of Restricted Stock Unit Award Agreement under the CDW Corporation 2021 Long-Term Incentive Plan.		
10.20§	Form of Non-Employee Director Restricted Stock Unit Award Agreement under the CDW Corporation Amended and Restated 2013 Long-Term Incentive Plan, previously filed as Exhibit 10.2 with CDW Corporation's Form 10-Q filed on May 6, 2020 and incorporated herein by reference.		
10.21§	Form of Non-Executive Chair Retainer Restricted Stock Unit Award Agreement under the CDW Corporation Amended and Restated 2013 Long-Term Incentive Plan, previously filed as Exhibit 10.1 with CDW Corporation's Form 10-Q filed on May 6, 2020 and incorporated herein by reference.		
10.22§	CDW LLC Nonqualified Deferred Compensation Plan, previously filed as Exhibit 10.3 with CDW Corporation's Form 10-Q filed on August 4, 2021 and incorporated herein by reference.		
10.23§*	CDW Director Deferred Compensation Plan.		
10.24§	Letter Agreement, dated as of February 12, 2018 by and between CDW Limited and Collin B. Kebo, previously filed as Exhibit 10.28 with CDW Corporation's Form 10-K filed on March 1, 2018 and incorporated herein by reference.		
10.25§	Letter of Understanding, dated as of September 29, 2021, by and among CDW Corporation, CDW LLC and Collin B. Kebo, previously filed as Exhibit 10.2 with CDW Corporation's Form 10-Q filed on November 3, 2021 and incorporated herein by reference.		
21.1*	<u>List of subsidiaries.</u>		
22.1*	List of Issuer and Guarantor subsidiaries.		
23.1*	Consent of Ernst & Young LLP.		

Exhibit Number	Description
31.1*	Certification of Chief Executive Officer pursuant to Rule 13a-14(a) or Rule 15d-14(a) under the Securities Exchange Act of 1934.
31.2*	Certification of Chief Financial Officer pursuant to Rule 13a-14(a) or Rule 15d-14(a) under the Securities Exchange Act of 1934.
32.1**	Certification of Chief Executive Officer pursuant to 18 U.S.C. 1350.
32.2**	Certification of Chief Financial Officer pursuant to 18 U.S.C. 1350.
101.INS*	XBRL Instance Document
101.SCH*	Inline XBRL Taxonomy Extension Schema Document
101.CAL*	Inline XBRL Taxonomy Extension Calculation Linkbase Document
101.DEF*	Inline XBRL Taxonomy Extension Definition Linkbase Document
101.LAB*	Inline XBRL Taxonomy Extension Label Linkbase Document
101.PRE*	Inline XBRL Taxonomy Extension Presentation Linkbase Document
104*	Cover Page Interactive Data File (embedded within the Inline XBRL document)

Filed herewith

^{**} These items are furnished and not filed.

§ A management contract or compensatory arrangement required to be filed as an exhibit pursuant to Item 601 of Regulation S-K.

Item 16. Form 10-K Summary

None.

SIGNATURES

Pursuant to the requirements of Section 13 or 15(d) of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned, thereunto duly authorized.

CDW CORPORATION

Date: February 28, 2022 By: /s/ Christine A. Leahy

Christine A. Leahy

President and Chief Executive Officer

Pursuant to the requirements of the Securities Exchange Act of 1934, this report has been signed below by the following persons on behalf of the registrant and in the capacities and on the dates indicated.

Signature	Title	Date
/s/ Christine A. Leahy Christine A. Leahy	President and Chief Executive Officer (principal executive officer) and Director	February 28, 2022
/s/ Albert J. Miralles Albert J. Miralles	Senior Vice President and Chief Financial Officer (principal financial officer)	February 28, 2022
/s/ Ilaria Mocciaro Ilaria Mocciaro	Vice President, Controller and Chief Accounting Officer (principal accounting officer)	February 28, 2022
/s/ David W. Nelms David W. Nelms	Non-Executive Chairman of the Board	February 28, 2022
/s/ Virginia C. Addicott Virginia C. Addicott	Director	February 28, 2022
/s/ James A. Bell James A. Bell	Director	February 28, 2022
/s/ Lynda M. Clarizio Lynda M. Clarizio	Director	February 28, 2022
/s/ Paul J. Finnegan Paul J. Finnegan	Director	February 28, 2022
/s/ Anthony R. Foxx Anthony R. Foxx	Director	February 28, 2022
/s/ Sanjay Mehrotra Sanjay Mehrotra	Director	February 28, 2022
/s/ Joseph R. Swedish Joseph R. Swedish	Director	February 28, 2022
/s/ Donna F. Zarcone Donna F. Zarcone	Director	February 28, 2022

DESCRIPTION OF THE REGISTRANT'S SECURITIES REGISTERED PURSUANT TO SECTION 12 OF THE SECURITIES EXCHANGE ACT OF 1934

The following is a summary of certain provisions of the common stock, par value \$0.01 per share ("common stock"), of CDW Corporation (the "Company"), which is the only security of the Company registered pursuant to Section 12 of the Securities Exchange Act of 1934. This summary does not purport to be complete and is subject to the applicable provisions of the General Corporation Law of the State of Delaware (the "DGCL"), as well as the Company's Restated Certificate of Incorporation ("Certificate of Incorporation"), and the Company's Amended and Restated Bylaws ("Bylaws"), each of which are included as exhibits to the Company's Annual Report on Form 10-K and incorporated by reference herein.

Authorized Common Stock

The Company's authorized common stock consists of 1,000,000,000 shares.

Common Stock Voting Rights

Each holder of common stock is entitled to one vote per share on each matter submitted to a vote of stockholders. The Bylaws provide that the presence, in person or by proxy, of holders of shares representing a majority of the outstanding shares of capital stock entitled to vote at a stockholders' meeting shall constitute a quorum. When a quorum is present, the affirmative vote of a majority of the votes cast is required to take action, unless otherwise specified by law or the Certificate of Incorporation. There are no cumulative voting rights.

Common Stock Dividend Rights

Each holder of shares of common stock is entitled to receive such dividends and other distributions in cash, stock or property as may be declared by the Company's board of directors ("Board") from time to time out of the Company's assets or funds legally available for dividends or other distributions. These rights are subject to the preferential rights of any other class or series of the Company's preferred stock that the Company may designate and issue in the future.

The DGCL permits a corporation to declare and pay dividends out of "surplus" or, if there is no "surplus," out of its net profits for the fiscal year in which the dividend is declared and/or the preceding fiscal year. "Surplus" is defined as the excess of the net assets of the corporation over the amount determined to be the capital of the corporation by the board of directors. The capital of the corporation is typically calculated to be (and cannot be less than) the aggregate par value of all issued shares of capital stock. Net assets equals the fair value of the total assets minus total liabilities. The DGCL also provides that dividends may not be paid out of net profits if, after the payment of the dividend, remaining capital would be less than the capital represented by the outstanding stock of all classes having a preference upon the distribution of assets.

Declaration and payment of any dividend will be subject to the discretion of the Board. The time and amount of dividends will be dependent upon the Company's results of operations, financial condition, business prospects, capital requirements, contractual restrictions, any potential indebtedness the Company may incur, the provisions of Delaware law affecting the payment of distributions to stockholders, tax considerations and other factors that the Board deems relevant. In addition, the Company's ability to pay dividends on the common stock will be limited by restrictions on the Company's ability to pay dividends or make distributions to the Company's stockholders and on the ability of the Company's subsidiaries to pay dividends or make distributions to the Company, in each case, under the terms of the Company's current and any future agreements governing the Company's indebtedness.

Other Rights

Each holder of common stock is subject to, and may be adversely affected by, the rights of the holders of any series of preferred stock that the Company may designate and issue in the future. Holders of common stock will have no preemptive, conversion or other rights to subscribe for additional shares.

Liquidation Rights

In the event of any voluntary or involuntary liquidation, dissolution or winding up of the Company's affairs, holders of the common stock would be entitled to share ratably in the Company's assets that are legally available for distribution to stockholders after payment of the Company's debts and other liabilities. If the Company has any preferred stock outstanding at such time, holders of the preferred stock may be entitled to distribution and/or liquidation preferences. In either such case, the Company must pay the applicable distribution to the holders of the Company's preferred stock before the Company may pay distributions to the holders of common stock.

Preferred Stock

The Board has the authority to issue shares of preferred stock from time to time on terms it may determine, to divide shares of preferred stock into one or more series and to fix the designations, preferences, privileges and restrictions of preferred stock, including dividend rights, conversion rights, voting rights, terms of redemption, liquidation preference, sinking fund terms and the number of shares constituting any series or the designation of any series to the fullest extent permitted by the DGCL. The issuance of the Company's preferred stock could have the effect of decreasing the trading price of the common stock, restricting dividends on the Company's capital stock, diluting the voting power of the common stock, impairing the liquidation rights of the Company's capital stock, or delaying or preventing a change in control of the Company.

Anti-Takeover Effects of the Certificate of Incorporation and Bylaws

The Certificate of Incorporation and Bylaws contain provisions that may delay, defer or discourage another party from acquiring control of the Company. The Company expects that these provisions, which are summarized below, will discourage coercive takeover practices or inadequate takeover bids. These provisions are also designed to encourage persons seeking to acquire control of the Company to first negotiate with the Board, which the Company believes may result in an improvement of the terms of any such acquisition in favor of the Company's stockholders. However, they also give the Board the power to discourage acquisitions that some stockholders may favor.

Undesignated Preferred Stock

The ability to authorize undesignated preferred stock will make it possible for the Board to issue preferred stock with super voting, special approval, dividend or other rights or preferences on a discriminatory basis that could impede the success of any attempt to acquire the Company. These and other provisions may have the effect of deferring, delaying or discouraging hostile takeovers, or changes in control or management of the Company.

Board of Directors

The Certificate of Incorporation provides that all directors will be of one class, other than those directors who may be elected by the holders of any series of preferred stock under specified circumstances, with such directors serving a one-year term.

Stockholder Action by Written Consent

The Certificate of Incorporation provides that any action required or permitted to be taken by the Company's stockholders may be effected only at a duly called annual or special meeting of the stockholders and cannot be taken by written consent in lieu of a meeting.

Special Meeting of Stockholders and Advance Notice Requirements for Stockholder Proposals

The Certificate of Incorporation and Bylaws provide that, except as otherwise required by law, special meetings of the stockholders can only be called by or at the direction of the Board pursuant to a written resolution adopted by the affirmative vote of the majority of the total number of directors that the Company would have if there were no vacancies.

In addition, the Bylaws require advance notice procedures for stockholder proposals to be brought before an annual meeting of the stockholders, including the nomination of directors. Stockholders at an annual meeting may only consider the proposals specified in the notice of meeting or brought before the meeting by or at the direction of the Board, or by a stockholder of record on the record date for the meeting, who is entitled to vote at the meeting and who has delivered a timely written notice, in proper form to the Company's secretary, of the stockholder's intention to bring such business before the meeting.

These provisions could have the effect of delaying until the next stockholder meeting any stockholder actions, even if they are favored by the holders of a majority of the Company's outstanding voting securities.

Business Combinations with Interested Stockholders

The Company is subject to Section 203 of the DGCL, an anti-takeover law. In general, Section 203 prohibits a publicly held Delaware corporation from engaging in a business combination, such as a merger, with a person or group owning 15% or more of the corporation's voting stock for a period of three years following the date the person became an interested stockholder, unless (with certain exceptions) the business combination or the transaction in which the person became an interested stockholder is approved in a prescribed manner.

Exclusive Jurisdiction of Certain Actions

The Certificate of Incorporation requires, to the fullest extent permitted by law, that derivative actions brought in the Company's name, actions against directors, officers and employees for breach of fiduciary duty and other similar actions may be brought only in the Court of Chancery in the State of Delaware. Although the Company believes this provision benefits the Company by providing increased consistency in the application of Delaware law in the types of lawsuits to which it applies, the provision may have the effect of discouraging lawsuits against the Company's directors and officers. The enforceability of similar exclusive jurisdiction provisions in other companies' certificates of incorporation has been challenged in legal proceedings, and it is possible that, in connection with any action, a court could find the exclusive jurisdiction provision contained in the Certificate of Incorporation to be inapplicable or unenforceable in such action.

Transfer Agent and Registrar

The transfer agent and registrar for the common stock is Computershare Trust Company, N.A.

Listing

The common stock is listed on the Nasdaq Global Select Market under the trading symbol "CDW."

CDW Corporation 2021 Long-Term Incentive Plan

Stock Option Agreement

CDW Corporation, a Delaware corporation (the "<u>Company</u>"), hereby grants to the individual ("<u>Optionee</u>") named in the award notice attached hereto (the "<u>Award Notice</u>") as of the date set forth in the Award Notice (the "<u>Option Date</u>"), pursuant to the provisions of the CDW Corporation 2021 Long-Term Incentive Plan (the "<u>Plan</u>"), an option to purchase from the Company the number of shares of the Company's Common Stock, par value \$0.01 per share ("<u>Common Stock</u>"), set forth in the Award Notice at the price per share set forth in the Award Notice (the "<u>Exercise Price</u>") (the "<u>Option</u>"), upon and subject to the terms and conditions set forth below, in the Award Notice and in the Plan. Capitalized terms not defined herein shall have the meanings specified in the Plan.

- 1. Option Subject to Acceptance of Agreement. The Option shall be null and void unless Optionee shall accept this Agreement by executing the Award Notice in the space provided therefor and returning an original execution copy of the Award Notice to the Company (or electronically accepting this Agreement within the Optionee's stock plan account with the Company's stock plan administrator according to the procedures then in effect).
 - 2. Time and Manner of Exercise of Option.
- 2.1. <u>Maximum Term of Option</u>. In no event may the Option be exercised, in whole or in part, after the expiration date set forth in the Award Notice (the "Expiration Date").
- 2.2. <u>Vesting and Exercise of Option</u>. The Option shall become vested and exercisable in accordance with the vesting schedule set forth in the Award Notice (the "<u>Vesting Schedule</u>"). The period of time prior to the full vesting of the Option shall be referred to herein as the "<u>Vesting Period</u>." The Option shall be vested and exercisable following a termination of Optionee's employment according to the following terms and conditions:
- (a) Termination due to Death or Disability. If Optionee's employment with the Company terminates prior to the end of the Vesting Period by reason of Optionee's death or a termination by the Company due to Disability, then in either such case, the Option shall be 100% vested as of the date of termination, and the Option may thereafter be exercised by Optionee or Optionee's executor, administrator, legal representative, guardian or similar person until and including the earlier to occur of (i) the date which is one year after the date of termination of employment and (ii) the Expiration Date.
- (b) <u>Termination due to Retirement</u>. If Optionee's employment with the Company terminates prior to the end of the Vesting Period by reason of Optionee's Retirement, then the Option shall continue to vest in accordance with the Vesting Schedule, provided that Optionee complies with all Restrictive Covenants through the expiration of the Vesting Period, and the Option may thereafter be exercised by Optionee until and including the earlier to occur of (i) the date which is three years after the date of termination and (ii) the Expiration Date.
- (c) <u>Termination other than for Cause, Death, Disability or Retirement</u>. Subject to <u>Section 2.2(e)</u>, if Optionee's employment with the Company terminates prior to the end of the Vesting Period by reason of a termination of Optionee's employment (i) by the Company for any reason other than for Cause, death or Disability or (ii) by the Optionee for any reason other than Retirement, the Option, only to the extent vested on the effective date of such termination of employment, may thereafter be exercised by Optionee until and including the earlier to occur of (i) the date which is ninety (90) days after the date of such termination of employment and (ii) the Expiration Date.
- (d) <u>Termination for Cause</u>. If Optionee's employment with the Company terminates by reason of the Company's termination of Optionee's employment for Cause, then the Option, whether or not vested, shall terminate immediately upon such termination of employment.

(e) Change in Control.

- (i) In the event of a Change in Control prior to the end of the Vesting Period pursuant to which the Option is not effectively assumed or continued by the surviving or acquiring corporation in such Change in Control (as determined by the Board or Committee, with appropriate adjustments to the number and kind of shares, in each case, that preserve the intrinsic value and other material terms and conditions of the outstanding Option as in effect immediately prior to the Change in Control and in accordance with Section 409A of the Code), the Option shall be 100% vested immediately prior to such Change in Control and the Optionee shall receive in full settlement for such Option a cash payment in an amount equal to the aggregate number of shares of Common Stock then subject to the Option multiplied by the excess, if any, of the Fair Market Value of a share of Common Stock as of the date of the Change in Control, over the Exercise Price.
- (ii) In the event of a Change in Control prior to the end of the Vesting Period pursuant to which the Option is effectively assumed or continued by the surviving or acquiring corporation in such Change in Control (as determined by the Board or Committee, with appropriate adjustments to the number and kind of shares, in each case, that preserve the intrinsic value and other material terms and conditions of the outstanding Option as in effect immediately prior to the Change in Control and in accordance with Section 409A of the Code) and the Company terminates Optionee's employment without Cause or Optionee resigns for Good Reason within 24 months following such Change in Control and Optionee executes and does not revoke a waiver and release of claims in the form prescribed by the Company within 60 days after the date of such termination, the Option shall be 100% vested upon such termination of employment, and the Option may thereafter be exercised by Optionee until and including the earlier to occur of (i) the date which is one year after the date of termination of employment and (ii) the Expiration Date.
- (f) <u>Termination of Option During Blackout Period</u>. If the Option shall expire under <u>Section 2.2</u> during any period when the Optionee is prohibited from trading in securities of the Company pursuant to the Company's insider trading policy or other policy of the Company or during a period when the exercise of the Option would violate applicable securities laws (each, a "<u>Blackout Period</u>"), then the period during which the Option is exercisable shall be extended to the date that is 30 days after the expiration of such Blackout Period.

(g) <u>Definitions</u>.

- (after written notice and reasonable opportunity to cure) to perform duties properly assigned which are consistent with the scope and nature of Optionee's position; (B) Optionee's commission of an act materially and demonstrably detrimental to the financial condition and/or goodwill of the Company or any of its Subsidiaries, which act constitutes gross negligence or willful misconduct in the performance of duties to the Company or any of its Subsidiaries; (C) Optionee's commission of any theft, fraud, act of dishonesty or breach of trust resulting in or intended to result in material personal gain or enrichment of Optionee at the direct or indirect expense of the Company or any of its Subsidiaries; (D) Optionee's conviction of, or plea of guilty or nolo contendere to, a felony; (E) Optionee's material violation of any Restrictive Covenant; or (F) Optionee's material and willful violation of the Company's written policies or of Optionee's statutory or common law duty of loyalty to the Company or its affiliates that in either case is materially injurious to the Company, monetarily or otherwise. No act or failure to act will be considered "willful" (x) unless it is done, or omitted to be done, by Optionee in bad faith or without reasonable belief that Optionee's action or omission was in the best interests of the Company or (y) if it is done, or omitted to be done, in reliance on the informed advice of the Company's outside counsel or independent accountants or at the express direction of the Board.
- (ii) <u>Disability</u>. For purpose of this Option, "<u>Disability</u>" shall mean Optionee's absence from the Optionee's duties with the Company on a full-time basis for at least

180 consecutive days as a result of the Optionee's incapacity due to physical or mental illness, or under such other circumstances as the Committee determines, in its sole discretion, constitute a Disability.

- (iii) Good Reason. For purposes of this Option, "Good Reason" shall mean that the Optionee resigns from employment with the Company and its Subsidiaries as a result of one or more of the following reasons: (A) the Company reduces the amount of the Optionee's base salary or cash bonus opportunity (it being understood that the Board shall have discretion to set the Company's and the Optionee's personal performance targets to which the cash bonus will be tied), (B) the Company adversely changes the Optionee's reporting responsibilities, titles or office as in effect as of the date hereof or reduces his/her position, authority, duties, responsibilities or status materially inconsistent with the positions, authority, duties, responsibilities or status the Optionee then holds, (C) any successor to the Company in any merger, consolidation or transfer of assets does not expressly assume any material obligation of the Company to the Optionee under any agreement or plan pursuant to which the Optionee receives benefits or rights, or (D) the Company changes the Optionee's place of work to a location more than fifty (50) miles from the Optionee's present place of work; provided, however, that the occurrence of any such condition shall not constitute Good Reason unless (1) Optionee provides written notice to the Company of the existence of such condition not later than 60 days after Optionee knows or reasonably should know of the existence of such condition, (2) the Company fails to remedy such condition within 30 days after receipt of such notice and (3) Optionee resigns due to the existence of such condition within 60 days after the expiration of the remedial period described in clause (2) hereof.
- (iv) <u>Restrictive Covenant</u>. For purposes of this Option, "<u>Restrictive Covenant</u>" shall mean any non-competition, non-solicitation, confidentiality or protection of trade secrets (or similar provision regarding intellectual property) covenant by which Optionee is bound under any agreement between Optionee and the Company and its Subsidiaries.
- (v) <u>Retirement</u>. For purposes of this Option, "<u>Retirement</u>" shall mean Optionee's termination of employment at a time when (A) the Optionee has attained age 55 and (B) the sum of the Optionee's age and years of employment with or service to the Company or its Subsidiaries equals or exceeds 65; provided that such termination occurs at least six months after the Option Date.
- 2.3. Method of Exercise. Subject to the limitations set forth in this Agreement, the Option, to the extent vested, may be exercised by Optionee (a) by delivering to the Company an exercise notice in the form prescribed by the Company specifying the number of whole shares of Common Stock to be purchased and by accompanying such notice with payment therefor in full (or by arranging for such payment to the Company's satisfaction) either (i) in cash, (ii) to the extent permitted by the Committee, by delivery to the Company (either actual delivery or by attestation procedures established by the Company) of shares of Common Stock having an aggregate Fair Market Value, determined as of the date of exercise, equal to the aggregate purchase price payable pursuant to the Option by reason of such exercise, (iii) to the extent permitted by the Committee, by authorizing the Company to withhold whole shares of Common Stock which would otherwise be delivered having an aggregate Fair Market Value, determined as of the date of exercise, equal to the amount necessary to satisfy such obligation, (iv) except as may be prohibited by applicable law, in cash by a broker-dealer acceptable to the Company to whom Optionee has submitted an irrevocable notice of exercise or (v) by a combination of (i), (ii) and (iii), and (b) by executing such documents as the Company may reasonably request. No share of Common Stock or certificate representing a share of Common Stock shall be issued or delivered until the full purchase price therefor and any withholding taxes thereon, as described in Section 5.1, have been paid.
- 2.4. <u>Termination of Option</u>. In no event may the Option be exercised after it terminates as set forth in this <u>Section 2.4</u>. The Option shall terminate, to the extent not earlier terminated pursuant to <u>Section 2.2</u> or exercised pursuant to <u>Section 2.3</u>, on the Expiration Date. Upon the termination of the Option, the Option and all rights hereunder shall immediately become null and void.

3. Clawback of Proceeds.

- 3.1. <u>Clawback of Proceeds</u>. This award is subject to the clawback provisions in Section 5.15 of the Plan. In addition, if Optionee materially violates any Restrictive Covenant and such violation occurs on or before the third anniversary of the date of Optionee's termination of employment: (i) the Option shall be forfeited and (ii) any and all Option Proceeds (as hereinafter defined) shall be immediately due and payable by the Optionee to the Company. For purposes of this Section, "<u>Option Proceeds</u>" shall mean, with respect to any portion of the Option which is exercised later than 24 months prior to the date of the Optionee's termination of employment or service with the Company (x) the difference between (A) the Fair Market Value of a share of Common Stock on the date such portion of the Option was exercised and (B) the per share exercise price of the Option, multiplied by (y) the number of shares of Common Stock purchased pursuant to the exercise of such portion of the Option. The remedy provided by this Section shall be in addition to and not in lieu of any rights or remedies which the Company may have against the Optionee in respect of a breach by the Optionee of any duty or obligation to the Company.
- 3.2. <u>Right of Setoff</u>. The Optionee agrees that by accepting the Award Notice the Optionee authorizes the Company and its affiliates to deduct any amount or amounts owed by the Optionee pursuant to this <u>Section 3</u> from any amounts payable by or on behalf of the Company or any affiliate to the Optionee, including, without limitation, any amount payable to the Optionee as salary, wages, vacation pay, bonus or the settlement of the Option or any stock-based award. This right of setoff shall not be an exclusive remedy and the Company's or an affiliate's election not to exercise this right of setoff with respect to any amount payable to the Optionee shall not constitute a waiver of this right of setoff with respect to any other amount payable to the Optionee or any other remedy.

4. <u>Transfer Restrictions and Investment Representations.</u>

- 4.1. <u>Nontransferability of Option</u>. The Option may not be transferred by Optionee other than by will or the laws of descent and distribution or pursuant to the designation of one or more beneficiaries on the form prescribed by the Company. Except to the extent permitted by the foregoing sentence, (i) during Optionee's lifetime the Option is exercisable only by Optionee or Optionee's legal representative, guardian or similar person and (ii) the Option may not be sold, transferred, assigned, pledged, hypothecated, encumbered or otherwise disposed of (whether by operation of law or otherwise) or be subject to execution, attachment or similar process. Upon any attempt to so sell, transfer, assign, pledge, hypothecate, encumber or otherwise dispose of the Option, the Option and all rights hereunder shall immediately become null and void.
- 4.2. <u>Investment Representation</u>. Optionee hereby represents and covenants that (a) any shares of Common Stock purchased upon exercise of the Option will be purchased for investment and not with a view to the distribution thereof within the meaning of the Securities Act of 1933, as amended (the "<u>Securities Act</u>"), unless such purchase has been registered under the Securities Act and any applicable state securities laws; (b) any subsequent sale of any such shares shall be made either pursuant to an effective registration statement under the Securities Act and any applicable state securities laws, or pursuant to an exemption from registration under the Securities Act and such state securities laws; and (c) if requested by the Company, Optionee shall submit a written statement, in a form satisfactory to the Company, to the effect that such representation (x) is true and correct as of the date of any purchase of any shares hereunder or (y) is true and correct as of the date of any sale of any such shares, as applicable. As a further condition precedent to any exercise of the Option, Optionee shall comply with all regulations and requirements of any regulatory authority having control of or supervision over the issuance or delivery of the shares and, in connection therewith, shall execute any documents which the Board or the Committee shall in its sole discretion deem necessary or advisable.

5. Additional Terms and Conditions.

5.1. Withholding Taxes. (a) As a condition precedent to the issuance of Common Stock following the exercise of the Option, Optionee shall, upon request by the Company, pay to the Company in addition to the purchase price of the shares, such amount as the Company determines is required, under all applicable federal, state, local or other laws or regulations, to be withheld and paid

over as income or other withholding taxes (the "Required Tax Payments") with respect to such exercise of the Option. If Optionee shall fail to advance the Required Tax Payments after request by the Company, the Company may, in its discretion, deduct any Required Tax Payments from any amount then or thereafter payable by the Company to Optionee.

- (b) Optionee may elect to satisfy his or her obligation to advance the Required Tax Payments by any of the following means: (i) a cash payment to the Company; (ii) to the extent permitted by the Committee, delivery to the Company (either actual delivery or by attestation procedures established by the Company) of previously owned whole shares of Common Stock having an aggregate Fair Market Value, determined as of the date on which such withholding obligation arises (the "Tax Date"), equal to the Required Tax Payments; (iii) to the extent permitted by the Committee, authorizing the Company to withhold whole shares of Common Stock which would otherwise be delivered to Optionee upon exercise of the Option having an aggregate Fair Market Value, determined as of the Tax Date, equal to the Required Tax Payments; (iv) except as may be prohibited by applicable law, a cash payment by a broker-dealer acceptable to the Company to whom Optionee has submitted an irrevocable notice of exercise or (v) any combination of (i), (ii) and (iii). Shares of Common Stock to be delivered or withheld may not have a Fair Market Value in excess of the amount determined by applying the maximum individual statutory tax rate in the Optionee's jurisdiction; provided that the Committee shall be permitted to limit the number of shares so delivered or withheld to a lesser number if necessary, as determined by the Committee, to avoid adverse accounting consequences or for administrative convenience; provided, however, that if a fraction of a share of Common Stock would be required to satisfy the maximum individual statutory rate in the Optionee's jurisdiction, then the number of shares of Common Stock to be delivered or withheld may be rounded up to the next nearest whole share of Common Stock. No share of Common Stock or certificate representing a share of Common Stock shall be issued or delivered until the Required Tax Payments have been satisfied in full.
- 5.2. Adjustment. In the event of any equity restructuring (within the meaning of Financial Accounting Standards Board Accounting Standards Codification Topic 718, Compensation—Stock Compensation) that causes the per share value of shares of Common Stock to change, such as a stock dividend, stock split, spinoff, rights offering or recapitalization through an extraordinary dividend, the number and class of securities subject to the Option and the Exercise Price shall be equitably adjusted by the Committee, such adjustment to be made in accordance with Section 409A of the Code. In the event of any other change in corporate capitalization, including a merger, consolidation, reorganization, or partial or complete liquidation of the Company, such equitable adjustments described in the foregoing sentence may be made as determined to be appropriate and equitable by the Committee (or, if the Company is not the surviving corporation in any such transaction, the board of directors of the surviving corporation) to prevent dilution or enlargement of rights of participants. The decision of the Committee regarding any such adjustment shall be final, binding and conclusive.
- 5.3. Compliance with Applicable Law. The Option is subject to the condition that if the listing, registration or qualification of the shares subject to the Option upon any securities exchange or under any law, or the consent or approval of any governmental body, or the taking of any other action is necessary or desirable as a condition of, or in connection with, the purchase or issuance of shares hereunder, the Option may not be exercised, in whole or in part, and such shares may not be issued, unless such listing, registration, qualification, consent, approval or other action shall have been effected or obtained, free of any conditions not acceptable to the Company. The Company agrees to use reasonable efforts to effect or obtain any such listing, registration, qualification, consent, approval or other action.
- 5.4. <u>Issuance or Delivery of Shares</u>. Upon the exercise of the Option, in whole or in part, the Company shall issue or deliver, subject to the conditions of this Agreement, the number of shares of Common Stock purchased against full payment therefor. Such issuance shall be evidenced by the appropriate entry on the books of the Company or of a duly authorized transfer agent of the Company. The Company shall pay all original issue or transfer taxes and all fees and expenses incident to such issuance, except as otherwise provided in Section 5.1.
- 5.5. Option Confers No Rights as Stockholder. Optionee shall not be entitled to any privileges of ownership with respect to shares of Common Stock subject to the Option unless and until such shares are purchased and issued upon the exercise of the Option, in whole or in part, and Optionee

becomes a stockholder of record with respect to such issued shares. Optionee shall not be considered a stockholder of the Company with respect to any such shares not so purchased and issued.

- 5.6. Option Confers No Rights to Continued Employment. In no event shall the granting of the Option or its acceptance by Optionee, or any provision of this Agreement or the Plan, give or be deemed to give Optionee any right to continued employment by the Company, any Subsidiary or any affiliate of the Company or affect in any manner the right of the Company, any Subsidiary or any affiliate of the Company to terminate the employment of any person at any time.
- 5.7. <u>Decisions of Board or Committee</u>. The Board or the Committee shall have the right to resolve all questions which may arise in connection with the Option or its exercise. Any interpretation, determination or other action made or taken by the Board or the Committee regarding the Plan or this Agreement shall be final, binding and conclusive.
- 5.8. <u>Successors</u>. This Agreement shall be binding upon and inure to the benefit of any successor or successors of the Company and any person or persons who shall, upon the death of Optionee, acquire any rights hereunder in accordance with this Agreement or the Plan.
- 5.9. Notices. All notices, requests or other communications provided for in this Agreement shall be made, if to the Company, to CDW Corporation, Attn: General Counsel, 200 N. Milwaukee Avenue, Vernon Hills, Illinois 60061, and if to Optionee, to the last known mailing address of Optionee contained in the records of the Company. All notices, requests or other communications provided for in this Agreement shall be made in writing either (a) by personal delivery, (b) by facsimile or electronic mail with confirmation of receipt, (c) by mailing in the United States mails or (d) by express courier service. The notice, request or other communication shall be deemed to be received upon personal delivery, upon confirmation of receipt of facsimile or electronic mail transmission or upon receipt by the party entitled thereto if by United States mail or express courier service; provided, however, that if a notice, request or other communication sent to the Company is not received during regular business hours, it shall be deemed to be received on the next succeeding business day of the Company.
- 5.10. Governing Law. This Agreement, the Option and all determinations made and actions taken pursuant hereto and thereto, to the extent not governed by the Code or the laws of the United States, shall be governed by the laws of the State of Delaware and construed in accordance therewith without giving effect to principles of conflicts of laws.
- 5.11. Agreement Subject to the Plan. This Agreement is subject to the provisions of the Plan and shall be interpreted in accordance therewith. In the event that the provisions of this Agreement and the Plan conflict, the Plan shall control. The Optionee hereby acknowledges receipt of a copy of the Plan.
- 5.12. <u>Entire Agreement</u>. This Agreement and the Plan constitute the entire agreement of the parties with respect to the subject matter hereof and supersede in their entirety all prior undertakings and agreements of the Company and the Optionee with respect to the subject matter hereof, and may not be modified adversely to the Optionee's interest except by means of a writing signed by the Company and the Optionee.
- 5.13. <u>Partial Invalidity</u>. The invalidity or unenforceability of any particular provision of this Agreement shall not effect the other provisions hereof and this Agreement shall be construed in all respects as if such invalid or unenforceable provisions were omitted.
- 5.14. <u>Amendment and Waiver</u>. The provisions of this Agreement may be amended or waived only by the written agreement of the Company and the Optionee, and no course of conduct or failure or delay in enforcing the provisions of this Agreement shall affect the validity, binding effect or enforceability of this Agreement.
- 5.15. <u>Counterparts</u>. The Award Notice may be executed in two counterparts, each of which shall be deemed an original and both of which together shall constitute one and the same instrument.

CDW Corporation 2021 Long-Term Incentive Plan

Performance Share Unit Award Agreement

CDW Corporation, a Delaware corporation (the "Company"), hereby grants to the individual (the "Holder") named in the award notice attached hereto (the "Award Notice") as of the date set forth in the Award Notice (the "Grant Date"), pursuant to the provisions of the CDW Corporation 2021 Long-Term Incentive Plan (the "Plan"), a performance share unit award (the "Award") with respect to the number of shares of the Company's Common Stock, par value \$0.01 per share ("Stock"), set forth in the Award Notice, upon and subject to the restrictions, terms and conditions set forth in the Plan and this agreement (the "Agreement"). Capitalized terms not defined herein shall have the meanings specified in the Plan.

- 1. <u>Award Subject to Acceptance of Agreement</u>. The Award shall be null and void unless the Holder accepts this Agreement by executing the Award Notice in the space provided therefor and returning an original execution copy of the Award Notice to the Company (or electronically accepting this Agreement within the Holder's stock plan account with the Company's stock plan administrator according to the procedures then in effect).
- Rights as a Stockholder. The Holder shall not be entitled to any privileges of ownership with respect to the shares of Stock subject to the Award unless and until, and only to the extent, such shares become vested pursuant to Section 3 hereof and the Holder becomes a stockholder of record with respect to such shares. As of each date on which the Company pays a cash dividend to record owners of shares of Stock (a "Dividend Date"), the number of shares subject to the Award shall increase by (i) the product of the total number of shares subject to the Award immediately prior to such Dividend Date multiplied by the dollar amount of the cash dividend paid per share of Stock by the Company on such Dividend Date, divided by (ii) the Fair Market Value of a share of Stock on such Dividend Date. Any such additional shares shall be subject to the same vesting conditions and payment terms set forth herein as the shares to which they relate.
 - 3. <u>Restriction Period, Vesting and Settlement.</u>
- 3.1. <u>Performance-Based Vesting Conditions</u>. Subject to the remainder of this <u>Section 3</u>, the shares of Stock subject to the Award shall vest pursuant to the terms of this Agreement and the Plan based on the achievement of the performance goals set forth in the Award Notice over the performance period set forth in the Award Notice (the "<u>Performance Period</u>"), provided that the Holder remains in continuous employment with the Company through the end of the Performance Period, and the vested portion of the Award shall be paid to the Holder within 70 days after the end of the Performance Period. Attainment of the performance goals shall be determined and certified by the Committee in writing prior to the settlement of the Award.

3.2. <u>Termination of Employment</u>

- (a) Termination due to Retirement. If the Holder's employment with the Company terminates prior to the end of the Performance Period and prior to a Change in Control by reason of the Holder's Retirement, the Performance Period shall continue through the last day thereof and the Holder shall be entitled to a prorated Award, provided that the Holder has continuously complied with the Restrictive Covenants. Such prorated Award shall be equal to the number of shares earned at the end of the Performance Period based on the actual performance during the Performance Period multiplied by a fraction, the numerator of which shall equal the number of full months in the Performance Period during which the Holder was employed by the Company and the denominator of which shall equal 36. Such vested Award shall be paid to the Holder within 70 days after the end of the Performance Period.
- (b) <u>Death or Disability</u>. If the Holder dies or terminates employment due to Disability on or before the end of the Performance Period, and in either case prior to a Change in Control,

then the Award shall become fully vested. If such death or termination occurs prior to the 24-month anniversary of the first day of the Performance Period, then the performance goals set forth in Section 3.1 shall be deemed to have been satisfied at the target level. If such death or termination occurs on or after the 24-month anniversary of the first day of the Performance Period, then the number of shares of Stock earned pursuant to Section 3.1 shall be based on the projected level of performance through the end of the Performance Period, as determined by the Company for purposes of its financial statements for the fiscal quarter ending prior to the date of such death or termination. The vested Award shall be settled within 70 days following the date of the Holder's death or termination of employment.

(c) <u>Termination other than due to Retirement, Death or Disability</u>. If the Holder's employment with the Company terminates prior to the end of the Performance Period and prior to a Change in Control by reason of (i) the Company's termination of the Holder's employment for any reason other than death or Disability or (ii) the Holder's resignation for any reason other than Retirement, then the Award shall be immediately forfeited by the Holder and cancelled by the Company.

3.3. <u>Change in Control</u>.

- (a) <u>Satisfaction of Performance Goals</u>. If a Change in Control occurs prior to the 24-month anniversary of the first day of the Performance Period, the performance goals set forth in <u>Section 3.1</u> shall be deemed to have been satisfied at the target level. If the Change in Control occurs on or after the 24-month anniversary of the first day of the Performance Period, the number of shares of Stock earned pursuant to <u>Section 3.1</u> shall be based on the projected level of performance through the end of the Performance Period, as determined by the Committee prior to the date of the Change in Control based on performance through the date of such determination. If the Change in Control occurs after the date on which the Participant's employment is terminated by reason of Retirement, pursuant to <u>Section 3.2(a)</u>, the number of shares earned for purposes of such section shall be determined as of the date of the Change in Control in accordance with this <u>Section 3.3(a)</u> and such number of shares, prorated in accordance with Section 3.2(a), shall be settled within 70 days following such Change in Control.
- (b) Vesting and Settlement of Award Not Assumed. In the event of a Change in Control prior to the end of the Performance Period pursuant to which the Award is not effectively assumed or continued by the surviving or acquiring corporation in such Change in Control (as determined by the Board or Committee, with appropriate adjustments to the number and kind of shares, in each case, that preserve the value of the shares subject to the Award and other material terms and conditions of the outstanding Award as in effect immediately prior to the Change in Control), the Award shall vest as of the date of the Change in Control, based on the performance level determined in accordance with Section 3.3(a). If the Change in Control constitutes a "change in control event," within the meaning of Section 409A of the Code, and the Company terminates all deferred compensation plans of the same type to the extent required under Section 409A of the Code, then the Award shall be settled in cash within 70 days following such Change in Control.
- (c) Vesting and Settlement of Award Assumed. In the event of a Change in Control prior to the end of the Performance Period pursuant to which the Award is effectively assumed or continued by the surviving or acquiring corporation in such Change in Control (as determined by the Board or Committee, with appropriate adjustments to the number and kind of shares, in each case, that preserve the value of the shares subject to the Award and other material terms and conditions of the outstanding Award as in effect immediately prior to the Change in Control) and (i) the Holder remains continuously employed through the end of the Performance Period, (ii) the Company terminates the Holder's employment without Cause or the Holder resigns for Good Reason within 24 months following such Change in Control and the Holder executes and does not revoke a waiver and release of claims in the form prescribed by the Company within 60 days after the date of such termination (iii) the Holder's employment terminates due to Retirement following such Change in Control, or (iv) the Holder dies or terminates employment due to Disability following such Change in Control, in any such case, the Award shall vest based on the performance level determined in accordance with Section 3.3(a) hereof. In the case of a termination pursuant to clause (ii) of this Section 3.3(c) (Retirement), the Award shall be paid in full, and in the case of a termination pursuant to clause (iii) of this Section 3.3(c) (Retirement), the Award shall be prorated in accordance with, and subject to the terms of, Section 3.2(a).

The vested Award shall be settled within 70 days following the end of the Performance Period or, if earlier, the Holder's termination of employment or death. If, following a Change in Control, the Holder experiences a termination of employment other than as set forth in this Section 3.3(c), the Award shall be immediately forfeited by the Holder and cancelled by the Company.

3.4. Definitions.

- (a) <u>Cause</u>. For purposes of this Award, "<u>Cause</u>" shall mean one or more of the following: (A) Holder's refusal (after written notice and reasonable opportunity to cure) to perform duties properly assigned which are consistent with the scope and nature of Holder's position; (B) Holder's commission of an act materially and demonstrably detrimental to the financial condition and/or goodwill of the Company or any of its Subsidiaries, which act constitutes gross negligence or willful misconduct in the performance of duties to the Company or any of its Subsidiaries; (C) Holder's commission of any theft, fraud, act of dishonesty or breach of trust resulting in or intended to result in material personal gain or enrichment of Holder at the direct or indirect expense of the Company or any of its Subsidiaries; (D) Holder's conviction of, or plea of guilty or nolo contendere to, a felony; (E) Holder's material violation of any Restrictive Covenant; or (F) Holder's material and willful violation of the Company's written policies or of Holder's statutory or common law duty of loyalty to the Company or its affiliates that in either case is materially injurious to the Company, monetarily or otherwise. No act or failure to act will be considered "willful" (x) unless it is done, or omitted to be done, by Holder in bad faith or without reasonable belief that Holder's action or omission was in the best interests of the Company or (y) if it is done, or omitted to be done, in reliance on the informed advice of the Company's outside counsel or independent accountants or at the express direction of the Board.
- (b) <u>Disability</u>. For purposes of this Award, "<u>Disability</u>" shall mean the Holder's absence from the Holder's duties with the Company on a full-time basis for at least 180 consecutive days as a result of the Holder's incapacity due to physical or mental illness, or under such other circumstances as the Committee determines, in its sole discretion, constitute a Disability.
- (c) Good Reason. For purposes of this Award, "Good Reason" shall mean that the Holder resigns from employment with the Company and its Subsidiaries as a result of one or more of the following reasons: (i) the Company reduces the amount of the Holder's base salary or cash bonus opportunity (it being understood that the Board shall have discretion to set the Company's and the Holder's personal performance targets to which the cash bonus will be tied), (ii) the Company adversely changes the Holder's reporting responsibilities, titles or office as in effect as of the date hereof or reduces his/her position, authority, duties, responsibilities or status materially inconsistent with the positions, authority, duties, responsibilities or status the Holder then holds, (iii) any successor to the Company in any merger, consolidation or transfer of assets does not expressly assume any material obligation of the Company to the Holder under any agreement or plan pursuant to which the Holder receives benefits or rights, or (iv) the Company changes the Holder's place of work to a location more than fifty (50) miles from the Holder's present place of work; provided, however, that the occurrence of any such condition shall not constitute Good Reason unless (A) the Holder provides written notice to the Company of the existence of such condition not later than 60 days after the Holder knows or reasonably should know of the existence of such condition within 60 days after the expiration of the remedial period described in clause (B) hereof.
- (d) <u>Restrictive Covenant</u>. For purposes of this Award, "<u>Restrictive Covenant</u>" shall mean any non-competition, non-solicitation, confidentiality or protection of trade secrets (or similar provision regarding intellectual property) covenant by which Holder is bound under any agreement between Holder and the Company and its Subsidiaries.
- (e) <u>Retirement</u>. For purposes of this Award, "<u>Retirement</u>" shall mean Holder's termination of employment at a time when (i) the Holder has attained age 55 and (B) the sum of the Holder's age and years of employment with or service to the Company or its Subsidiaries equals or exceeds 65; provided that such termination occurs at least six months after the Grant Date.

4. <u>Issuance or Delivery of Shares.</u> Subject to <u>Section 7.12</u>, the Company shall issue or deliver, subject to the conditions of this Agreement, the vested shares of Stock to the Holder at the time specified in Section 3 hereof; <u>provided, however,</u> that in the event of vesting of the Award pursuant to <u>Section 3.3</u>, if the Award constitutes nonqualified deferred compensation (within the meaning of Section 409A of the Code) and such Change in Control is not a "change in control event" (within the meaning of Section 409A of the Code, then such Award shall be paid within 70 days after the earliest to occur of (i) the last day of the Performance Period, (ii) the Holder's termination of employment and (iii) the Holder's death; provided that if the Holder's right to such payment is subject to a substantial risk of forfeiture as of such payment date, then such payment shall be delayed and paid within 70 days after such substantial risk of forfeiture lapses. Such issuance or delivery shall be evidenced by the appropriate entry on the books of the Company or of a duly authorized transfer agent of the Company. The Company shall pay all original issue or transfer taxes and all fees and expenses incident to such issuance or delivery, except as otherwise provided in <u>Section 7</u>. Prior to the issuance to the Holder of the shares of Stock subject to the Award, the Holder shall have no direct or secured claim in any specific assets of the Company or in such shares of Stock, and will have the status of a general unsecured creditor of the Company.

5. Clawback of Proceeds.

- 5.1. <u>Clawback of Proceeds</u>. This award is subject to the clawback provisions in Section 5.15 of the Plan. In addition, if the Holder materially violates any Restrictive Covenant and such violation occurs on or before the third anniversary of the date of the Holder's termination of employment: (i) the Award shall be forfeited and (ii) any and all Performance Share Proceeds (as hereinafter defined) shall be immediately due and payable by the Holder to the Company. For purposes of this Section, "<u>Performance Share Proceeds</u>" shall mean, with respect to any portion of the Award which is settled later than 24 months prior to the date of the Holder's termination of employment or service with the Company the Fair Market Value of a share of Stock on the date such portion of the Award was settled, multiplied by the number of shares of Stock issued to the Holder pursuant to the settlement of such portion of the Award. The remedy provided by this Section shall be in addition to and not in lieu of any rights or remedies which the Company may have against the Holder in respect of a breach by the Holder of any duty or obligation to the Company.
- 5.2. <u>Right of Setoff</u>. The Holder agrees that by accepting the Award the Holder authorizes the Company and its affiliates to deduct any amount or amounts owed by the Holder pursuant to this <u>Section 5</u> from any amounts payable by or on behalf of the Company or any affiliate to the Holder, including, without limitation, any amount payable to the Holder as salary, wages, vacation pay, bonus or the vesting or settlement of the Award or any stock-based award. This right of setoff shall not be an exclusive remedy and the Company's or an affiliate's election not to exercise this right of setoff with respect to any amount payable to the Holder shall not constitute a waiver of this right of setoff with respect to any other amount payable to the Holder or any other remedy.

6. <u>Transfer Restrictions and Investment Representation.</u>

- 6.1. <u>Nontransferability of Award</u>. The Award may not be transferred by the Holder other than by will or the laws of descent and distribution. Except to the extent permitted by the foregoing sentence, the Award may not be sold, transferred, assigned, pledged, hypothecated, encumbered or otherwise disposed of (whether by operation of law or otherwise) or be subject to execution, attachment or similar process. Upon any attempt to so sell, transfer, assign, pledge, hypothecate, encumber or otherwise dispose of the Award, the Award and all rights hereunder shall immediately become null and void.
- 6.2. <u>Investment Representation</u>. The Holder hereby covenants that (a) any sale of any share of Stock acquired upon the vesting of the Award shall be made either pursuant to an effective registration statement under the Securities Act of 1933, as amended (the "<u>Securities Act</u>"), and any applicable state securities laws, or pursuant to an exemption from registration under the Securities Act and such state securities laws and (b) the Holder shall comply with all regulations and requirements of any regulatory authority having control of or supervision over the issuance of the shares and, in

connection therewith, shall execute any documents which the Committee shall in its sole discretion deem necessary or advisable.

7. <u>Additional Terms and Conditions of Award</u>.

- 7.1. Withholding Taxes. As a condition precedent to the issuance or delivery of the Stock upon the vesting of the Award, at the Company's discretion either (i) the Holder shall pay to the Company such amount as the Company (or an affiliate) determines is required, under all applicable federal, state, local, foreign or other laws or regulations, to be withheld and paid over as income or other withholding taxes (the "Required Tax Payments") with respect to the Award or (ii) the Company or an affiliate may, in its discretion, deduct any Required Tax Payments from any amount then or thereafter payable by the Company or an affiliate to the Holder, which may include the withholding of whole shares of Stock which would otherwise be delivered to the Holder having an aggregate Fair Market Value, determined as of the date on which such withholding obligation arises, equal to the Required Tax Payments, in either case in accordance with such terms, conditions and procedures that may be prescribed by the Company. Shares of Stock withheld may not have a Fair Market Value in excess of the amount determined by applying the maximum individual statutory tax rate in the Holder's jurisdiction; provided that the Company shall be permitted to limit the number of shares so withheld to a lesser number if necessary, as determined by the Company, to avoid adverse accounting consequences or for administrative convenience; provided, however, that if a fraction of a share of Stock would be required to satisfy the maximum individual statutory rate in the Holder's jurisdiction, then the number of shares of Stock to be withheld may be rounded up to the next nearest whole share of Stock. No certificate representing a share of Stock shall be delivered until the Required Tax Payments have been satisfied in full. Any determination by the Company with respect to the withholding of shares of Stock to satisfy the Required Tax Payments shall be made by the Committee if the Holder is subject to Section 16 of the Exchange Act.
- 7.2. Compliance with Applicable Law. The Award is subject to the condition that if the listing, registration or qualification of the shares of Stock subject to the Award upon any securities exchange or under any law, or the consent or approval of any governmental body, or the taking of any other action is necessary or desirable as a condition of, or in connection with, the delivery of shares hereunder, the shares of Stock subject to the Award shall not be delivered, in whole or in part, unless such listing, registration, qualification, consent, approval or other action shall have been effected or obtained, free of any conditions not acceptable to the Company agrees to use reasonable efforts to effect or obtain any such listing, registration, qualification, consent, approval or other action.
- 7.3. Award Confers No Rights to Continued Employment. In no event shall the granting of the Award or its acceptance by the Holder, or any provision of the Agreement or the Plan, give or be deemed to give the Holder any right to continued employment by the Company, any Subsidiary or any affiliate of the Company or affect in any manner the right of the Company, any Subsidiary or any affiliate of the Company to terminate the employment of any person at any time.
- 7.4. <u>Decisions of Board or Committee</u>. The Board or the Committee shall have the right to resolve all questions which may arise in connection with the Award. Any interpretation, determination or other action made or taken by the Board or the Committee regarding the Plan or this Agreement shall be final, binding and conclusive.
- 7.5. <u>Successors</u>. This Agreement shall be binding upon and inure to the benefit of any successor or successors of the Company and any person or persons who shall, upon the death of the Holder, acquire any rights hereunder in accordance with this Agreement or the Plan.
- 7.6. Notices. All notices, requests or other communications provided for in this Agreement shall be made, if to the Company, to CDW Corporation, Attn: General Counsel, 200 N. Milwaukee Avenue, Vernon Hills, Illinois 60061, and if to the Holder, to the last known mailing address of the Holder contained in the records of the Company. All notices, requests or other communications provided for in this Agreement shall be made in writing either (a) by personal delivery, (b) by facsimile or electronic mail with confirmation of receipt, (c) by mailing in the United States mails or (d) by express

courier service. The notice, request or other communication shall be deemed to be received upon personal delivery, upon confirmation of receipt of facsimile or electronic mail transmission or upon receipt by the party entitled thereto if by United States mail or express courier service; <u>provided</u>, <u>however</u>, that if a notice, request or other communication sent to the Company is not received during regular business hours, it shall be deemed to be received on the next succeeding business day of the Company.

- 7.7. <u>Governing Law</u>. This Agreement, the Award and all determinations made and actions taken pursuant hereto and thereto, to the extent not governed by the laws of the United States, shall be governed by the laws of the State of Delaware and construed in accordance therewith without giving effect to principles of conflicts of laws.
- 7.8. <u>Agreement Subject to the Plan</u>. This Agreement is subject to the provisions of the Plan and shall be interpreted in accordance therewith. In the event that the provisions of this Agreement and the Plan conflict, the Plan shall control. The Holder hereby acknowledges receipt of a copy of the Plan.
- 7.9. Entire Agreement. This Agreement and the Plan constitute the entire agreement of the parties with respect to the subject matter hereof and supersede in their entirety all prior undertakings and agreements of the Company and the Holder with respect to the subject matter hereof, and may not be modified adversely to the Holder's interest except by means of a writing signed by the Company and the Holder.
- 7.10. <u>Partial Invalidity</u>. The invalidity or unenforceability of any particular provision of this Agreement shall not affect the other provisions hereof and this Agreement shall be construed in all respects as if such invalid or unenforceable provision was omitted.
- 7.11. Amendment and Waiver. The Company may amend the provisions of this Agreement at any time; <u>provided</u> that an amendment that would adversely affect the Holder's rights under this Agreement shall be subject to the written consent of the Holder. No course of conduct or failure or delay in enforcing the provisions of this Agreement shall affect the validity, binding effect or enforceability of this Agreement.
- 7.12. Compliance With Section 409A of the Code. This Award is intended to be exempt from or comply with Section 409A of the Code, and shall be interpreted and construed accordingly. To the extent this Agreement provides for the Award to become vested and be settled upon the Holder's termination of employment, the applicable shares of Stock shall be transferred to the Holder or his or her beneficiary upon the Holder's "separation from service," within the meaning of Section 409A of the Code; provided that if the Holder is a "specified employee," within the meaning of Section 409A of the Code, then to the extent the Award constitutes nonqualified deferred compensation, within the meaning of Section 409A of the Code, such shares of Stock shall be transferred to the Holder or his or her beneficiary upon the earlier to occur of (i) the six-month anniversary of such separation from service and (ii) the date of the Holder's death. For purposes of complying with Section 409A of the Code, the Award shall be treated as payable upon the later to occur of (A) the date on which the Award ceases to be subject to a substantial risk of forfeiture and (B) the earliest of (x) the last day of the Performance Period, (y) the Holder's separation from service and (z) the Holder's death.

CDW Corporation 2021 Long-Term Incentive Plan

Restricted Stock Unit Award Agreement

CDW Corporation, a Delaware corporation (the "<u>Company</u>"), hereby grants to the individual (the "<u>Holder</u>") named in the award notice attached hereto (the "<u>Award Notice</u>") as of the date set forth in the Award Notice (the "<u>Grant Date</u>"), pursuant to the provisions of the CDW Corporation 2021 Long-Term Incentive Plan (the "<u>Plan</u>"), a restricted stock unit award (the "<u>Award</u>") with respect to the number of shares of the Company's Common Stock, par value \$0.01 per share ("<u>Stock</u>"), set forth in the Award Notice, upon and subject to the restrictions, terms and conditions set forth in the Plan and this agreement (the "<u>Agreement</u>"). Capitalized terms not defined herein shall have the meanings specified in the Plan.

- 1. <u>Award Subject to Acceptance of Agreement</u>. The Award shall be null and void unless the Holder accepts this Agreement by executing the Award Notice in the space provided therefor and returning an original execution copy of the Award Notice to the Company or electronically accepting this Agreement within the Holder's stock plan account with the Company's stock plan administrator according to the procedures then in effect.
- Rights as a Stockholder. The Holder shall not be entitled to any privileges of ownership with respect to the shares of Stock subject to the Award unless and until, and only to the extent, such shares become vested pursuant to Section 3 hereof and the Holder becomes a stockholder of record with respect to such shares. As of each date on which the Company pays a cash dividend to record owners of shares of Stock (a "Dividend Date"), the Holder shall have no entitlement to receive such cash dividend, and the number of shares subject to the Award shall increase by (i) the product of the total number of shares subject to the Award immediately prior to such Dividend Date multiplied by the dollar amount of the cash dividend paid per share of Stock by the Company on such Dividend Date, divided by (ii) the Fair Market Value of a share of Stock on such Dividend Date. Any such additional shares shall be subject to the same vesting conditions and payment terms set forth herein as the shares to which they relate.
 - 3. Restriction Period and Vesting.
- 3.1. <u>Service-Based Vesting Condition</u>. Except as otherwise provided in this <u>Section 3</u>, the Award shall vest in accordance with the vesting schedule set forth in the Award Notice. The period of time prior to the full vesting of the Award shall be referred to herein as the "<u>Restriction Period</u>."
 - 3.2. <u>Termination of Employment.</u>
- (a) <u>Termination of Employment Due to Death or Disability</u>. If the Holder's employment with the Company and/or a Subsidiary terminates prior to the end of the Restriction Period by reason of the Holder's death or Disability, then in any such case, the Award shall be 100% vested upon such termination of employment.
- (b) <u>Termination of Employment Other Than Due to Death or Disability</u>. If the Holder's employment with the Company and/or a Subsidiary terminates prior to the end of the Restriction Period and prior to a Change in Control for any reason other than death or Disability, then the Award shall be immediately and automatically forfeited by the Holder and cancelled by the Company.

3.3. Change in Control.

- (a) <u>Vesting of Award Not Assumed</u>. In the event of a Change in Control prior to the end of the Restriction Period pursuant to which the Award is not effectively assumed or continued by the surviving or acquiring corporation in such Change in Control (as determined by the Board or Committee, with appropriate adjustments to the number and kind of shares, in each case, that preserve the value of the shares subject to the Award and other material terms and conditions of the outstanding Award as in effect immediately prior to the Change in Control), the Award shall vest in its entirety as of the date of the Change in Control.
- (b) <u>Vesting of Award Assumed</u>. In the event of a Change in Control prior to the end of the Restriction Period pursuant to which the Award is effectively assumed or continued by the surviving or acquiring corporation in such Change in Control (as determined by the Board or Committee, with appropriate adjustments to the number and kind of shares, in each case, that preserve the value of the shares subject to the Award and other material terms and conditions of the outstanding Award as in effect immediately prior to the Change in Control) and (i) the Holder remains continuously employed through the end of the Restriction Period, (ii) the Company terminates the Holder's employment without Cause or the Holder resigns for Good Reason within 24 months following such Change in Control and the Holder executes and does not revoke a waiver and release of claims in the form prescribed by the Company within 60 days after the date of such termination or (iii) the Holder's employment terminates due to death or Disability following such Change in Control, in any such case, the Award shall become fully vested as of the end of the Restriction Period or, if earlier, the Holder's termination of employment. If, following a Change in Control, the Holder experiences a termination of employment other than as set forth in this Section 3.3(b), the Award shall be immediately and automatically forfeited by the Holder and cancelled by the Company.

3.4. Definitions.

- (a) <u>Cause</u>. For purposes of this Award, "<u>Cause</u>" shall mean one or more of the following: (A) the Holder's refusal (after written notice and reasonable opportunity to cure) to perform duties properly assigned which are consistent with the scope and nature of the Holder's position; (B) the Holder's commission of an act materially and demonstrably detrimental to the financial condition and/or goodwill of the Company or any of its Subsidiaries, which act constitutes gross negligence or willful misconduct in the performance of duties to the Company or any of its Subsidiaries; (C) the Holder's commission of any theft, fraud, act of dishonesty or breach of trust resulting in or intended to result in material personal gain or enrichment of the Holder at the direct or indirect expense of the Company or any of its Subsidiaries; (D) the Holder's conviction of, or plea of guilty or nolo contendere to, a felony; (E) the Holder's material violation of any Restrictive Covenant; or (F) the Holder's material and willful violation of the Company's written policies or of the Holder's statutory or common law duty of loyalty to the Company or its affiliates that in either case is materially injurious to the Company, monetarily or otherwise. No act or failure to act will be considered "willful" (x) unless it is done, or omitted to be done, by the Holder in bad faith or without reasonable belief that the Holder's action or omission was in the best interests of the Company or (y) if it is done, or omitted to be done, in reliance on the informed advice of the Company's outside counsel or independent accountants or at the express direction of the Board.
- (b) <u>Disability</u>. For purposes of this Award, "<u>Disability</u>" shall mean the Holder's absence from the Holder's duties with the Company on a full-time basis for at least 180 consecutive days as a result of the Holder's incapacity due to physical or mental illness, or under such other circumstances as the Committee determines, in its sole discretion, constitute a Disability.

- (c) Good Reason. For purposes of this Award, "Good Reason" shall mean that the Holder resigns from employment with the Company and its Subsidiaries as a result of one or more of the following reasons: (i) the Company reduces the amount of the Holder's base salary or cash bonus opportunity (it being understood that the Board shall have discretion to set the Company's and the Holder's personal performance targets to which the cash bonus will be tied), (ii) the Company adversely changes the Holder's reporting responsibilities, titles or office as in effect as of the date hereof or reduces his/her position, authority, duties, responsibilities or status materially inconsistent with the positions, authority, duties, responsibilities or status the Holder then holds, (iii) any successor to the Company in any merger, consolidation or transfer of assets does not expressly assume any material obligation of the Company to the Holder under any agreement or plan pursuant to which the Holder receives benefits or rights, or (iv) the Company changes the Holder's place of work to a location more than fifty (50) miles from the Holder's present place of work; provided, however, that the occurrence of any such condition shall not constitute Good Reason unless (A) the Holder provides written notice to the Company of the existence of such condition not later than 60 days after the Holder knows or reasonably should know of the existence of such condition, (B) the Company fails to remedy such condition within 30 days after receipt of such notice and (C) the Holder resigns due to the existence of such condition within 60 days after the expiration of the remedial period described in clause (B) hereof.
- (d) <u>Restrictive Covenant</u>. For purposes of this Award, "<u>Restrictive Covenant</u>" shall mean any non-competition, non-solicitation, confidentiality or protection of trade secrets (or similar provision regarding intellectual property) covenant by which Holder is bound under any agreement between Holder and the Company and its Subsidiaries.
- 4. <u>Issuance or Delivery of Shares</u>. Subject to <u>Section 7.12</u> and except as otherwise provided for herein, within 70 days after the vesting of the Award (or if the Holder vests during the Restriction Period under <u>Section 3.2(a)</u>, no later than 70 days following the end of the calendar year in which the Holder vests under <u>Section 3.2(a)</u>), the Company shall issue or deliver, subject to the conditions of this Agreement, the vested shares of Stock to the Holder. Such issuance or delivery shall be evidenced by the appropriate entry on the books of the Company or of a duly authorized transfer agent of the Company. The Company shall pay all original issue or transfer taxes and all fees and expenses incident to such issuance or delivery, except as otherwise provided in <u>Section 7</u>. Prior to the issuance to the Holder of the shares of Stock subject to the Award, the Holder shall have no direct or secured claim in any specific assets of the Company or in such shares of Stock, and will have the status of a general unsecured creditor of the Company.

5. Clawback of Proceeds.

5.1. Clawback of Proceeds. This award is subject to the clawback provisions in Section 5.15 of the Plan. In addition, if the Holder materially violates any Restrictive Covenant and such violation occurs on or before the third anniversary of the date of the Holder's termination of employment: (i) the Award shall be forfeited and (ii) any and all Award Proceeds (as hereinafter defined) shall be immediately due and payable by the Holder to the Company. For purposes of this Section, "Award Proceeds" shall mean, with respect to any portion of the Award which is settled later than 24 months prior to the date of the Holder's termination of employment or service with the Company the Fair Market Value of a share of Stock on the date such portion of the Award was settled, multiplied by the number of shares of Stock issued to the Holder pursuant to the settlement of such portion of the Award. The remedy provided by this Section shall be in addition to and not in lieu of any rights or remedies which the Company may have against the Holder in respect of a breach by the Holder of any duty or obligation to the Company.

5.2. <u>Right of Setoff</u>. The Holder agrees that by accepting the Award the Holder authorizes the Company and its affiliates to deduct any amount or amounts owed by the Holder pursuant to this <u>Section 5</u> from any amounts payable by or on behalf of the Company or any affiliate to the Holder, including, without limitation, any amount payable to the Holder as salary, wages, vacation pay, bonus or the vesting or settlement of the Award or any stock-based award. This right of setoff shall not be an exclusive remedy and the Company's or an affiliate's election not to exercise this right of setoff with respect to any amount payable to the Holder or any other remedy.

6. <u>Transfer Restrictions and Investment Representation.</u>

- 6.1. Nontransferability of Award. The Award may not be transferred by the Holder other than by will or the laws of descent and distribution. Except to the extent permitted by the foregoing sentence, the Award may not be sold, transferred, assigned, pledged, hypothecated, encumbered or otherwise disposed of (whether by operation of law or otherwise) or be subject to execution, attachment or similar process. Upon any attempt to so sell, transfer, assign, pledge, hypothecate, encumber or otherwise dispose of the Award, the Award and all rights hereunder shall immediately become null and void.
- 6.2. <u>Investment Representation</u>. The Holder hereby covenants that (a) any sale of any share of Stock acquired upon the vesting of the Award shall be made either pursuant to an effective registration statement under the Securities Act of 1933, as amended (the "<u>Securities Act</u>"), and any applicable state securities laws, or pursuant to an exemption from registration under the Securities Act and such state securities laws and (b) the Holder shall comply with all regulations and requirements of any regulatory authority having control of or supervision over the issuance of the shares and, in connection therewith, shall execute any documents which the Committee shall in its sole discretion deem necessary or advisable.

7. Additional Terms and Conditions of Award.

7.1. Withholding Taxes. As a condition precedent to the issuance or delivery of the Stock upon the vesting of the Award, at the Company's discretion either (i) the Holder shall pay to the Company such amount as the Company (or an affiliate) determines is required, under all applicable federal, state, local, foreign or other laws or regulations, to be withheld and paid over as income or other withholding taxes (the "Required Tax Payments") with respect to the Award or (ii) the Company or an affiliate may, in its discretion, deduct any Required Tax Payments from any amount then or thereafter payable by the Company or an affiliate to the Holder, which may include the withholding of whole shares of Stock which would otherwise be delivered to the Holder having an aggregate Fair Market Value, determined as of the date on which such withholding obligation arises, equal to the Required Tax Payments, in either case in accordance with such terms, conditions and procedures that may be prescribed by the Company. Shares of Stock withheld may not have a Fair Market Value in excess of the amount determined by applying the maximum individual statutory tax rate in the Holder's jurisdiction; provided that the Company shall be permitted to limit the number of shares so withheld to a lesser number if necessary, as determined by the Company, to avoid adverse accounting consequences or for administrative convenience; provided, however, that if a fraction of a share of Stock would be required to satisfy the maximum individual statutory rate in the Holder's jurisdiction, then the number of shares of Stock to be withheld may be rounded up to the next nearest whole share of Stock. No certificate representing a share of Stock shall be delivered until the Required Tax Payments have been satisfied in full. Any determination by the Company with respect to the withholding of shares of Stock to satisfy the Required Tax Payments shall be made by the Committee if the Holder is subject to Section 16 of the Exchange Act.

- 7.2. Compliance with Applicable Law. The Award is subject to the condition that if the listing, registration or qualification of the shares of Stock subject to the Award upon any securities exchange or under any law, or the consent or approval of any governmental body, or the taking of any other action is necessary or desirable as a condition of, or in connection with, the delivery of shares hereunder, the shares of Stock subject to the Award shall not be delivered, in whole or in part, unless such listing, registration, qualification, consent, approval or other action shall have been effected or obtained, free of any conditions not acceptable to the Company. The Company agrees to use reasonable efforts to effect or obtain any such listing, registration, qualification, consent, approval or other action.
- 7.3. <u>Award Confers No Rights to Continued Employment</u>. In no event shall the granting of the Award or its acceptance by the Holder, or any provision of the Agreement or the Plan, give or be deemed to give the Holder any right to continued employment by the Company, any Subsidiary or any affiliate of the Company or affect in any manner the right of the Company, any Subsidiary or any affiliate of the Company to terminate the employment of any person at any time.
- 7.4. <u>Decisions of Board or Committee</u>. The Board or the Committee shall have the right to resolve all questions which may arise in connection with the Award. Any interpretation, determination or other action made or taken by the Board or the Committee regarding the Plan or this Agreement shall be final, binding and conclusive.
- 7.5. <u>Successors</u>. This Agreement shall be binding upon and inure to the benefit of any successor or successors of the Company and any person or persons who shall, upon the death of the Holder, acquire any rights hereunder in accordance with this Agreement or the Plan.
- 7.6. Notices. All notices, requests or other communications provided for in this Agreement shall be made, if to the Company, to CDW Corporation, Attn: General Counsel, 200 N. Milwaukee Avenue, Vernon Hills, Illinois 60061, and if to the Holder, to the last known mailing address of the Holder contained in the records of the Company. All notices, requests or other communications provided for in this Agreement shall be made in writing either (a) by personal delivery, (b) by facsimile or electronic mail with confirmation of receipt, (c) by mailing in the United States mails or (d) by express courier service. The notice, request or other communication shall be deemed to be received upon personal delivery, upon confirmation of receipt of facsimile or electronic mail transmission or upon receipt by the party entitled thereto if by United States mail or express courier service; provided, however, that if a notice, request or other communication sent to the Company is not received during regular business hours, it shall be deemed to be received on the next succeeding business day of the Company.
- 7.7. <u>Governing Law</u>. This Agreement, the Award and all determinations made and actions taken pursuant hereto and thereto, to the extent not governed by the laws of the United States, shall be governed by the laws of the State of Delaware and construed in accordance therewith without giving effect to principles of conflicts of laws.
- 7.8. Agreement Subject to the Plan. This Agreement is subject to the provisions of the Plan and shall be interpreted in accordance therewith. In the event that the provisions of this Agreement and the Plan conflict, the Plan shall control. The Holder hereby acknowledges receipt of a copy of the Plan.
- 7.9. Entire Agreement. This Agreement and the Plan constitute the entire agreement of the parties with respect to the subject matter hereof and supersede in their entirety all prior undertakings and agreements of the Company and the Holder with respect to the subject

matter hereof, and may not be modified adversely to the Holder's interest except by means of a writing signed by the Company and the Holder.

- 7.10. <u>Partial Invalidity</u>. The invalidity or unenforceability of any particular provision of this Agreement shall not affect the other provisions hereof and this Agreement shall be construed in all respects as if such invalid or unenforceable provision was omitted.
- 7.11. <u>Amendment and Waiver</u>. The Company may amend the provisions of this Agreement at any time; <u>provided</u> that an amendment that would adversely affect the Holder's rights under this Agreement shall be subject to the written consent of the Holder. No course of conduct or failure or delay in enforcing the provisions of this Agreement shall affect the validity, binding effect or enforceability of this Agreement.
- 7.12. Compliance With Section 409A of the Code. This Award is intended to be exempt from or comply with Section 409A of the Code, and shall be interpreted and construed accordingly. To the extent this Agreement provides for the Award to become vested and be settled upon the Holder's termination of employment, the applicable shares of Stock shall be transferred to the Holder or his or her beneficiary upon the Holder's "separation from service," within the meaning of Section 409A of the Code, then to the extent the Award constitutes nonqualified deferred compensation, within the meaning of Section 409A of the Code, such shares of Stock shall be transferred to the Holder or his or her beneficiary upon the earlier to occur of (i) the six-month anniversary of such separation from service and (ii) the date of the Holder's death.

	Exhibit 10.23
CDIA Director Deferred Commencation Dlan Adoption Agreement	
CDW Director Deferred Compensation Plan Adoption Agreemer	IL

Table of Contents

1.01	Preamble	1
1.02	Plan	1
1.03	Plan Sponsor	1
1.04	Employer	2
1.05	Administrator	2
1.06	Key Employee Determination Dates	2
2.01	Participation	3
3.01	Compensation	4
3.02	Bonuses	5
4.01	Participant Contributions	6
5.01	Employer Contributions	8
6.01	Distributions	11
7.01	Vesting	17
8.01	Unforeseeable Emergency	21
9.01	Investment Decisions	22
10.01	Trust	23
11.01	Termination Upon Change In Control	24
11.02	Automatic Distribution Upon Change In Control	24
11.03	Change In Control	24
12.01	Governing State Law	25
Appendix A		27

-i-January 2022

Adoption Agreement

1.01 Preamble

	By the execution of this Adoption Agreement the Plan Sponsor hereby [complete (a) or (b)]
	(a) ⊠ adopts a new plan as of <u>January 1, 2022</u>
	(b) amends and restates its existing plan as of [month, day, year] which is the Amendment Effective Date. Except as otherwise provided in Appendix A, all amounts deferred under the Plan prior to the Amendment Effective Date shall be governed by the terms of the Plan as in effect on the day before the Amendment Effective Date.
	Original Effective Date: [month, day, year]
	Pre-409A Grandfathering: \square Yes \square No
1.02	Plan
	Plan Name: CDW Director Deferred Compensation Plan
	Plan Year: January 1 – December 31
1.03	Plan Sponsor
	Name: CDW Corporation
	Address: 200 North Milwaukee Avenue, Vernon Hills, Illinois 60061
	Phone #: 847-465-6000
	EIN #: 36-3310735
	Fiscal Year: January 1 – December 31
	Is stock of the Plan Sponsor, any Employer or any Related Employer publicly traded on an established securities market? ⊠ Yes □ No

1.04 Employer

The following entities have been authorized by the Plan Sponsor to participate in and have adopted the Plan [insert "Not Applicable" if none have been authorized]:

Entity Publicly Traded on Est. Securities Market Yes No

1.05 Administrator

The Plan Sponsor has designated the following party or parties to be responsible for the administration of the Plan:

Name: CDW Corporation

Address: 200 North Milwaukee Avenue, Vernon Hills, Illinois 60061

Note: The Administrator is the person or persons designated by the Plan Sponsor to be responsible for the administration of the Plan. Neither Fidelity Employer Services Company nor any other Fidelity affiliate can be the Administrator.

1.06 Key Employee Determination Dates

The Employer has designated [month, day, year] as the Identification Date for purposes of determining Key Employees.

In the absence of a designation, the Identification Date is December 31.

The Employer has designated [month, day, year] as the effective date for purposes of applying the six month delay in distributions to Key Employees.

In the absence of a designation, the effective date is the first day of the fourth month following the Identification Date.

2.01 Participation

	(a)	\square Emplo	oyees [complete (i), (ii) or (iii)]
		(i)	\square Eligible Employees are selected by the Employer.
follow	ring c	(ii) riteria:	$\hfill\Box$ Eligible Employees are those employees of the Employer who satisfy the
		(iii)	oximes Employees are not eligible to participate.
	(b)	⊠ Direct	tors [complete (i), (ii) or (iii)]
		(i) 🛮 Al	l nonemployee Directors are eligible to participate.
		(ii)	\square Only Directors selected by the Employer are eligible to participate.
		(iii)	\square Directors are not eligible to participate.

-3-

3.01 Compensation

	For p Comp	urpo ensa	oses of determining Participant contributions under Article 4 and Employer contributions under Article ation shall be defined in the following manner [complete (a) or (b) and select (c) and/or (d), if applicable]:	5,
	(a)		Compensation is defined as:	
imita	(b) ation in	□ n Sect	Compensation as defined in <u>[insert name of qualified plan]</u> without regard to the tion 401(a)(17) of the Code for such Plan Year.	
	(c)	\boxtimes	Director Compensation is defined as:	
			Cash retainer fees	
	(4)		Communication of all formall plan annual to binite of the de	
	(d)		Compensation shall, for all Plan purposes, be limited to \$	
	(e)		Not Applicable.	
4-			January	v 2023
			January	y 2022

3.02 Bonuses

Compensation, as defined in Section 3.01 of the Adoption Agreement, includes the following type of bonuses that will be the subject of a separate deferral election:

[Will be treated as]	
<u>Performance Base</u>	ed Compensation
Yes	No
	[Will be to Performance Base Yes

 \boxtimes Not Applicable.

4.01 Participant Contributions

If Participant contributions are permitted, complete (a), (b), and (c). Otherwise complete (d).

(a) Amount of Deferrals

A Participant may elect within the period specified in Section 4.01(b) of the Adoption Agreement to defer the following amounts of remuneration. For each type of remuneration listed, complete "dollar amount" and/or "percentage amount".

(i) Compensation other than Bonuses [do not complete if you complete (iii)]

	Dollar	Amount	% Am	ount	Increment
Type of Remuneration	Min	Max	Min	Max	
			%	%	%
			%	%	%
			%	%	%

Note: The increment is required to determine the permissible deferral amounts. For example, a minimum of 0% and maximum of 20% with a 5% increment would allow an individual to defer 0%, 5%, 10%, 15% or 20%.

(ii) Bonuses [do not complete if you complete (iii)]

	Dollar Amount		% Amo	unt	Increment
Type of Bonus	Min	Max	Min	Max	
			%	%	%
			%	%	%
			%	%	%

(iii) Compensation [do not complete if you completed (i) and (ii)]

		% Amount	Amount	Dollar .
Increment	Max	Min	Max	Min
%	%	%		

(iv) Director Compensation

	Dollar Am	ount	% Amour	nt	Increment
Type of Compensation	Min	Max	Min	Max	
Cash Retainer Fees			1%	100%	1%
Other:			%	%	%
Other:			%	%	%

(b) Election Period

(c) No Participant Contributions

-7-

 $\hfill\square$ Participant contributions are not permitted under the Plan.

(i)	Performance Based Compensation A special election period
\square Does	
⊠ Does Not	
	apply to each eligible type of performance based compensation referenced in Section 3.02 of the Adoption Agreement.
	The special election period, if applicable, will be determined by the Employer.
(ii)	Newly Eligible Participants
	An employee who is classified or designated as an Eligible Employee during a Plan Year
□ May	
⊠ May Not	
	elect to defer Compensation earned during the remainder of the Plan Year by completing a deferral agreemen within the 30 day period beginning on the date he is eligible to participate in the Plan.
	The special election period, if applicable, will be determined by the Employer.

5.01 Employer Contributions

If Employer contributions are permitted, complete (a) and/or (b). Otherwise complete (c).

(a)	Matching Contributions	
	(2)	

(i)	Amount
	For each Plan Year, the Employer shall make a matching contribution on behalf of each Participant who defers Compensation for the Plan Year and satisfies the requirements of Section 5.01(a)(ii) of the Adoption Agreement equal to [complete the ones that are applicable]:
	(A) \Box <u>[insert percentage]%</u> of the Compensation the Participant has elected to defer for the Plan Year
	(B) \square An amount determined by the Employer in its sole discretion
	(C) ☐ Matching contributions for each Participant shall be limited to \$\(\sigma\) and/or [\(\text{insert percentage}\)]\(\sigma\) of Compensation
	(D) \square Other:
	(E)
(ii)	Eligibility for matching contribution
	A Participant who defers Compensation for the Plan Year shall receive an allocation of matching contributions determined in accordance with Section 5.01(a)(i) provided he satisfies the following requirements [complete the ones that are applicable]:
	(A) Describe requirements:
	(B) \square Is selected by the Employer in its sole discretion to receive an allocation of matching contributions
	(C) \square No requirements

	(iii)	Time of Allocation
		Matching contributions, if made, shall be treated as allocated [select one]:
		(A) \Box As of the last day of the Plan Year
		(B) \square At such times as the Employer shall determine in its sole discretion
		(C) ☐ At the time the Compensation on account of which the matching contribution is being made would otherwise have been paid to the Participant
		(D) \square Other:
(b)	Othe	er Contributions
(-)	(i)	Amount
		The Employer shall make a contribution on behalf of each Participant who satisfies the requirements of Section 5.01(b)(ii) equal to [complete the ones that are applicable]:
		(A) ☐ An amount equal to [insert percentage]% of the Participant's Compensation
		(B) \square An amount determined by the Employer in its sole discretion
		(C) \square Contributions for each Participant shall be limited to \$
		(D) \square Other:
		(E)
		J

	(ii)	Eligibility for Other Contribution	
		A Participant shall receive an allocation of other Employer contributions determined in accordance v Section 5.01(b)(i) for the Plan Year if he satisfies the following requirements [complete the one that applicable]:	vith t is
		(A) Describe requirements:	
		(B) \square Is selected by the Employer in its sole discretion to receive an allocation of other Employer contributions	
		(C) \square No requirements	
	(iii)	Time of Allocation	
		Employer contributions, if made, shall be treated as allocated [select one]:	
		(A) \Box As of the last day of the Plan Year	
		(B) \square At such times or times as the Employer shall determine in its sole discretion	
		(C) □ Other:	
(c)	No E	Employer Contributions	
		oxtimes Employer contributions are not permitted under the Plan.	
		I	202

6.01 Distributions

The timing and form of payment of distributions made from the Participant's vested Account shall be made in accordance with the elections made in this Section 6.01 of the Adoption Agreement except when Section 9.6 of the Plan requires a six month delay for certain distributions to Key Employees of publicly traded companies.

(a)	Timing	of	Distributions

(i)	All distributions shall commence in accordance with the following [choose one]:
	(A) ⊠ As soon as administratively feasible following the distribution event but in no event later than the time prescribed by Treas. Reg. Sec. 1.409A-3(d).
	(B) ☐ Monthly on specified day [insert day]
	(C) \square Annually on specified month and day [insert month and day]
	(D) \square Calendar quarter on specified month and day [insert month and day] Q[insert numerical quarter 1, 2, 3, or 4]
(ii)	The timing of distributions as determined in Section 6.01(a)(i) shall be modified by the adoption of:
	(A) ☐ Event Delay – Distribution events other than those based on Specified Date or Specified Age will be treated as not having occurred for months
	(B) ☐ Hold Until Next Year – Distribution events other than those based on Specified Date or Specified Age will be treated as not having occurred for twelve months from the date of the event if payment pursuant to Section 6.01(a)(i) will thereby occur in the next calendar year or on the first payment date in the next calendar year in all other cases
	(C) ☐ Immediate Processing – The timing method selected by the Plan Sponsor under Section 6.01(a)(i) shall be overridden for the following distribution events [insert events]:
	(D) ⊠ Not applicable

(b) Distribution Events

(i) Participant Contributions under Section 4.01(a)

Participants may elect the following payment events and the associated form or forms of payment. If multiple events for each year are selected, the earliest to occur will trigger payment. For installments, insert the range of available periods (e.g., 5-15) or insert the periods available (e.g., 5, 7, 9).

		<u>Lump Sum</u>	Installments
(A)	⊠ Specified Date	\boxtimes	<u>5-15</u> years
(B)	☐ Specified Age		years
(C)	⊠ Separation from Service	\boxtimes	<u>5-15</u> years
(D)	$\hfill \square$ Separation from Service plus 6 months		years
(E)	☐ Separation from Service plus months [not to exceedmonths]		years
(F)	☐ Retirement		years
(G)	☐ Retirement plus 6 months		years
(H)	☐ Retirement plus_months		years
(I)	☐ Disability		years
(J)	□ Death		years
(K)	☐ Change in Control		years
	inimum deferral period for Specified Date or Specifi ments may be paid [select each that applies]	ed Age event shall l	be <u>2 y</u> ears.
□ Mo ⊠ Qua ⊠ An	arterly		

ı	· · · · ·		1 0 .			<i>~</i> ~
ı	iil	Employer Contributions	s under Section	5 0113	al and l	۱h۱
۱	11)	Linployer Continuation,	s under section		aj anu j	v

Participants may elect the following payment events and the associated form or forms of payment. If multiple events for each year are selected, the earliest to occur will trigger payment. For installments, insert the range of available periods (e.g., 5-15) or insert the periods available (e.g., 5, 7, 9).

		Lump Sum	<u>Installments</u>
(A)	☐ Specified Date		years
(B)	☐ Specified Age		years
(C)	☐ Separation from Service		years
(D)	\square Separation from Service plus 6 months		years
(E) exce	☐ Separation from Service plus _ months [not to ed_months]		years
(F)	☐ Retirement		years
(G)	☐ Retirement plus 6 months		years
(H)	☐ Retirement plus_months		years
(I)	□ Disability		years
(J)	□ Death		years
(K)	\Box Change in Control		years
The year	minimum deferral period for Specified Date or Specifie s.	d Age event shall b	e
Inst	allments may be paid [select each that applies]		
□ M	Ionthly		
\Box C	uarterly		

 \square Annually

-13-

(c)	Specified Date and Specified Age elections may not extend beyond age [insert age or "Not Applicab applies].	<u>le" if no r</u>	naximum age
(d)	Payment Election Override		
	Payment of the remaining vested balance of the Participant's Account will automatically occur a Section 6.01(a) of the Adoption Agreement in the form indicated upon the earliest to occur o [check each event that applies and for each event include only a single form of payment]:	t the tim f the foll	e specified in owing events
	Events Form of Payment Lui	mp Sum	<u>Installments</u>
	☐ Separation from Service ☐		
	\square Separation from Service before Retirement \square		
×	☑ Death ☑		
	□ Disability □		
	□ Not Applicable □		
(e)	Involuntary Cashouts		
	☑ If the Participant's vested Account at the time of his Separation from Service does not exceed \$50,000, distribution of the vested Account shall automatically be made in the form of a single lump sum in accordance with Section 9.5 of the Plan.		
	☐ There are no involuntary cashouts.		
(f)	Retirement		
	□ Retirement shall be defined as a Separation from Service that occurs on or after the Participant [insert description of requirements]:		
	oxtimes No special definition of Retirement applies.		
-14-			January 20

(g)	Distribution Election Change
	A Participant
	⊠ Shall
	Shall Not
	be permitted to modify a scheduled distribution date and/or payment option in accordance with Section 9.2 of the Plan.
	A Participant shall generally be permitted to elect such modification a maximum of <u>5</u> times.
	Administratively, allowable distribution events will be modified to reflect all options necessary to fulfill the distribution change election provision.
(h)	Frequency of Elections
	The Plan Sponsor
	⊠ Has
	Has Not
	elected to permit annual elections of a time and form of payment for amounts deferred under the Plan. If a single election of a time and/or form of payment is required, the Participant will make such election at the time he first completes a deferral agreement which, in all cases, will be no later than the time required by Reg. Sec. 1.409A-2.

-15-

(i)	Disability
	For Purposes of Section 2.11 of the Plan, Disability shall be defined as
	\square Total disability as determined by the Social Security Administration or the Railroad Retirement Board.
[\square As determined by the Employer's long term disability insurance policy.
[☐ As follows [insert description of requirements]:

-16-January 2022

oxtimes Not applicable.

7.1 Vesting

(a) Matching Contributions

The Participant's vested interest in the amount credited to his Account attributable to matching contributions shall be based on the following schedule:

basca on	the following schedul	ic.	
	Years of Service	Vesting %	
	0	<u> </u>	[Insert "100" if there is immediate vesting]
	1	<u> </u>	
	2	<u> </u>	
	3	<u> </u>	
	4	<u> </u>	
	5	<u> </u>	
	6	<u> </u>	
	7	<u> </u>	
	8	<u> </u>	
	9	<u> %</u>	
□ Othe	r:		

 \square Class year vesting applies:

 \boxtimes Not applicable.

(b) Other Employer Contributions

The Participant's vested interest in the amount credited to his Account attributable to Employer contributions other than matching contributions shall be based on the following schedule:

-			8
	Years of Service	Vesting %	
	0	<u></u> %	[Insert "100" if there is immediate vesting]
	1	<u></u> %	
	2	<u> </u>	
	3	<u></u> %	
	4	<u> %</u>	
	5	<u> %</u>	
	6	<u></u> %	
	7	<u> </u>	
	8	<u></u> %	
	9	<u> %</u>	

 \square Other:

☐ Class year vesting applies:

 $oxed{\boxtimes}$ Not applicable.

-18-

(c)	Acceleration of Vesting				
	The Participant's vested interest in his Account will automatically be 100% upon the occurrence of the following events [select the ones that are applicable]:				
	(i)	\square Death.			
	(ii)	\square Disability.			
	(iii)	\square Change in Control.			
	(iv)	\square Eligibility for Retirement.			
	(v)	□ Other:			
	(vi) ⊠ Not applicable.				
(d)	Years of Service				
	(i)	A Participant's Years of Service shall include all service performed for the Employer and			
		□ Shall			
		☐ Shall Not			
		include service performed for the Related Employer.			
)_					

-19-

	(ii)	Years of	Service shall also include service performed for the following entities:	
	(iii)	Years of	Service shall be determined in accordance with [select one]:	
		(A) \Box	The elapsed time method in Treas. Reg. Sec. 1.410(a)-7	
		(B) □	The general method in DOL Reg. Sec. 2530.200b-1 through b-4	
		(C) 🗆	Participant's Years of Service credited under:	
		(D) 🗆	Other:	
-20-		(iv) ⊠	Not applicable.	January 2022

8.1 Unforeseeable Emergency

(a)	A withdrawal due to an Unforeseeable Emergency as defined in Section 2.24:
	⊠ Will
	☐ Will Not [if Unforeseeable Emergency withdrawals are not permitted, proceed to Section 9.01]
	be allowed.
(b)	Upon a withdrawal due to an Unforeseeable Emergency, a Participant's deferral election for the remainder of the Plan Year:
	□ Will
	⊠ Will Not
	be cancelled. If cancellation occurs, the Participant may resume participation in accordance with Article 4 of the Plan.

-21-

9.01 Investment Decisions

Investment decisions	regarding the hyn	othetical amount	s credited to a Partic	inant's Account sha	ll he made hy	[select one]
mivestificht accisions	regarding the hyp	otiicticai aiiiotiit	s ci cuitca to a i ai tic	ipani s riccount sin	in be made by	[SCICCE OffC]

(a) 🗵 The Participant or his Beneficiary

(b) ☐ The Employer

-22-

10.01Trust

The Employer [select one]:
\square Does
⊠ Does Not
intend to establish a rabbi trust as provided in Article 11 of the Plan.

-23-

11.1	Termination Upon Change In Control			
	The Plan Sponsor			
	⊠ Reserves			
	□ Does Not Reserves			
	the right to terminate the Plan and distribute all vested amounts credited to Participant Accounts upon a Change in Control as described in Section 9.7.			
11.2	Automatic Distribution Upon Change In Control			
	Distribution of the remaining vested balance of each Participant's Account			
	□ Shall			
	⊠ Shall Not			
	automatically be paid as a lump sum payment upon the occurrence of a Change in Control as provided in Section 9.7.			
11.3	Change In Control			
	A Change in Control for Plan purposes includes the following [select each definition that applies]:			
	(a) \boxtimes A change in the ownership of the Employer as described in Section 9.7(c) of the Plan.			
	(b) \boxtimes A change in the effective control of the Employer as described in Section 9.7(d) of the Plan.			
	(c) \boxtimes A change in the ownership of a substantial portion of the assets of the Employer as described in Section 9.7(e) of the Plan.			
	(d) \square Not Applicable.			

-24-

12.01Governing State Law

The laws of <u>Illinois</u> shall apply in the administration of the Plan to the extent not preempted by ERISA.

-25-

Execution Page

The Plan Sponsor has caused this Adoption Agreement to be executed this $__$ day of $___$, $\underline{20}$.

Plan Sponsor: CDW Corporation By: /s/ Elizabeth Connelly

Title: Chief Human Resources Officer

-26-

Appendix A

Special Effective Dates

Not Applicable

-27-

CDW Director Deferred Compensation Plan

January 1, 2022

IMPORTANT NOTE

This document has not been approved by the Department of Labor, Internal Revenue Service or any other governmental entity. An adopting Employer must determine whether the Plan is subject to the Federal securities laws and the securities laws of the various states. An adopting Employer may not rely on this document to ensure any particular tax consequences or to ensure that the Plan is "unfunded and maintained primarily for the purpose of providing deferred compensation to a select group of management or highly compensated employees" under Title I of the Employee Retirement Income Security Act of 1974, as amended, with respect to the Employer's particular situation. Fidelity Employer Services Company, its affiliates and employees cannot provide you with legal advice in connection with the execution of this document. This document should be reviewed by the Employer's attorney prior to execution.

-28-

Table of Contents

Preamble		1
Article 1 - General		1-1
1.1.	Plan	1-1
1.2.	Effective Dates	1-1
1.3.	Amounts Not Subject to Code Section 409A	1-1
Article 2 - Definitions		2-1
2.1.	Account	2-1
2.2.	Administrator	2-1
2.3.	Adoption Agreement	2-1
2.4.	Beneficiary	2-1
2.5.	Board or Board of Directors	2-1
2.6.	Bonus	2-1
2.7.	Change in Control	2-1
2.8.	Code	2-1
2.9.	Compensation	2-1
2.10.	Director	2-2
2.11.	Disability	2-2
2.12.	Eligible Employee	2-2
2.13.	Employer	2-2
2.14.	ERISA	2-2
2.15.	Identification Date	2-2
2.16.	Key Employee	2-2
2.17.	Participant	2-2
2.18	Plan	2-2

CDW LLC CDW Director Deferred Compensation Plan TOC-i

2.19.	Plan Sponsor	2-3
2.20.	Plan Year	2-3
2.21.	Related Employer	2-3
2.22.	Retirement	2-3
2.23.	Separation from Service	2-3
2.24.	Unforeseeable Emergency	2-4
2.25.	Valuation Date	2-4
2.26.	Years of Service	2-4
Article 3 - Participation		3-1
3.1.	Participation	3-1
3.2.	Termination of Participation	3-1
Article 4 - Participant Elections		4-1
4.1.	Deferral Agreement	4-1
4.2.	Amount of Deferral	4-1
4.3.	Timing of Election to Defer	4-1
4.4.	Election of Payment Schedule and Form of Payment	4-2
Article 5 - Er	mployer Contributions	5-1
5.1.	Matching Contributions	5-1
5.2.	Other Contributions	5-1
Article 6 - Ac	ccounts and Credits	6-1
6.1.	Establishment of Account	6-1
6.2.	Credits to Account	6-1
Article 7 - In	envestment of Contributions	7-1
7.1.	Investment Options	7-1
7.2.	Adjustment of Accounts	7-1

CDW LLC CDW Director Deferred Compensation Plan TOC-ii

Article 8 - Rig	ght to Benefits	8-1
8.1.	Vesting	8-1
8.2.	Death	8-1
8.3.	Disability	8-1
Article 9 - Di	stribution of Benefits	9-1
9.1.	Amount of Benefits	9-1
9.2.	Method and Timing of Distributions	9-1
9.3.	Unforeseeable Emergency	9-1
9.4.	Payment Election Overrides	9-2
9.5.	Cashouts of Amounts Not Exceeding Stated Limit	9-2
9.6.	Required Delay in Payment to Key Employees	9-2
9.7.	Change in Control	9-3
9.8.	Permissible Delays in Payment	9-6
9.9.	Permitted Acceleration of Payment	9-7
Article 10 - A	mendment and Termination	10-1
10.1.	Amendment by Plan Sponsor	10-1
10.2.	Plan Termination Following Change in Control or Corporate Dissolution	10-1
10.3.	Other Plan Terminations	10-1
Article 11 - T	'he Trust	11-1
11.1.	Establishment of Trust	11-1
11.2.	Rabbi Trust	11-1
11.3.	Investment of Trust Funds	11-1
Article 12 - P	lan Administration	12-1
12.1.	Powers and Responsibilities of the Administrator	12-1
12.2.	Claims and Review Procedures	12-1
12.3.	Plan Administrative Costs	12-1

CDW LLC CDW Director Deferred Compensation Plan TOC-iii

Article 13 - Miscellaneous		13-1
13.1.	Unsecured General Creditor of the Employer	13-1
13.2.	Employer's Liability	13-1
13.3.	Limitation of Rights	13-1
13.4.	Anti-Assignment	13-1
13.5.	Facility of Payment	13-2
13.6.	Notices	13-2
13.7.	Tax Withholding	13-2
13.8.	Indemnification	13-3
13.9.	Successors	13-4
13.10.	Disclaimer	13-4
13 11	Governing Law	13-4

CDW LLC CDW Director Deferred Compensation Plan TOC-iv

Preamble The Plan is intended to be a "plan which is unfunded and is maintained by an employer primarily for the purpose of providing deferred compensation for a select group of management or highly compensated employees" within the meaning of Sections 201(2), 301(a)(3) and 401(a)(1) of the Employee Retirement Income Security Act of 1974, as amended, or an "excess benefit plan" within the meaning of Section 3(36) of the Employee Retirement Income Security Act of 1974, as amended, or a combination of both. The Plan is further intended to conform with the requirements of Internal Revenue Code Section 409A and the final regulations issued thereunder and shall be interpreted, implemented and administered in a manner consistent therewith.

CDW, LLC CDW Director Deferred Compensation Plan Preamble

Article 1 - General

1.1. Plan

The Plan will be referred to by the name specified in the Adoption Agreement.

1.2. Effective Dates

- (a) Original Effective Date. The Original Effective Date is the date as of which the Plan was initially adopted.
- (b) <u>Amendment Effective Date.</u> The Amendment Effective Date is the date specified in the Adoption Agreement as of which the Plan is amended and restated. Except to the extent otherwise provided herein or in the Adoption Agreement, the Plan shall apply to amounts deferred and benefit payments made on or after the Amendment Effective Date.
- (c) <u>Special Effective Date.</u> A Special Effective Date may apply to any given provision if so specified in Appendix A of the Adoption Agreement. A Special Effective Date will control over the Original Effective Date or Amendment Effective Date, whichever is applicable, with respect to such provision of the Plan.

1.3. Amounts Not Subject to Code Section 409A

Except as otherwise indicated by the Plan Sponsor in Section 1.01 of the Adoption Agreement, amounts deferred before January 1, 2005 that are earned and vested on December 31, 2004 will be separately accounted for and administered in accordance with the terms of the Plan as in effect on December 31, 2004.

CDW, LLC CDW Director Deferred Compensation Plan Article 1-1

Article 2 - Definitions

Pronouns used in the Plan are in the masculine gender but include the feminine gender unless the context clearly indicates otherwise. Wherever used herein, the following terms have the meanings set forth below, unless a different meaning is clearly required by the context:

2.1. Account

"Account" means an account established for the purpose of recording amounts credited on behalf of a Participant and any income, expenses, gains, losses or distributions included thereon. The Account shall be a bookkeeping entry only and shall be utilized solely as a device for the measurement and determination of the amounts to be paid to a Participant or to the Participant's Beneficiary pursuant to the Plan.

2.2. Administrator

"Administrator" means the person or persons designated by the Plan Sponsor in Section 1.05 of the Adoption Agreement to be responsible for the administration of the Plan. If no Administrator is designated in the Adoption Agreement, the Administrator is the Plan Sponsor.

2.3. Adoption Agreement

"Adoption Agreement" means the agreement adopted by the Plan Sponsor that establishes the Plan.

2.4. Beneficiary

"Beneficiary" means the persons, trusts, estates or other entities entitled under Section 8.2 to receive benefits under the Plan upon the death of a Participant.

2.5. Board or Board of Directors

"Board" or "Board of Directors" means the Board of Directors of the Plan Sponsor.

2.6. Bonus

"Bonus" means an amount of incentive remuneration payable by the Employer to a Participant.

2.7. Change in Control

"Change in Control" means the occurrence of an event involving the Plan Sponsor that is described in Section 9.7.

2.8. Code

"Code" means the Internal Revenue Code of 1986, as amended.

2.9. **Compensation**

"Compensation" has the meaning specified in Section 3.01 of the Adoption Agreement.

2.10. Director

"Director" means a non-employee member of the Board who has been designated by the Employer as eligible to participate in the Plan.

2.11. Disability

"Disability" means a determination by the Administrator that the Participant is either (a) unable to engage in any substantial gainful activity by reason of any medically determinable physical or mental impairment which can be expected to result in death or can be expected to last for a continuous period of not less than 12 months, or (b) is, by reason of any medically determinable physical or mental impairment which can be expected to result in death or last for a continuous period of not less than twelve months, receiving income replacement benefits for a period of not less than three months under an accident and health plan covering employees of the Employer in accordance with the definition in Section 6.01(i) of the Adoption Agreement. A Participant will be considered to have incurred a Disability if he is determined to be totally disabled by the Social Security Administration or the Railroad Retirement Board.

2.12. Eligible Employee

"Eligible Employee" means an employee of the Employer who satisfies the requirements in Section 2.01 of the Adoption Agreement.

2.13. Employer

"Employer" means the Plan Sponsor and any other entity which is authorized by the Plan Sponsor to participate in and, in fact, does adopt the Plan.

2.14. ERISA

"ERISA" means the Employee Retirement Income Security Act of 1974, as amended.

2.15. Identification Date

"Identification Date" means the date as of which Key Employees are determined which is specified in Section 1.06 of the Adoption Agreement.

2.16. Key Employee

"Kev Employee" means an employee who satisfies the conditions set forth in Section 9.6.

2.17. Participant

"Participant" means an Eligible Employee or Director who commences participation in the Plan in accordance with Article 3.

2.18. Plan

"Plan" means the unfunded plan of deferred compensation set forth herein, including the Adoption Agreement and any trust agreement, as adopted by the Plan Sponsor and as amended from time to time.

2.19. Plan Sponsor

"Plan Sponsor" means the entity identified in Section 1.03 of the Adoption Agreement or any successor by merger, consolidation or otherwise.

2.20. Plan Year

"Plan Year" means the period identified in Section 1.02 of the Adoption Agreement.

2.21. Related Employer

"Related Employer" means the Employer and (a) any corporation that is a member of a controlled group of corporations as defined in Code Section 414(b) that includes the Employer and (b) any trade or business that is under common control as defined in Code Section 414(c) that includes the Employer.

2.22. Retirement

"Retirement" has the meaning specified in 6.01(f) of the Adoption Agreement.

2.23. Separation from Service

"Separation from Service" means the date that the Participant dies, retires or otherwise has a termination of employment with respect to all entities comprising the Related Employer. A Separation from Service does not occur if the Participant is on military leave, sick leave or other bona fide leave of absence if the period of leave does not exceed six months or such longer period during which the Participant's right to re-employment is provided by statute or contract. If the period of leave exceeds six months and the Participant's right to re-employment is not provided either by statute or contract, a Separation from Service will be deemed to have occurred on the first day following the sixmonth period. If the period of leave is due to any medically determinable physical or mental impairment that can be expected to result in death or can be expected to last for a continuous period of not less than six months, where the impairment causes the Participant to be unable to perform the duties of his position of employment or any substantially similar position of employment, a 29 month period of absence may be substituted for the six month period.

Whether a termination of employment has occurred is based on whether the facts and circumstances indicate that the Related Employer and the Participant reasonably anticipated that no further services would be performed after a certain date or that the level of bona fide services the Participant would perform after such date (whether as an employee or as an independent contractor) would permanently decrease to no more than 20 percent of the average level of bona fide services performed (whether as an employee or an independent contractor) over the immediately preceding 36 month period (or the full period of services to the Related Employer if the employee has been providing services to the Related Employer for less than 36 months).

An independent contractor is considered to have experienced a Separation from Service with the Related Employer upon the expiration of the contract (or, in the case of more than one contract, all contracts) under which services are performed for the Related Employer if the expiration constitutes a good-faith and complete termination of the contractual relationship.

If a Participant provides services as both an employee and an independent contractor of the Related Employer, the Participant must separate from service both as an employee and as an independent contractor to be treated as having incurred a Separation from Service. If a Participant ceases providing services as an independent contractor and begins providing services as an employee, or ceases providing services as an employee and begins providing services as an independent contractor, the Participant will not be considered to have experienced a Separation from Service until the Participant has ceased providing services in both capacities.

If a Participant provides services both as an employee and as a member of the Board of Directors of a corporate Related Employer (or an analogous position with respect to a noncorporate Related Employer), the services provided as a Director are not taken into account in determining whether the Participant has incurred a Separation from Service as an employee for purposes of a nonqualified deferred compensation plan in which the Participant participates as an employee that is not aggregated under Code Section 409A with any plan in which the Participant participates as a Director.

If a Participant provides services both as an employee and as a member of the Board of Directors of a corporate related Employer (or an analogous position with respect to a noncorporate Related Employer), the services provided as an employee are not taken into account in determining whether the Participant has experienced a Separation from Service as a Director for purposes of a nonqualified deferred compensation plan in which the Participant participates as a Director that is not aggregated under Code Section 409A with any plan in which the Participant participates as an employee.

All determinations of whether a Separation from Service has occurred will be made in a manner consistent with Code Section 409A and the final regulations thereunder.

2.24. Unforeseeable Emergency

"Unforeseeable Emergency" means a severe financial hardship of the Participant resulting from an illness or accident of the Participant, the Participant's spouse, the Participant's Beneficiary, or the Participant's dependent (as defined in Code Section 152, without regard to Code section 152(b)(1), (b)(2) and (d)(1)(B); loss of the Participant's property due to casualty; or other similar extraordinary and unforeseeable circumstances arising as a result of events beyond the control of the Participant.

2.25. Valuation Date

"Valuation Date" means each business day of the Plan Year that the New York Stock Exchange is open.

2.26. Years of Service

"Years of Service" means each one year period for which the Participant receives service credit in accordance with the provisions of Section 7.01(d) of the Adoption Agreement.

CDW, LLC CDW Director Deferred Compensation Plan Article 2-4

Article 3 - Participation

3.1. Participation

The Participants in the Plan shall be those Directors and employees of the Employer who satisfy the requirements of Section 2.01 of the Adoption Agreement.

3.2. Termination of Participation

The Administrator may terminate a Participant's participation in the Plan in a manner consistent with Code Section 409A. If the Employer terminates a Participant's participation before the Participant experiences a Separation from Service, the Participant's vested Accounts shall be paid in accordance with the provisions of Article 9.

CDW, LLC CDW Director Deferred Compensation Plan Article 3-1

Article 4 - Participant Elections

4.1. Deferral Agreement

If permitted by the Plan Sponsor in accordance with Section 4.01 of the Adoption Agreement, each Eligible Employee and Director may elect to defer his Compensation within the meaning of Section 3.01 of the Adoption Agreement by executing in writing or electronically, a deferral agreement in accordance with rules and procedures established by the Administrator and the provisions of this Article 4.

A new deferral agreement must be timely executed for each Plan Year during which the Eligible Employee or Director desires to defer Compensation. An Eligible Employee or Director who does not timely execute a deferral agreement shall be deemed to have elected zero deferrals of Compensation for such Plan Year.

A deferral agreement may be changed or revoked during the period specified by the Administrator. Except as provided in Section 9.3, a deferral agreement becomes irrevocable at the close of the specified period.

4.2. Amount of Deferral

An Eligible Employee or Director may elect to defer Compensation in any amount permitted by Section 4.01(a) of the Adoption Agreement.

4.3. Timing of Election to Defer

Each Eligible Employee or Director who desires to defer Compensation otherwise payable during a Plan Year must execute a deferral agreement within the period preceding the Plan Year specified by the Administrator. Each Eligible Employee who desires to defer Compensation that is a Bonus must execute a deferral agreement within the period preceding the Plan Year during which the Bonus is earned that is specified by the Administrator, except that if the Bonus can be treated as performance based compensation as described in Code Section 409A(a)(4)(B)(iii), the deferral agreement may be executed within the period specified by the Administrator, which period, in no event, shall end after the date which is six months prior to the end of the period during which the Bonus is earned, provided the Participant has performed services continuously from the later of the beginning of the performance period or the date the performance criteria are established through the date the Participant executed the deferral agreement and provided further that the compensation has not yet become 'readily ascertainable' within the meaning of Treas. Reg. § 1.409A-2(a) (8). In addition, if the Compensation qualifies as 'fiscal year compensation' within the meaning of Treas. Reg. § 1.409A-2(a)(6), the deferral agreement may be made not later than the end of the Employer's taxable year immediately preceding the first taxable year of the Employer in which any services are performed for which such Compensation is payable.

CDW, LLC CDW Director Deferred Compensation Plan Article 4-1

Except as otherwise provided below, an employee who is classified or designated as an Eligible Employee during a Plan Year or a Director who is designated as eligible to participate during a Plan Year may elect to defer Compensation otherwise payable during the remainder of such Plan Year in accordance with the rules of this Section 4.3 by executing a deferral agreement within the thirty (30) day period beginning on the date the employee is classified or designated as an Eligible Employee or the date the Director is designated as eligible, whichever is applicable, if permitted by Section 4.01(b)(ii) of the Adoption Agreement. If Compensation is based on a specified performance period that begins before the Eligible Employee or Director executes his deferral agreement, the election will be deemed to apply to the portion of such Compensation equal to the total amount of Compensation for the performance period multiplied by the ratio of the number of days remaining in the performance period after the election becomes irrevocable and effective over the total number of days in the performance period. The rules of this paragraph shall not apply unless the Eligible Employee or Director can be treated as initially eligible in accordance with Treas. Reg. § 1.409A-2(a)(7).

4.4. Election of Payment Schedule and Form of Payment

All elections of a payment schedule and a form of payment will be made in accordance with rules and procedures established by the Administrator and the provisions of this Section 4.4.

- (a) If the Plan Sponsor has elected to permit annual distribution elections in accordance with Section 6.01(h) of the Adoption Agreement the following rules apply. At the time an Eligible Employee or Director completes a deferral agreement, the Eligible Employee or Director must elect a distribution event (which includes a specified time) and a form of payment for the Compensation subject to the deferral agreement from among the options the Plan Sponsor has made available for this purpose and which are specified in 6.01(b) of the Adoption Agreement. Prior to the time required by Treas. Reg. § 1.409A-2, the Eligible Employee or Director shall elect a distribution event (which includes a specified time) and a form of payment for any Employer contributions that may be credited to the Participant's Account during the Plan Year. If an Eligible Employee or Director fails to elect a distribution event, he shall be deemed to have elected Separation from Service as the distribution event. If he fails to elect a form of payment, he shall be deemed to have elected a lump sum form of payment.
- (b) If the Plan Sponsor has elected not to permit annual distribution elections in accordance with Section 6.01(h) of the Adoption Agreement the following rules apply. At the time an Eligible Employee or Director first completes a deferral agreement but in no event later than the time required by Treas. Reg. § 1.409A-2, the Eligible Employee or Director must elect a distribution event (which includes a specified time) and a form of payment for amounts credited to his Account from among the options the Plan Sponsor has made available for this purpose and which are specified in Section 6.01(b) of the Adoption Agreement. If an Eligible Employee or Director fails to elect a distribution event, he shall be deemed to have elected Separation from Service in the distribution event. If the fails to elect a form of payment, he shall be deemed to have elected a lump sum form of payment.

CDW, LLC CDW Director Deferred Compensation Plan Article 4-2

Article 5 - Employer Contributions

5.1. Matching Contributions

If elected by the Plan Sponsor in Section 5.01(a) of the Adoption Agreement, the Employer will credit the Participant's Account with a matching contribution determined in accordance with the formula specified in Section 5.01(a) of the Adoption Agreement. The matching contribution will be treated as allocated to the Participant's Account at the time specified in Section 5.01(a)(iii) of the Adoption Agreement.

5.2. Other Contributions

If elected by the Plan Sponsor in Section 5.01(b) of the Adoption Agreement, the Employer will credit the Participant's Account with a contribution determined in accordance with the formula or method specified in Section 5.01(b) of the Adoption Agreement. The contribution will be treated as allocated to the Participant's Account at the time specified in Section 5.01(b)(iii) of the Adoption Agreement.

CDW, LLC CDW Director Deferred Compensation Plan Article 5-1

Article 6 - Accounts and Credits

6.1. Establishment of Account

For accounting and computational purposes only, the Administrator will establish and maintain an Account on behalf of each Participant which will reflect the credits made pursuant to Section 6.2, distributions or withdrawals, along with the earnings, expenses, gains and losses allocated thereto, attributable to the hypothetical investments made with the amounts in the Account as provided in Article 7. The Administrator may establish and maintain such other records and accounts, as it decides in its discretion to be reasonably required or appropriate to discharge its duties under the Plan.

6.2. Credits to Account

A Participant's Account will be credited for each Plan Year with the amount of his elective deferrals under Section 4.1 at the time the amount subject to the deferral election would otherwise have been payable to the Participant and the amount of Employer contributions, if any, treated as allocated on his behalf under Article 5.

CDW, LLC CDW Director Deferred Compensation Plan Article 6-1

Article 7 - Investment of Contributions

7.1. Investment Options

The amount credited to each Account shall be treated as invested in the investment options designated for this purpose by the Administrator.

7.2. Adjustment of Accounts

The amount credited to each Account shall be adjusted for hypothetical investment earnings, expenses, gains or losses in an amount equal to the earnings, expenses, gains or losses attributable to the investment options selected by the party designated in Section 9.01 of the Adoption Agreement from among the investment options provided in Section 7.1. If permitted by Section 9.01 of the Adoption Agreement, a Participant (or the Participant's Beneficiary after the death of the Participant) may, in accordance with rules and procedures established by the Administrator, select the investments from among the options provided in Section 7.1 to be used for the purpose of calculating future hypothetical investment adjustments to the Account or to future credits to the Account under Section 6.2 effective as of the Valuation Date coincident with or next following notice to the Administrator. Each Account shall be adjusted as of each Valuation Date to reflect: (a) the hypothetical earnings, expenses, gains and losses described above; (b) amounts credited pursuant to Section 6.2; and (c) distributions or withdrawals. In addition, each Account may be adjusted for its allocable share of the hypothetical costs and expenses associated with the maintenance of the hypothetical investments provided in Section 7.1.

CDW, LLC CDW Director Deferred Compensation Plan Article 7-1

Article 8 - Right to Benefits

8.1. Vesting

A Participant, at all times, has a 100% nonforfeitable interest in the amounts credited to his Account attributable to his elective deferrals made in accordance with Section 4.1.

A Participant's right to the amounts credited to his Account attributable to Employer contributions made in accordance with Article 5 shall be determined in accordance with the relevant schedule and provisions in Section 7.01 of the Adoption Agreement. Upon a Separation from Service and after application of the provisions of Section 7.01 of the Adoption Agreement, the Participant shall forfeit the nonvested portion of his Account.

8.2. Death

The Plan Sponsor may elect to accelerate vesting upon the death of the Participant in accordance with Section 7.01(c) of the Adoption Agreement and/or to accelerate distributions upon death in accordance with Section 6.01(b) or Section 6.01(d) of the Adoption Agreement. If the Plan Sponsor does not elect to accelerate distributions upon death in accordance with Section 6.01(b) or Section 6.01(d) of the Adoption Agreement, the vested amount credited to the Participant's Account will be paid in accordance with the provisions of Article 9.

A Participant may designate a Beneficiary or Beneficiaries, or change any prior designation of Beneficiary or Beneficiaries in accordance with rules and procedures established by the Administrator. Whenever a Participant designates a new Beneficiary, all former Beneficiary designations by such Participant shall be revoked automatically. If a Participant and the Participant's spouse divorce, any designations of the spouse as Beneficiary shall become null and void. The former spouse shall be treated as the Beneficiary under the Plan only if after the divorce is final, the Participant expressly re-designates the former spouse as the Participant's Beneficiary.

A copy of the death notice or other sufficient documentation must be filed with and approved by the Administrator. If upon the death of the Participant there is, in the opinion of the Administrator, no designated Beneficiary for part or all of the Participant's vested Account, such amount will be paid to his estate (such estate shall be deemed to be the Beneficiary for purposes of the Plan) in accordance with the provisions of Article 9.

8.3. Disability

If the Plan Sponsor has elected to accelerate vesting upon the occurrence of a Disability in accordance with Section 7.01(c) of the Adoption Agreement and/or to permit distributions upon Disability in accordance with Section 6.01(b) or Section 6.01(d) of the Adoption Agreement, the determination of whether a Participant has incurred a Disability shall be made by the Administrator in its sole discretion in a manner consistent with the requirements of Code Section 409A.

CDW, LLC CDW Director Deferred Compensation Plan Article 8-1

Article 9 - Distribution of Benefits

9.1. Amount of Benefits

The vested amount credited to a Participant's Account as determined under Articles 6, 7 and 8 shall determine and constitute the basis for the value of benefits payable to the Participant under the Plan.

9.2. Method and Timing of Distributions

Except as otherwise provided in this Article 9, distributions under the Plan shall be made in accordance with the elections made or deemed made by the Participant under Article 4. Subject to the provisions of Section 9.6 requiring a six month delay for certain distributions to Key Employees, distributions following a payment event shall commence at the time specified in Section 6.01(a) of the Adoption Agreement. If permitted by Section 6.01(g) of the Adoption Agreement, a Participant may elect, at least twelve months before a scheduled distribution event, to delay the payment date for a minimum period of sixty months from the originally scheduled date of payment, provided the election does not take effect for at least twelve months from the date on which the election is made. The distribution election change must be made in accordance with procedures and rules established by the Administrator. The Participant may, at the same time the date of payment is deferred, change the form of payment but such change in the form of payment may not effect an acceleration of payment in violation of Code Section 409A or the provisions of Treas. Reg. § 1.409A-2(b). For purposes of this Section 9.2, a series of installment payments is always treated as a single payment and not as a series of separate payments.

9.3. Unforeseeable Emergency

A Participant may request a distribution due to an Unforeseeable Emergency if the Plan Sponsor has elected to permit Unforeseeable Emergency withdrawals under Section 8.01(a) of the Adoption Agreement. The request must be in writing and must be submitted to the Administrator along with evidence that the circumstances constitute an Unforeseeable Emergency. The Administrator has the discretion to require whatever evidence it deems necessary to determine whether a distribution is warranted, and may require the Participant to certify that the need cannot be met from other sources reasonably available to the Participant. Whether a Participant has incurred an Unforeseeable Emergency will be determined by the Administrator on the basis of the relevant facts and circumstances in its sole discretion, but, in no event, will an Unforeseeable Emergency be deemed to exist if the hardship can be relieved: (a) through reimbursement or compensation by insurance or otherwise, (b) by liquidation of the Participant's assets to the extent such liquidation would not itself cause severe financial hardship, or (c) by cessation of deferrals under the Plan. A distribution due to an Unforeseeable Emergency must be limited to the amount reasonably necessary to satisfy the emergency need and may include any amounts necessary to pay any federal, state, foreign or local income taxes and penalties reasonably anticipated to result from the distribution. The distribution will be made in the form of a single lump sum cash payment. If permitted by Section 8.01(b) of the Adoption Agreement, a Participant's deferral elections for the remainder of the Plan Year will be cancelled upon a withdrawal due to an Unforeseeable Emergency. If the payment of all or any portion of the Participant's vested Account is being delayed in accordance with Section 9.6 at the time he experiences an Unforeseeable Emergency, the amount being delayed shall not be subject to the provisions of this Section 9.3 until the expiration of the six month period of delay required by section 9.6.

CDW, LLC CDW Director Deferred Compensation Plan Article 9-1

9.4. Payment Election Overrides

If the Plan Sponsor has elected one or more payment election overrides in accordance with Section 6.01(d) of the Adoption Agreement, the following provisions apply. Upon the occurrence of the first event selected by the Plan Sponsor, the remaining vested amount credited to the Participant's Account shall be paid in the form designated to the Participant or his Beneficiary regardless of whether the Participant had made different elections of time and/or form of payment or whether the Participant was receiving installment payments at the time of the event.

9.5. Cashouts of Amounts Not Exceeding Stated Limit

If the vested amount credited to the Participant's Account does not exceed the limit established for this purpose by the Plan Sponsor in Section 6.01(e) of the Adoption Agreement at the time he incurs a Separation from Service for any reason, the Employer shall distribute such amount to the Participant at the time specified in Section 6.01(a) of the Adoption Agreement in a single lump sum cash payment following such Separation from Service regardless of whether the Participant had made different elections of time or form of payment as to the vested amount credited to his Account or whether the Participant was receiving installments at the time of such termination. A Participant's Account, for purposes of this Section 9.5, shall include any amounts described in Section 1.3.

9.6. Required Delay in Payment to Key Employees

Except as otherwise provided in this Section 9.6, a distribution made on account of Separation from Service (or Retirement, if applicable) to a Participant who is a Key Employee as of the date of his Separation from Service (or Retirement, if applicable) shall not be made before the date which is six months after the Separation from Service (or Retirement, if applicable).

- (a) A Participant is treated as a Key Employee if: (i) he is employed by a Related Employer any of whose stock is publicly traded on an established securities market, and (ii) he satisfies the requirements of Code Section 416(i)(1) (A)(i), (ii) or (iii), determined without regard to Code Section 416(i)(5), at any time during the twelve month period ending on the Identification Date.
- (b) A Participant who is a Key Employee on an Identification Date shall be treated as a Key Employee for purposes of the six month delay in distributions for the twelve month period beginning on the first day of a month no later than the fourth month following the Identification Date. The Identification Date and the effective date of the delay in distributions shall be determined in accordance with Section 1.06 of the Adoption Agreement.

CDW, LLC CDW Director Deferred Compensation Plan Article 9-2

- (c) The Plan Sponsor may elect to apply an alternative method to identify Participants who will be treated as Key Employees for purposes of the six month delay in distributions if the method satisfies each of the following requirements: (i) is reasonably designed to include all Key Employees, (ii) is an objectively determinable standard providing no direct or indirect election to any Participant regarding its application, and (iii) results in either all Key Employees or no more than 200 Key Employees being identified in the class as of any date. Use of an alternative method that satisfies the requirements of this Section 9.6(c) will not be treated as a change in the time and form of payment for purposes of Treas. Reg. § 1.409A-2(b).
- (d) The six month delay does not apply to payments described in Section 9.9(a), (b) or (d) or to payments that occur after the death of the Participant. If the payment of all or any portion of the Participant's vested Account is being delayed in accordance with this Section 9.6 at the time he incurs a Disability which would otherwise require a distribution under the terms of the Plan, no amount shall be paid until the expiration of the six month period of delay required by this Section 9.6.

9.7. Change in Control

If the Plan Sponsor has elected to permit distributions upon a Change in Control, the following provisions shall apply. A distribution made upon a Change in Control will be made at the time specified in Section 6.01(a) of the Adoption Agreement in the form elected by the Participant in accordance with the procedures described in Article 4. Alternatively, if the Plan Sponsor has elected in accordance with Section 11.02 of the Adoption Agreement to require distributions upon a Change in Control, the Participant's remaining vested Account shall be paid to the Participant or the Participant's Beneficiary at the time specified in Section 6.01(a) of the Adoption Agreement as a single lump sum payment. A Change in Control, for purposes of the Plan, will occur upon a change in the ownership of the Plan Sponsor, a change in the effective control of the Plan Sponsor or a change in the ownership of a substantial portion of the assets of the Plan Sponsor, but only if elected by the Plan Sponsor in Section 11.03 of the Adoption Agreement. The Plan Sponsor, for this purpose, includes any corporation identified in this Section 9.7. All distributions made in accordance with this Section 9.7 are subject to the provisions of Section 9.6.

If a Participant continues to make deferrals in accordance with Article 4 after he has received a distribution due to a Change in Control, the residual amount payable to the Participant shall be paid at the time and in the form specified in the elections he makes in accordance with Article 4 or upon his death or Disability as provided in Article 8.

Whether a Change in Control has occurred will be determined by the Administrator in accordance with the rules and definitions set forth in this Section 9.7. A distribution to the Participant will be treated as occurring upon a Change in Control if the Plan Sponsor terminates the Plan in accordance with Section 10.2 and distributes the Participant's benefits within twelve months of a Change in Control as provided in Section 10.3.

- (a) Relevant Corporations. To constitute a Change in Control for purposes of the Plan, the event must relate to: (i) the corporation for whom the Participant is performing services at the time of the Change in Control, (ii) the corporation that is liable for the payment of the Participant's benefits under the Plan (or all corporations liable if more than one corporation is liable) but only if either the deferred compensation is attributable to the performance of services by the Participant for such corporation (or corporations) or there is a bona fide business purpose for such corporation (or corporations) to be liable for such payment and, in either case, no significant purpose of making such corporation (or corporations) liable for such payment is the avoidance of federal income tax, or (iii) a corporation that is a majority shareholder of a corporation identified in (i) or (ii), or any corporation in a chain of corporations in which each corporation is a majority shareholder of another corporation in the chain, ending in a corporation identified in (i) or (ii). A majority shareholder is defined as a shareholder owning more than fifty percent (50%) of the total fair market value and voting power of such corporation.
- (b) Stock Ownership. Code Section 318(a) applies for purposes of determining stock ownership. Stock underlying a vested option is considered owned by the individual who owns the vested option (and the stock underlying an unvested option is not considered owned by the individual who holds the unvested option). If, however, a vested option is exercisable for stock that is not substantially vested (as defined by Treas. Reg. § 1.83-3(b) and (j)) the stock underlying the option is not treated as owned by the individual who holds the option.
- (c) Change in the Ownership of a Corporation. A change in the ownership of a corporation occurs on the date that any one person or more than one person acting as a group, acquires ownership of stock of the corporation that, together with stock held by such person or group, constitutes more than fifty percent (50%) of the total fair market value or total voting power of the stock of such corporation. If any one person or more than one person acting as a group is considered to own more than fifty percent (50%) of the total fair market value or total voting power of the stock of a corporation, the acquisition of additional stock by the same person or persons is not considered to cause a change in the ownership of the corporation (or to cause a change in the effective control of the corporation as discussed below in Section 9.7(d)). An increase in the percentage of stock owned by any one person, or persons acting as a group, as a result of a transaction in which the corporation acquires its stock in exchange for property will be treated as an acquisition of stock. Section 9.7(c) applies only when there is a transfer of stock of a corporation (or issuance of stock of a corporation) and stock in such corporation remains outstanding after the transaction. For purposes of this Section 9.7(c), persons will not be considered to be acting as a group solely because they purchase or own stock of the same corporation at the same time or as a result of a public offering. Persons will, however, be considered to be acting as a group if they are owners of a corporation that enters into a merger, consolidation, purchase or acquisition of stock, or similar business transaction with the corporation. If a person, including an entity, owns stock in both corporations that enter into a merger, consolidation, purchase or acquisition of stock, or similar transaction, such shareholder is considered to be acting as a group with other shareholders in a corporation only with respect to ownership in that corporation prior to the transaction giving rise to the change and not with respect to the ownership interest in the other corporation.

(d) Change in the Effective Control of a Corporation. A change in the effective control of a corporation occurs on the date that either (i) any one person, or more than one person acting as a group, acquires (or has acquired during the twelve month period ending on the date of the most recent acquisition by such person or persons) ownership of stock of the corporation possessing thirty percent (30%) or more of the total voting power of the stock of such corporation, or (ii) a majority of members of the corporation's Board of Directors is replaced during any twelve month period by Directors whose appointment or election is not endorsed by a majority of the members of the corporation's Board of Directors prior to the date of the appointment or election, provided that for purposes of this paragraph (ii), the term corporation refers solely to the relevant corporation identified in Section 9.7(a) for which no other corporation is a majority shareholder for purposes of Section 9.7(a). In the absence of an event described in Section 9.7(d)(i) or (ii), a change in the effective control of a corporation will not have occurred. A change in effective control may also occur in any transaction in which either of the two corporations involved in the transaction has a change in the ownership of such corporation as described in Section 9.7(c) or a change in the ownership of a substantial portion of the assets of such corporation as described in Section 9.7(e). If any one person, or more than one person acting as a group, is considered to effectively control a corporation within the meaning of this Section 9.7(d), the acquisition of additional control of the corporation by the same person or persons is not considered to cause a change in the effective control of the corporation or to cause a change in the ownership of the corporation within the meaning of Section 9.7(c). For purposes of this Section 9.7(d), persons will or will not be considered to be acting as a group in accordance with rules similar to those set forth in Section 9.7(c) with the following exception. If a person, including an entity, owns stock in both corporations that enter into a merger, consolidation, purchase or acquisition of stock, or similar transaction, such shareholder is considered to be acting as a group with other shareholders in a corporation only with respect to the ownership in that corporation prior to the transaction giving rise to the change and not with respect to the ownership interest in the other corporation.

Change in the Ownership of a Substantial Portion of a Corporation's Assets. A change in the ownership of a (a) substantial portion of a corporation's assets occurs on the date that any one person, or more than one person acting as a group (as determined in accordance with rules similar to those set forth in Section 9.7(d)), acquires (or has acquired during the twelve month period ending on the date of the most recent acquisition by such person or persons) assets from the corporation that have a total gross fair market value equal to or more than forty percent (40%) of the total gross fair market value of all of the assets of the corporation immediately prior to such acquisition or acquisitions. For this purpose, gross fair market value means the value of the assets of the corporation or the value of the assets being disposed of determined without regard to any liabilities associated with such assets. There is no Change in Control event under this Section 9.7(e) when there is a transfer to an entity that is controlled by the shareholders of the transferring corporation immediately after the transfer. A transfer of assets by a corporation is not treated as a change in ownership of such assets if the assets are transferred to (i) a shareholder of the corporation (immediately before the asset transfer) in exchange for or with respect to its stock, (ii) an entity, fifty percent (50%) or more of the total value or voting power of which is owned, directly or indirectly, by the corporation, (iii) a person, or more than one person acting as a group, that owns, directly or indirectly, fifty percent (50%) or more of the total value or voting power of all the outstanding stock of the corporation, or (iv) an entity, at least fifty (50%) of the total value or voting power of which is owned, directly or indirectly, by a person described in Section 9.7(e)(iii). For purposes of the foregoing, and except as otherwise provided, a person's status is determined immediately after the transfer of assets.

9.8. Permissible Delays in Payment

Distributions may be delayed beyond the date payment would otherwise occur in accordance with the provisions of Articles 8 and 9 in any of the following circumstances (as long as the Employer treats all payments to similarly situated Participants on a reasonably consistent basis):

- (a) The Employer may delay payment if it reasonably anticipates that its deduction with respect to such payment would be limited or eliminated by the application of Code Section 162(m). Payment must be made during the Participant's first taxable year in which the Employer reasonably anticipates, or should reasonably anticipate, that if the payment is made during such year the deduction of such payment will not be barred by the application of Code Section 162(m) or during the period beginning with the Participant's Separation from Service and ending on the later of the last day of the Employer's taxable year in which the Participant separates from service or the 15th day of the third month following the Participant's Separation from Service. If a scheduled payment to a Participant is delayed in accordance with this Section 9.8(a), all scheduled payments to the Participant that could be delayed in accordance with this Section 9.8(a) will also be delayed.
- (b) The Employer may also delay payment if it reasonably anticipates that the making of the payment will violate federal securities laws or other applicable laws provided payment is made at the earliest date on which the Employer reasonably anticipates that the making of the payment will not cause such violation.

The Employer reserves the right to amend the Plan to provide for a delay in payment upon such other events and (c) conditions as the Secretary of the Treasury may prescribe in generally applicable guidance published in the Internal Revenue Bulletin.

9.9. Permitted Acceleration of Payment

The Employer may permit acceleration of the time or schedule of any payment or amount scheduled to be paid pursuant to a payment under the Plan provided such acceleration would be permitted by the provisions of Treas. Reg. § 1.409A-3(j) (4), including the following events:

- Domestic Relations Order. A payment may be accelerated if such payment is made to an alternate payee pursuant (a) to and following the receipt and qualification of a domestic relations order as defined in Code Section 414(p).
- Compliance with Ethics Agreement and Legal Requirements. A payment may be accelerated as may be necessary to (b) comply with ethics agreements with the Federal government or as may be reasonably necessary to avoid the violation of Federal, state, local or foreign ethics law or conflicts of laws, in accordance with the requirements of Code Section 409A.
- De Minimis Amounts. A payment will be accelerated if (i) the amount of the payment is not greater than the (c) applicable dollar amount under Code Section 402(g)(1)(B), (ii) at the time the payment is made the amount constitutes the Participant's entire interest under the Plan and all other plans that are aggregated with the Plan under Treas. Reg. § 1.409A-1(c)(2).
- (d) FICA Tax. A payment may be accelerated to the extent required to pay the Federal Insurance Contributions Act tax imposed under Code Sections 3101, 3121(a) and 3121(v)(2) of the Code with respect to compensation deferred under the Plan (the "FICA Amount"). Additionally, a payment may be accelerated to pay the income tax on wages imposed under Code Section 3401 of the Code on the FICA Amount and to pay the additional income tax at source on wages attributable to the pyramiding Code Section 3401 wages and taxes. The total payment under this subsection (d) may not exceed the aggregate of the FICA Amount and the income tax withholding related to the FICA Amount.
- (e) Section 409A Additional Tax. A payment may be accelerated if the Plan fails to meet the requirements of Code Section 409A; provided that such payment may not exceed the amount required to be included in income as a result of the failure to comply with the requirements of Code Section 409A.
- (f) Offset. A payment may be accelerated in the Employer's discretion as satisfaction of a debt of the Participant to the Employer, where such debt is incurred in the ordinary course of the service relationship between the Participant and the Employer, the entire amount of the reduction in any of the Employer's taxable years does not exceed \$5,000, and the reduction is made at the same time and in the same amount as the debt otherwise would have been due and collected from the Participant.
- Other Events. A payment may be accelerated in the Administrator's discretion in connection with such other (g) events and conditions as permitted by Code Section 409A.

Article 10 - Amendment and Termination

10.1. Amendment by Plan Sponsor

The Plan Sponsor reserves the right to amend the Plan (for itself and each Employer) through action of its Board of Directors. No amendment can directly or indirectly deprive any current or former Participant or Beneficiary of all or any portion of his Account which had accrued and vested prior to the amendment.

10.2. Plan Termination Following Change in Control or Corporate Dissolution

If so elected by the Plan Sponsor in 11.01 of the Adoption Agreement, the Plan Sponsor reserves the right to terminate the Plan and distribute all amounts credited to all Participant Accounts within the 30 days preceding or the twelve months following a Change in Control as determined in accordance with the rules set forth in Section 9.7. For this purpose, the Plan will be treated as terminated only if all agreements, methods, programs and other arrangements sponsored by the Related Employer immediately after the Change in Control which are treated as a single plan under Treas. Reg. § 1.409A-1(c)(2) are also terminated so that all Participants under the Plan and all similar arrangements are required to receive all amounts deferred under the terminated arrangements within twelve months of the date the Plan Sponsor irrevocably takes all necessary action to terminate the arrangements. In addition, the Plan Sponsor reserves the right to terminate the Plan within twelve months of a corporate dissolution taxed under Code Section 331 or with the approval of a bankruptcy court pursuant to 11 U. S. C. Section 503(b)(1)(A) provided that amounts deferred under the Plan are included in the gross incomes of Participants in the latest of (a) the calendar year in which the termination and liquidation occurs, (b) the first calendar year in which the amount is no longer subject to a substantial risk of forfeiture, or (c) the first calendar year in which payment is administratively practicable.

10.3. Other Plan Terminations

The Plan Sponsor retains the discretion to terminate the Plan if (a) all arrangements sponsored by the Plan Sponsor that would be aggregated with any terminated arrangement under Code Section 409A and Treas. Reg. § 1.409A-1(c)(2) are terminated, (b) no payments other than payments that would be payable under the terms of the arrangements if the termination had not occurred are made within twelve months of the termination of the arrangements, (c) all payments are made within twenty-four months of the date the Plan Sponsor takes all necessary action to irrevocably terminate and liquidate the arrangements,

(d)the Plan Sponsor does not adopt a new arrangement that would be aggregated with any terminated arrangement under Code Section 409A and the regulations thereunder at any time within the three year period following the date of termination of the arrangement, and

(e) the termination does not occur proximate to a downturn in the financial health of the Plan Sponsor. The Plan Sponsor also reserves the right to amend the Plan to provide that termination of the Plan will occur under such conditions and events as may be prescribed by the Secretary of the Treasury in generally applicable guidance published in the Internal Revenue Bulletin.

Article 11 - The Trust

11.1. Establishment of Trust

The Plan Sponsor may but is not required to establish a trust to hold amounts which the Plan Sponsor may contribute from time to time to correspond to some or all amounts credited to Participants under Section 6.2. In the event that the Plan Sponsor wishes to establish a trust to provide a source of funds for the payment of Plan benefits, any such trust shall be constructed to constitute an unfunded arrangement that does not affect the status of the Plan as an unfunded plan for purposes of Title I of ERISA and the Code. If the Plan Sponsor elects to establish a trust in accordance with Section 10.01 of the Adoption Agreement, the provisions of Sections 11.2 and 11.3 shall become operative.

11.2. Rabbi Trust

Any trust established by the Plan Sponsor shall be between the Plan Sponsor and a trustee pursuant to a separate written agreement under which assets are held, administered and managed, subject to the claims of the Plan Sponsor's creditors in the event of the Plan Sponsor's insolvency. The trust is intended to be treated as a rabbi trust in accordance with existing guidance of the Internal Revenue Service, and the establishment of the trust shall not cause the Participant to realize current income on amounts contributed thereto. The Plan Sponsor must notify the trustee in the event of a bankruptcy or insolvency.

11.3. Investment of Trust Funds

Any amounts contributed to the trust by the Plan Sponsor shall be invested by the trustee in accordance with the provisions of the trust and the instructions of the Administrator. Trust investments need not reflect the hypothetical investments selected by Participants under Section 7.1 for the purpose of adjusting Accounts and the earnings or investment results of the trust need not affect the hypothetical investment adjustments to Participant Accounts under the Plan.

CDW, LLC CDW Director Deferred Compensation Plan Article 11-1

Article 12 - Plan Administration

12.1. Powers and Responsibilities of the Administrator

The Administrator has the full power and the full responsibility to administer the Plan in all of its details; subject, however, to the applicable requirements of ERISA. The Administrator's powers and responsibilities include, but are not limited to, the following:

- (a) To make and enforce such rules and procedures as it deems necessary or proper for the efficient administration of the Plan;
- (b) To interpret the Plan, its interpretation thereof to be final, except as provided in Section 12.2, on all persons claiming benefits under the Plan;
- (c) To decide all questions concerning the Plan and the eligibility of any person to participate in the Plan;
- (d) To administer the claims and review procedures specified in Section 12.2;
- (e) To compute the amount of benefits which will be payable to any Participant, former Participant or Beneficiary in accordance with the provisions of the Plan;
- (f) To determine the person or persons to whom such benefits will be paid;
- (g) To authorize the payment of benefits;
- (h) To make corrections and recover the overpayment of any benefits;
- (i) To comply with the reporting and disclosure requirements of Part 1 of Subtitle B of Title I of ERISA;
- (j) To appoint such agents, counsel, accountants, and consultants as may be required to assist in administering the Plan;
- (k) By written instrument, to allocate and delegate its responsibilities, including the formation of an Administrative Committee to administer the Plan.

CDW, LLC CDW Director Deferred Compensation Plan Article 12-1

12.2. Claims and Review Procedures

- (a) <u>Claims Procedure.</u> If any person believes he is being denied any rights or benefits under the Plan, such person may file a claim in writing with the Administrator. If any such claim is wholly or partially denied, the Administrator will notify such person of its decision in writing. Such notification will contain (i) specific reasons for the denial,
 - (ii)specific reference to pertinent Plan provisions, (iii) a description of any additional material or information necessary for such person to perfect such claim and an explanation of why such material or information is necessary, and (iv) a description of the Plan's review procedures and the time limits applicable to such procedures, including a statement of the person's right to bring a civil action following an adverse decision on review. If the claim involves a Disability, the denial must also include the standards that governed the decision, including the basis for disagreeing with any health care professionals, vocational professionals or the Social Security Administration as well as an explanation of the scientific or clinical judgement underlying the denial. Such notification will be given within 90 days (45 days in the case of a claim regarding Disability) after the claim is received by the Administrator. The Administrator may extend the period for providing the notification by 90 days (30 days in the case of a claim regarding Disability, which may be extended an additional 30 days) if special circumstances require an extension of time for processing the claim and if written notice of such extension and circumstance is given to such person within the initial 90 day period (45 day period in the case of a claim regarding Disability). If such notification is not given within such period, the claim will be considered denied as of the last day of such period and such person may request a review of his claim.
- (b) Review Procedure. Within 60 days (180 days in the case of a claim regarding Disability) after the date on which a person receives a written notification of denial of claim (or, if written notification is not provided, within 60 days (180 days in the case of a claim regarding Disability) of the date denial is considered to have occurred), such person (or his duly authorized representative) may (i) file a written request with the Administrator for a review of his denied claim and of pertinent documents and
 - (ii) submit written issues and comments to the Administrator. The Administrator will notify such person of its decision in writing. Such notification will be written in a manner calculated to be understood by such person and will contain specific reasons for the decision as well as specific references to pertinent Plan provisions. The notification will explain that the person is entitled to receive, upon request and free of charge, reasonable access to and copies of all pertinent documents and has the right to bring a civil action following an adverse decision on review. The decision on review will be made within 60 days (45 days in the case of a claim regarding Disability). The Administrator may extend the period for making the decision on review by 60 days (45 days in the case of a claim regarding Disability) if special circumstances require an extension of time for processing the request such as an election by the Administrator to hold a hearing, and if written notice of such extension and circumstances is given to such person within the initial 60-day period (45 days in the case of a claim regarding Disability). If the decision on review is not made within such period, the claim will be considered denied.

If the claim is regarding Disability, and the determination of Disability has not been made by the Social Security Administration, the Railroad Retirement Board, or under the Plan Sponsor's long-term disability plan, the person may, upon written request and free of charge, also receive the identification of medical or vocational experts whose advice was obtained in connection with the denial of a claim regarding Disability, even if the advice was not relied upon.

Before issuing any decision with respect to a claim involving Disability, the Administrator will provide to the person, free of charge, the following information as soon as possible and sufficiently in advance of the date on which the response is required to be provided to the person to allow the person a reasonable opportunity to respond prior to the due date of the response:

- (i) Any new or additional evidence considered, relied upon, or generated by the Administrator or other person making the decision; and
- (ii) A new or addition rationale if the decision will be based on that rationale.
- (c) <u>Exhaustion of Claims Procedures and Right to Bring Legal Claim.</u> No action at law or equity shall be brought more than one year after the Administrator's affirmation of a denial of a claim, or, if earlier, more than four years after the facts or events giving rising to the claimant's allegation(s) or claim(s) first occurred.

12.3. Plan Administrative Costs

All reasonable costs and expenses (including legal, accounting, and employee communication fees) incurred by the Administrator in administering the Plan shall be paid by the Plan to the extent not paid by the Employer.

CDW, LLC
CDW Director Deferred Compensation Plan Article 12-3

Article 13 - Miscellaneous

13.1. Unsecured General Creditor of the Employer

Participants and their Beneficiaries, heirs, successors and assigns shall have no legal or equitable rights, interests or claims in any property or assets of the Employer. For purposes of the payment of benefits under the Plan, any and all of the Employer's assets shall be, and shall remain, the general, unpledged, unrestricted assets of the Employer. Each Employer's obligation under the Plan shall be merely that of an unfunded and unsecured promise to pay money in the future.

13.2. Employer's Liability

Each Employer's liability for the payment of benefits under the Plan shall be defined only by the Plan and by the deferral agreements entered into between a Participant and the Employer. An Employer shall have no obligation or liability to a Participant under the Plan except as provided by the Plan and a deferral agreement or agreements. An Employer shall have no liability to Participants employed by other Employers.

13.3. Limitation of Rights

Neither the establishment of the Plan, nor any amendment thereof, nor the creation of any fund or account, nor the payment of any benefits, will be construed as giving to the Participant or any other person any legal or equitable right against the Employer, the Plan or the Administrator, except as provided herein; and in no event will the terms of employment or service of the Participant be modified or in any way affected hereby.

13.4. Anti-Assignment

Except as may be necessary to fulfill a domestic relations order within the meaning of Code Section 414(p), none of the benefits or rights of a Participant or any Beneficiary of a Participant shall be subject to the claim of any creditor. In particular, to the fullest extent permitted by law, all such benefits and rights shall be free from attachment, garnishment, or any other legal or equitable process available to any creditor of the Participant and his Beneficiary. Neither the Participant nor his Beneficiary shall have the right to alienate, anticipate, commute, pledge, encumber, or assign any of the payments which he may expect to receive, contingently or otherwise, under the Plan, except the right to designate a Beneficiary to receive death benefits provided hereunder. Notwithstanding the preceding, the benefit payable from a Participant's Account may be reduced, at the discretion of the Administrator, to satisfy any debt or liability to the Employer.

CDW, LLC
CDW Director Deferred Compensation Plan Article 13-1

13.5. Facility of Payment

If the Administrator determines, on the basis of medical reports or other evidence satisfactory to the Administrator, that the recipient of any benefit payments under the Plan is incapable of handling his affairs by reason of minority, illness, infirmity or other incapacity, the Administrator may direct the Employer to disburse such payments to a person or institution designated by a court which has jurisdiction over such recipient or a person or institution otherwise having the legal authority under State law for the care and control of such recipient. The receipt by such person or institution of any such payments therefore, and any such payment to the extent thereof, shall discharge the liability of the Employer, the Plan and the Administrator for the payment of benefits hereunder to such recipient.

13.6. Notices

Any notice or other communication to the Employer or Administrator in connection with the Plan shall be deemed delivered in writing if addressed to the Plan Sponsor at the address specified in Section 1.03 of the Adoption Agreement and if either actually delivered at said address or, in the case or a letter, five business days shall have elapsed after the same shall have been deposited in the United States mails, first-class postage prepaid and registered or certified.

13.7. Tax Withholding

If the Employer concludes that tax is owing with respect to any deferral or payment hereunder, the Employer shall withhold such amounts from any payments due the Participant or from amounts deferred, as permitted by law, or otherwise make appropriate arrangements with the Participant or his Beneficiary for satisfaction of such obligation. Tax, for purposes of this Section 13.7 means any federal, state, local or any other governmental income tax, employment or payroll tax, excise tax, or any other tax or assessment owing with respect to amounts deferred, any earnings thereon, and any payments made to Participants under the Plan.

CDW, LLC
CDW Director Deferred Compensation Plan Article 13-2

13.8. Indemnification

- (a) Each Indemnitee (as defined in Section 13.8(e)) shall be indemnified and held harmless by the Employer for all actions taken by him and for all failures to take action (regardless of the date of any such action or failure to take action), to the fullest extent permitted by the law of the jurisdiction in which the Employer is incorporated, against all expense, liability, and loss (including, without limitation, attorneys' fees, judgments, fines, taxes, penalties, and amounts paid or to be paid in settlement) reasonably incurred or suffered by the Indemnitee in connection with any Proceeding (as defined in subsection (e)). No indemnification pursuant to this Section shall be made, however, in any case where (1) the act or failure to act giving rise to the claim for indemnification is determined by a court to have constituted willful misconduct or recklessness or (2) there is a settlement to which the Employer does not consent.
- (b) The right to indemnification provided in this Section shall include the right to have the expenses incurred by the Indemnitee in defending any Proceeding paid by the Employer in advance of the final disposition of the Proceeding, to the fullest extent permitted by the law of the jurisdiction in which the Employer is incorporated; provided that, if such law requires, the payment of such expenses incurred by the Indemnitee in advance of the final disposition of a Proceeding shall be made only on delivery to the Employer of an undertaking, by or on behalf of the Indemnitee, to repay all amounts so advanced without interest if it shall ultimately be determined that the Indemnitee is not entitled to be indemnified under this Section or otherwise.
- (c) Indemnification pursuant to this Section shall continue as to an Indemnitee who has ceased to be such and shall inure to the benefit of his heirs, executors, and admin- istrators. The Employer agrees that the undertakings made in this Section shall be binding on its successors or assigns and shall survive the termination, amendment or restatement of the Plan.
- (d) The foregoing right to indemnification shall be in addition to such other rights as the Indemnitee may enjoy as a matter of law or by reason of insurance coverage of any kind and is in addition to and not in lieu of any rights to indemnification to which the Indemnitee may be entitled pursuant to the by-laws of the Employer.
- (e) For the purposes of this Section, the following definitions shall apply:
 - (i) "Indemnitee" shall mean each person serving as an Administrator (or any other person who is an employee, Director, or officer of the Employer) who was or is a party to, or is threatened to be made a party to, or is otherwise involved in, any Proceeding, by reason of the fact that he is or was performing administrative functions under the Plan.
 - (ii) "Proceeding" shall mean any threatened, pending, or completed action, suit, or proceeding (including, without limitation, an action, suit, or proceeding by or in the right of the Employer), whether civil, criminal, administrative, investigative, or through arbitration.

13.9. Successors

The provisions of the Plan shall bind and inure to the benefit of the Plan Sponsor, the Employer and their successors and assigns and the Participant and the Participant's designated Beneficiaries.

13.10. Disclaimer

It is the Plan Sponsor's intention that the Plan comply with the requirements of Code Section 409A. Neither the Plan Sponsor nor the Employer shall have any liability to any Participant should any provision of the Plan fail to satisfy the requirements of Code Section 409A.

13.11. Governing Law

The Plan will be construed, administered and enforced according to the laws of the State specified by the Plan Sponsor in Section 12.01 of the Adoption Agreement.

Exhibit 21.1

LIST OF SUBSIDIARIES

Subsidiary	Jurisdiction of Organization	
CDW LLC	Illinois	
CDW Finance Corporation	Delaware	
CDW Technologies LLC	Wisconsin	
CDW Direct, LLC	Illinois	
CDW Government LLC	Illinois	
CDW Logistics LLC	Illinois	
CDW Canada Corp.	Nova Scotia	
CDW NA Limited	United Kingdom	
CDW International Holdings Limited	United Kingdom	
CDW Finance Bidco Limited	United Kingdom	
CDW Finance Holdings Limited	United Kingdom	
CDW Limited	United Kingdom	
CDW Finance Topco Limited	Jersey	
Sirius Computer Solutions, LLC	Texas	

LIST OF ISSUER AND GUARANTOR SUBSIDIARIES

The following subsidiaries of CDW Corporation serve as an issuer or guarantor, as applicable, for each outstanding series of senior notes:

Subsidiary	Jurisdiction of Organization	Type of Obligor
CDW LLC	Illinois	Issuer
CDW Finance Corporation	Delaware	Issuer
CDW Corporation	Delaware	Guarantor
CDW Technologies LLC	Wisconsin	Guarantor
CDW Direct, LLC	Illinois	Guarantor
CDW Government LLC	Illinois	Guarantor
CDW Logistics LLC	Illinois	Guarantor
Amplified IT LLC	Virginia	Guarantor

Consent of Independent Registered Public Accounting Firm

We consent to the incorporation by reference in the following Registration Statements:

- (1) Registration Statement (Form S-3 No. 333-249491) of CDW Corporation,
- (2) Registration Statement (Form S-8 No. 333-256399) pertaining to the Amended and Restated Coworker Stock Purchase Plan of CDW Corporation,
- (3) Registration Statement (Form S-8 No. 333-212912) pertaining to the 2021 Long-Term Incentive Plan and Amended and Restated 2013 Long-Term Incentive Plan of CDW Corporation,
- (4) Registration Statement (Form S-8 No. 333-256398) pertaining to the 2021 Long-Term Incentive Plan of CDW Corporation,
- (5) Registration Statement (Form S-8 No. 333-189622) pertaining to the 2013 Long-Term Incentive Plan and Coworker Stock Purchase Plan of CDW Corporation;

of our reports dated February 28, 2022, with respect to the consolidated financial statements of CDW Corporation and subsidiaries and the effectiveness of internal control over financial reporting of CDW Corporation and subsidiaries included in this Annual Report (Form 10-K) of CDW Corporation for the year ended December 31, 2021.

/s/ Ernst & Young LLP Chicago, Illinois February 28, 2022

CERTIFICATION PURSUANT TO RULE 13a-14(a) or 15d-14(a) UNDER THE SECURITIES EXCHANGE ACT OF 1934

I, Christine A. Leahy, certify that:

- 1. I have reviewed this annual report on Form 10-K of the registrant;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:
 - a. Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
 - b. Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
 - c. Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
 - d. Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of registrant's board of directors (or persons performing the equivalent functions):
 - a. All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
 - b. Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

/s/ Christine A. Leahy

Christine A. Leahy President and Chief Executive Officer CDW Corporation February 28, 2022

CERTIFICATION PURSUANT TO RULE 13a-14(a) or 15d-14(a) UNDER THE SECURITIES EXCHANGE ACT OF 1934

I, Albert J. Miralles, certify that:

- 1. I have reviewed this annual report on Form 10-K of the registrant;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:
 - a. Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared:
 - b. Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
 - c. Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
 - d. Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of registrant's board of directors (or persons performing the equivalent functions):
 - a. All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
 - b. Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

/s/ Albert J. Miralles

Albert J. Miralles
Senior Vice President and Chief Financial Officer
CDW Corporation
February 28, 2022

CERTIFICATION PURSUANT TO SECTION 1350 OF CHAPTER 63 OF TITLE 18 OF THE UNITED STATES CODE

I, Christine A. Leahy, the chief executive officer of CDW Corporation ("CDW"), certify that (i) the Annual Report on Form 10-K for the year ended December 31, 2021 (the "10-K") of CDW fully complies with the requirements of Section 13(a) or 15(d) of the Securities Exchange Act of 1934 and (ii) the information contained in the 10-K fairly presents, in all material respects, the financial condition and results of operations of CDW.

/s/ Christine A. Leahy

Christine A. Leahy
President and Chief Executive Officer
CDW Corporation
February 28, 2022

CERTIFICATION PURSUANT TO SECTION 1350 OF CHAPTER 63 OF TITLE 18 OF THE UNITED STATES CODE

I, Albert J. Miralles, the chief financial officer of CDW Corporation ("CDW"), certify that (i) the Annual Report on Form 10-K for the year ended December 31, 2021 (the "10-K") of CDW fully complies with the requirements of Section 13(a) or 15(d) of the Securities Exchange Act of 1934 and (ii) the information contained in the 10-K fairly presents, in all material respects, the financial condition and results of operations of CDW.

/s/ Albert J. Miralles

Albert J. Miralles Senior Vice President and Chief Financial Officer CDW Corporation February 28, 2022